THE UNIVERSITY OF CALGARY

Existence, Uniqueness and Numerical Realization of Solutions for Thermistor Equations

by

Shijun Zhou

A DISSERTATION

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

DEPARTMENT OF

MATHEMATICS AND STATISTICS

CALGARY, ALBERTA

October, 1993

©Shijun Zhou 1993

National Library of Canada

Acquisitions and Bibliographic Services Branch

395 Wellington Street Ottawa, Ontario K1A 0N4 Bibliothèque nationale du Canada

Direction des acquisitions et des services bibliographiques

395, rue Wellington Ottawa (Ontario) K1A 0N4

Your lile Votre référence

Our file Notre référence

The author has granted an irrevocable non-exclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of his/her thesis by any means and in any form or format, making this thesis available to interested persons.

The author retains ownership of the copyright in his/her thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without his/her permission.

`anadä

L'auteur a accordé une licence irrévocable et exclusive non permettant à la Bibliothèque Canada nationale du de reproduire, prêter, distribuer ou vendre des copies de sa thèse de quelque manière et sous quelque forme que ce soit pour mettre des exemplaires de cette disposition thèse à la des personnes intéressées.

L'auteur conserve la propriété du droit d'auteur qui protège sa thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

ISBN 0-315-94020-4

Name

Dissertation Abstracts International is arranged by broad, general subject categories. Please select the one subject which most nearly describes the content of your dissertation. Enter the corresponding four-digit code in the spaces provided.

SUBJECT TERM

SUBJECT CODE

Subject Categories

THE HUMANITIES AND SOCIAL SCIENCES

COMMUNICATIONS AND THE ARTS

Architecture	0729
Art History	0377
Cinema	0900
Dance	0378
Fine Arts	0357
Information Science	0723
Journalism	0391
Library Science	0399
Mass Communications	0708
Music	0413
Speech Communication	0459
Theater	0464

EDUCATION

General	05	15
Administration	05	14
Adult and Continuing	05	16
Aaricultural	05	17
Art	02	73
Bilingual and Multicultural	02	82
Business	06	88
Community College	02	75
Curriculum and Instruction	07	27
Early Childhood	05	18
Elementary	05	24
Finance	02	77
Guidance and Counseling	05	19
Health	06	80
Higher	07	45
History of	05	20
Home Economics	02	78
Industrial	05	21
Language and Literature	02	79
Mathematics	02	80
Music	05	22
Philosophy of	09	98
Physical	05	23

Psychology 0525 Reading 0535 Religious 0527 Sciences 0714 Secondary 0533 Social Sciences 0534 Sociology of 0340 Special 0529 Teacher Training 0530 Technology 0710 Tests and Measurements 0288 Vocational 0747

LANGUAGE, LITERATURE AND LINGUISTICS

Language	
General	0679
Ancient	0289
Linguistics	0290
Modern	0291
Literature	
General	0401
Classical	0294
Comparative	0295
Medieval	0297
Modern	0298
African	0316
American	059
Asian,	030
Canadian (English)	0352
Canadian (French)	0355
Enalish	0593
Germanic	0311
Latin American	0312
Middle Eastern	0313
Romance	0313
Slavic and East European	0314

PHILOSOPHY, RELIGION AND

THEOLOGY	
Philosophy	0422
Religion	
General	0318
Biblical Studies	0321
Clergy	0319
History of	0320
Philosophy of	0322
Theology	0469
SOCIAL SCIENCES	
American Studies	0323
Anthropology	
Archaeology	0324
Cultural	0326
Physical	0327
Business Administration	
General	0310
Accounting	0272
Banking	0770
Management	0454
Marketing	0338
Canadian Studies	0385
Economics	0501
Agricultural	0501
Commerce-Business	0505
Finance	0508
History	0509
Labor	0510
Theory	0511
Folklore	0358
Geography	0366
Gerontology	0351
History	
General	0578

Ancient	057	'9
Medieval	058	1
Modern	058	2
Black	032	8
African	033	ĭĭ
Asia Australia and Oceania	033	iż.
Canadian	033	ĺ,
Furopeop	033	5
Lotopeur	033	ž
Atidalla Eastern	033	50
Vildale Easiern	033	2
United States	033	
listory of Science	028	S.
aw	035	8
'olifical Science	<u>.</u>	_
General	061	5
International Law and		
Relations	061	6
Public Administration	061	7
Recreation	081	4
iocial Work	045	2
iociology		
General	062	26
Criminology and Penology	062	7
Demography	093	Ŕ
Ethnic and Racial Studies	063	ĩ
Individual and Family	000	
Studies	062	20
Industrial and Labor	002	.0
Polotions	042	0
Dublic and Costal M/alfana	202	.7
Public and Social Weirare	003	U
Social Structure and	0-70	~
Development	0/0	0
Iheory and Methods	034	4
ransportation	070	19
Jrban and Regional Planning	099	9
Nomen's Studies	045	53

THE SCIENCES AND ENGINEERING

BIOLOGICAL SCIENCES Agriculture

General	0473
Agronomy	0285
Animal Culture and	
Nutrition	0475
Animal Pathology	0476
Food Science and	
Technology	0359
Forestry and Wildlife	0478
Plant Culture	0470
Plant Pathology	
Plant Physiology	0400
Plant Physiology	0017
Kange Management	
wood rechnology	0/40
Biology	000/
General	0300
Anatomy	
Biostatistics	0308
Bołany :	0309
Cell	03/9
Ecology	0329
Entomology	0353
Genetics	0369
Limnology	0793
Microbiology	0410
Molecular	0307
Neuroscience	0317
Oceanoaraphy	0416
Physiology	0433
Radiation	0821
Veterinary Science	0778
Zoology	0472
Biophysics	
Genera	0786
Medical	0780
NAMES AND AND	

EARTH SCIENCES

Biogeochemist	iry	0423
Geochemistry		0996

Palynology 042 Physical Geography036 Physical Oceanography041

HEALTH AND ENVIRONMENTAL SCIENCES

JULINULJ	
Environmental Sciences	0768
Health Sciences	
General	0566
Audiology	0300
Chemotherapy	. 0992
Dentistry	0567
Education	0350
Hospital Management	0769
Human Development	0758
Immunology	0982
Medicine and Surgery	0564
Mental Health	0347
Nursing	0565
Nutrition	05/0
Obstetrics and Gynecology	0380
Occupational Health and	000
Inerapy	0354
Ophthalmology	0381
Pathology	05/1
Pharmacology	0415
Pharmacy	05/4
Physical Inerapy	
Public Fiedlin	05/3
Radiology	03/4
kecreation	05/5

Speech Pathology0460 Toxicology0383 Home Economics0386

PHYSICAL SCIENCES

Pure Sciences

Chemistry
General
Aaricultural
Analytical0486
Biochemistry
Inorganic
Nuclear
Organic
Pharmaceutical
Physical0494
Polymer
Radiation0754
Mathematics
hysics
General
Acoustics
Astronomy and
Astrophysics
Atmospheric Science
Atomic0748
Electronics and Electricity 0607
Elementary Particles and
High Energy0/98
Fluid and Plasma
Molecular
Nuclear
Optics
Kadiation
Solid State
Diatistics
Applied Sciences
Applied Mechanics
Computer Science

 Engineering
 0537

 General
 0538

 Agricultural
 0539

 Automotive
 0540

 Biomedical
 0541

 Chemical
 0542

 Civil
 0543

 Electronics and Electrical
 0544

 Heat and Thermodynamics
 0348

 Hydraulic
 0545

 Industrial
 0546

 Materials Science
 0794

 Mechanical
 0548

 Metallurgy
 07433

 Mining
 0551

 Nuclear
 0552

 Packaging
 0549

 Petroleum
 0760

 System Science
 0790

 Geotechnology
 0428

 Operations Research
 0796

 Plastics Technology
 0795

 Textile Technology
 0994

PSYCHOLOGY

General	
Behavioral	
Clinical	
Developmental	
Experimental	
Industrial	
Personality	
Physiological	
Psýchobiology	034
Psychometrics	
Social	045

(*

Dissertation Abstracts International est organisé en catégories de sujets. Veuillez s.v.p. choisir le sujet qui décrit le mieux votre thèse et inscrivez le code numérique approprié dans l'espace réservé ci-dessous.

SUJET

CODE DE SUJET

Catégories par sujets

HUMANITÉS ET SCIENCES SOCIALES

COMMUNICATIONS ET LES ARTS

Architecture	0729
Beaux-arts	0357
Bibliothéconomie	0399
Cinéma	0900
Communication verbale	0459
Communications	0708
Danse	0378
Histoire de l'art	0377
lournalisme	0391
Musique	0413
Sciences de l'information	0723
Théâtre	0465

ÉDUCATION

Généralités	515
Administration	0514
Art	0273
Collèges communautaires	0275
Commerce	0688
Economie domestique	0278
Education permanente	0516
Education préscolaire	0518
Education sanitaire	0680
Enseignement agricole	0517
Enseignement bilingue et	
_ multiculturel	0282
Enseignement industriel	0521
Enseignement primaire	0524
Enseignement protessionnel	0747
Enseignement religieux	0527
Enseignement secondaire	0533
Enseignement spécial	0529
Enseignement supérieur	0745
Evaluation	0288
Finances	0277
Formation des enseignants	0530
Histoire de l'éducation	0520
Langues et littérature	0279

Musique 0280 Musique 0522 Orientation et consultation 0519 Philosophie de l'éducation 0998 Physique 0523 Programmes d'études et originames 0707 Programmes d'études et enseignement 0727 Psychologie 0525 Sciences 0714 Sciences sociales 0534 Sociologie de l'éducation 0340 Technologie 0710

Lecture0535

LANGUE, LITTÉRATURE ET LINGUISTIQUE

Anciennes0289 Littérature Généralités 0401

PHILOSOPHIE, RELIGION ET

Philosophie	0422
Religion	0210
Clorgó	0318
Études bibliques	.0321
Histoire des religions	.0320
Philosophie de la religion	0322
Théologie	0469

SCIENCES SOCIALES

Asthuanalania
Aunitopologie
Archeologie
Culturelle
Physique
Droit 0398
Économia
Economie
Generalites
Commerce-Attaires
Economie agricole
Économie du travail 0510
Financos 0508
Littating 0500
Filstoire
, [héorie
Etudes américaines0323
Études canadiennes 0385
Étudos táministos
roikiore
Geographie
Gérontologie0351
Gestion des affaires
Généralités 0310
A designation 0454
Administration
Banques
Comptabilité0272
Marketina
Histoire
Histoiro généralo 0578

0328 Sociologie interethniques et des relations raciales0631 Structure et développement

SCIENCES ET INGÉNIERIE

SCIENCES BIOLOGIQUES Agriculture

Généralités	.04/.
Agronomie	.0285
Alimentation et technologie	
alimentaire	0359
Culture	.0479
Élevage et alimentation	.0475
Exploitation des péturages	.0777
Pathologie animale	.0476
Pathologie végétale	.0480
Physiologie végétale	.0817
Sylviculture et toune	0478
Technologie du bois	0746
Biologie	
Généralités	.0306
Anatomie	0287
Biologie (Statistiques)	0308
Biologie moléculaire	0307
Botanique	0309
Cellule	0379
Écologie	0329
Entomologie	0353
Génétique	0369
Limnologie	0793
Microbiologie	0410
Neurologie	0317
Océanoaraphie	0416
Physiologie	0433
Rediction	0821
Science vétéringire	0778
Zoologie	0472
Bionhysique	
Généralités	0784
Medicale	0760
	,

SCIENCES DE LA TERRE

Biogeochimie	0423
Géochimie	0996
Géodésie	0370
Géographie physique	0368

SCIENCES DE LA SANTÉ ET DE L'ENVIRONNEMENT

Économie domestique Sciences de l'environnement Sciences de la santé	.0386 .0768
Administration des hipitaux Administration des hipitaux Alimentation et nutrition	.0566
Chimiothérapie Dentisterie Développement humain	.0992
Enseignement Immunologie Loisirs	.0350 .0982 .0575
Médecine du travail et thérapie Médecine et chirurgie	.0354
Obstetrique et gynecologie Ophtalmologie Orthophonie	.0380
Pharmacie Pharmacologie Pharmacologie	.0572
Radiologie Santé mentale Santé publique	.0574
Soins infirmiers Toxicologie	.0569

SCIENCES PHYSIQUES

Sciences Pures
Chimie
Genéralités0485
Biochimie
Chimie agricole
Chimie analytique0486
Chimie minérale0488
Chimie nucléaire0738
Chimie organique0490
Chimie pharmaceutique 0491
Physique
PolymÇres0495
Radiation
Mathématiques0405
Physique
Généralités
Acoustique
Astronomie et
_astrophysique0606
Electronique et électricité0607
Fluides et plasma
Méléorologie0608
Optique0/52
Particules (Physique
nucleaire)0/98
Physique atomique
Physique de l'état solide
Physique moleculaire
Padiation 0610
Radiation
oranishques
· · · · · · · · ·

Sciences Appliqués Et

lechnologie	
nformatique	.0984
ngénierie	
Généralités	.0537
Aaricole	.0539
Automobile	.0540

Biomédicale	.0541
Chaleur et ther	
modynamique	.0348
Conditionnement	
(Emballage)	.0549
Génie aérospatial	.0538
Génie chimique	.0542
Génie civil	0543
Génie électronique et	
électrique	0544
Génie industriel	0546
Génie mécanique	0548
Génie nuclégire	0552
Ingénierie des systèmes	0790
Mécanique novale	0547
Métalluraie	0743
Science des matériaux	0794
Technique du nétrole	0765
Technique minière	0551
Techniques sanitaires et	.0001
municipales	0554
Technologie bydraulique	0545
Mécanique appliquée	0346
Géotechnologie	0428
Matières plastiques	0420
(Technologia)	0705
Pacharcha opérationnalla	0706
Toxtilos et tissus (Teshnologia)	0704
revines et ussos (recutiologie)	0774
PSYCHOLOGIE	
Généralités	0421
	0021

zeneralites	.062
ersonnalité	.062
sychobiologie	.034
sýchologie clinique	.062
sychologie du comportement	.038
sýchologie du développement .	.062
sýchologie expérimentale	.062
sýchologie industrielle	.062
sychologie physiologique	.098
sýchologie sociale	.045
sychométrie	.063

F

THE UNIVERSITY OF CALGARY FACULTY OF GRADUATE STUDIES

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies for acceptance, a dissertation entitled "Existence, Uniqueness and Numerical Realization of Solutions for Thermistor Equations" submitted by Shijun Zhou in partial fulfillment of the requirements for the degree of Doctor of Philosophy.

Supervisor, Dr. D. R. Westbrook, Department of Mathematics and Statistics

Ser Aguarisala

Dr. B. D. Aggarwala, Department of Mathematics and Statistics

halune

Dr. R. S. Dhaliwal, Department of Mathematics and Statistics

Dr. J. Rokne, Department of Computer Sciences

mmon

External Examiner Dr. P.H. Sammon Computer Modelling Group

Date_Nov 8

ABSTRACT

A coupled pair of nonlinear parabolic partial differential equations, which describe two dimensional heat and current distribution within a thermistor, is reviewed and discussed. For certain boundary conditions two nonlinear parabolic one (space) dimensional differential equations and their corresponding steady state differential equations with different properties are derived from these equations. As the two kinds of partial differential equations have different properties, which are called PTC and NTC respectively, they are discussed separately. Several different methods are used for the theoretical study of the equations. One method is to transform the differential equations into integral equations, through the proof of existence and uniqueness of integral equations, hence existence and uniqueness are obtained. Another method is to change the variables and to change the original boundary value problems to initial value problems. By proving existence and uniqueness for the initial value problems, existence and uniqueness are obtained again. The third one is a monotone method. By using the concept of upper and lower solution, existence and uniqueness (if applicable) are proved. Since the property of the Joule heating function for NTC problems is quite different from that for PTC problems, only existence is obtained and under some special meaning a uniqueness is also obtained. Many numerical experiments have been done. Numerical results are listed in tables and demonstrated by figures. The common property for time dependent problems with an external circuit is that the solutions have a surge. For some unstable solutions, a brief stability analysis is given.

Acknowledgements

I would like to express deepest appreciation to my supervisor Dr. D.R. Westbrook: for his encouraging me to research on the present direction; for his keeping me on the right path and then seriously considering my results and checking every detail, including my Fortran programs and English; for his unfailing support all the time and help in every small aspect; for his giving me the push needed to finish a dissertation; for his financial assistance from an NSERC operating grant and his confidence in me. Without his support, the dissertation would not have been possible.

Thanks to D.R. Westbrook, R.S. Dhaliwal, B.D. Aggarwala, L.B. Bos and C. Nasim. When I attended their lectures, I learnt lots of new materials from them. Thanks to B.D. Aggarwala for his partial financial support from an NSERC operating grant. Thanks to those who let me talk to them about this work, and to all those who gave other help or encouragement.

I also would like to thank the Department of Mathematics and Statistics and the Faculty of Graduate Studies for both the financial and emotional support which enabled me to thrive as a graduate student.

Thanks to my wife for her support during my survival as a graduate student in the University of Calgary.

CONTENTS

Approval page ii
Abstract iii
Acknowledgementsiv
List of Tables
List of Figures
Chapter 1. Introduction1
Chapter 2. Preliminaries and Historical Remarks
2.1 Basic Definitions and Conventions5
2.2 Development and Results12
2.3 Application
Chapter 3. Existence and Uniqueness
3.1 Steady State Problem for PTC19
3.2 Steady State Problem for NTC36
3.3 Existence for Both PTC and NTC
3.4 Existence of Solutions for the PTC and NTC Problems with
External Circuit42
3.5 Monotone Method for the PTC and NTC Problems46
3.6 Time Dependent Problem
Chapter 4. The Numerical Results
4.1 Steady State Problem for PTC

4.1.1 Without External Circuit54
4.1.2 With External Circuit61
4.2 Time Dependent Problem for PTC68
4.2.1 Moving Meshpoints Method69
4.2.2 Fixed Meshpoints Method80
4.3 Numerical Solutions for NTC Problem
4.3.1 Without External Circuit86
4.3.2 With External Circuit88
4.4 Time Dependent Problem for NTC94
4.5 Numerical Solutions for Smooth Functions
4.5.1 Smooth Function for PTC Problem
4.5.2 Smooth Function for NTC Problem
4.6 Time Dependent Problems for Smooth Functions106
4.6.1 Smooth Function for PTC Problem106
4.6.2 Smooth Function for NTC Problem109
4.7 Conclusions for the Numerical Results111
Chapter 5. Convergence and Stability Analysis
5.1 Linear Stability Analysis
5.2 The Numerical Results for Stability And Stability Analysis 120
Chapter 6. Conclusion
Bibliography132
Appendix A

.

473

.

List of Tables

.

Table 4.1.1	
Table 4.1.2	
Table 4.2.1	
Table 4.2.2	
Table 4.2.3	
Table 4.2.4	
Table 4.3.1	
Table 4.5.1	
Table 4.5.2	
Table 4.5.3	
Table 4.6.1	
Table 4.6.2	
Table 4.6.3	
Table 4.6.4	
Table 5.1.1	
Table 5.1.2	
Table 5.1.3	

List of Figures

•

Figure 2.2.1
Figure 2.2.2
Figure 2.2.3
Figure 4.1.0
Figure 4.1.0a
Figure 4.1.0b
Figure 4.1.1
Figure 4.1.2
Figure 4.1.3
Figure 4.3.0
Figure 4.3.1
Figure 4.3.2
Figure 4.3.3
Figure 4.3.4
Figure 4.5.1
Figure 4.5.2
Figure 4.5.3
Figure 4.5.4
Figure 4.5.5
Figure 4.7.1

.

Figure 4.7.2	
Figure 4.7.3	
Figure 4.7.4	

.

.

•

.

.

•

Chapter 1

Introduction

In this dissertation, attention is focused on a system of nonlinear parabolic (heat diffusion) differential equations which model the thermistor, an electrical device which is widely used for surge protection, etc. A review of results for the problem is given in chapter two. By considering different boundary conditions two different one (space) dimensional problems are derived. One problem is called PTC problem and the other the NTC problem. This dissertation is devoted to both numerical solutions and existence of theoretical solutions of these problems as they are related to thermistors.

In recent years an active branch of numerical solutions for nonlinear partial differential equations is in the solution of problems with moving boundaries. To solve such problems, a moving finite element method [23, 24, 25, 26, 27, 28] may be used. According to M.J. Baines [27, 28], this method was invented by K. Miller [25] who used it to approximate the solution of diffusion problems with steep moving fronts. The main idea is that the meshpoints are allowed to move according to the conditions at the interface points (or curve). The governing equations at the moving meshpoints are adjusted and the meshsize changes correspondingly. One possible application of moving boundary problems is to the problem of the thermistor.

A typical mathematical description of heat distribution within the thermistor is given by nonlinear parabolic equations. Its corresponding steady state equation is a nonlinear elliptic equation. The boundary conditions considered here are Robin boundary conditions.

X. Chen and A. Friedman [37, 38, 39], by using the concept of weak solution, proved the existence of a solution for this problem. They also use a conformal mapping method to prove the existence and uniqueness. H. Xie and W. Allegretto [40, 41] discussed the existence of solutions under certain assumptions in which $\sigma(u)$ and k(u) take some special forms. D.R. Westbrook [1] gave some numerical methods for obtaining approximate solutions of some steady state problems. More recently A.C. Fowler, I. Frigaard and S.D. Howison[3] have used perturbation and numerical methods to examine a one dimensional time dependent problem which is a special case of the general problem (this is the problem which is here called the NTC problem).

Here we also concentrate on one dimensional equations. Numerical experiments are described and some theoretical proofs of existence and uniqueness of solutions have been studied.

In § 3.1, § 3.2 and § 3.3, steady state problems have been considered. They are divided into two cases, NTC (negative temperature characteristic) and PTC (positive temperature characteristic, both to be defined in § 2.1), which are quite different. The PTC problem has a unique solution for given parameters and bound-ary conditions but the NTC problem may have more than one solution for some given parameters and boundary conditions. Therefore, different methods are needed to prove the existence (also uniqueness if applicable). We define an operator to transform the differential equations into integral equations and then prove the equivalence of the differential and integral equations. Though the operator itself is not a contraction operator, it can be proved that a power of that operator is a contraction

operator. Therefore, by using the fixed point principle, there is a fixed point for that operator. As has been mentioned, there are more than one solution for NTC, so that uniqueness is only proved for the PTC problem. Although the uniqueness is lost for NTC problems, the uniqueness is still true in some other sense (refer § 3.3). The existence, uniqueness and nonuniqueness have also been studied both for NTC and PTC problems when an external circuit is connected.

In chapter 4, many numerical experiments have been done. For the steady state situation, numerical results coincide very well with the theoretical ones. Approximate relations of parameters are also given for one, two and three solutions. For time dependent problems, two methods are used. One is a moving mesh finite element method, the other is for fixed meshpoints. Numerical solutions by both methods converge to the numerical solutions of the steady state problems. If external circuitry is connected, for some parameters, there are more than one steady state solution, one of which is numerically unstable. The details and numerical experiments are given in chapter 4. The comparison of numerical results obtained by both methods has been made. Several figures have been given to demonstrate the results.

In chapter 5, a brief review of stability and instability is given. The analysis of stability and instability is based on perturbation theory. It was found that all the steady state solutions could be obtained numerically, thus theoretically, the corresponding solutions of time dependent problems should converge to one of the solutions of the steady state problems. However, the numerical experiments show that for some steady state solutions there are no corresponding solutions of time dependent problems which converge to these steady state solutions. Hence, there is a need to discuss the stability of these solutions. The method is to investigate the solutions for time dependent problems close to the corresponding solutions for steady state problems. If some increase rapidly in the neighborhood of solutions of the steady state problems, the corresponding solutions for time dependent problems are said to be unstable, otherwise they are stable. As the exact solutions are not explicit expressions, numerical estimates of the tendency in the neighborhood of the solutions are obtained. The numerical results coincide with and demonstrate the conclusion about the stability and instability.

Chapter 2

Preliminaries and Historical Remarks

In this chapter, we review the basic idea of the thermistor problem. In § 2.1, the basic definitions and conventions are given. From the same basic equation, two different kinds of one dimensional equation are derived under certain boundary conditions. One has the PTC property and the other has NTC property. Hence thereafter, one is called the PTC problem and the other is called the NTC problem. Most of the recent results have been listed in § 2.2. It can be seen that the most of them require special conditions. For time dependent problems, the nonlinear Joule heating terms are always assumed to be monotone, smooth with bounded derivatives for existence and uniqueness of solutions. A schematic representation of the thermistor is given in § 2.3.

2.1 Basic Definitions and Conventions

The equations here describe heat and current distribution within a thermistor. The thermistor is an electrical device made of ceramic material that can be used as a current surge regulator[1]. In appearance [3], this is a cylinder of typical radius 5mm and typical thickness 2mm, connected into its circuit via wires soldered to the top and bottom; these surfaces are covered with a thin conducting sheet of metal acting as contact. (Figure 2.2.1 is a schematic view of the thermistor). The basis for its performance is its temperature-dependent electrical resistivity, which varies strongly with temperature, increasing by about five orders of magnitude over a temperature range of 100-200°C (see Figure 2.2.2).

There are two kinds of thermistors:[3] negative temperature characteristic (NTC) thermistors, whose electrical conductivity σ increases with temperature, and positive temperature characteristic (PTC) thermistors, for which σ decreases with temperature. (Here positive and negative refer to materials whose resistivity is an increasing or nonincreasing function of temperature respectively; conductivity is the inverse of resistivity.)





A Thermistor

The general equations to describe the heat distribution within the thermistor,

after some scaling of variables, are as follows (in two dimensions):

$$\begin{cases} \nabla(\sigma(u)\nabla\phi) = 0 & , -a < x < a, -b < y < b \\ \frac{\partial u}{\partial t} = \nabla(k(u)\nabla u) + \gamma\sigma(u)|\nabla\phi|^2 & , \frac{\partial u}{\partial n} + \beta u|_{\partial\Omega} = 0 \end{cases}$$
(2.1)

This is a time dependent problem with the two dimensional geometry taken as Cartesian rather than axisymmetric (It is felt that this will not lead to any qualitative



Figure 2.2.2

Conductivity as A Function of Temperature

differences in the results). The corresponding steady state problem is as following:

$$\begin{cases} \nabla(\sigma(u)\nabla\phi) = 0 &, -a < x < a, -b < y < b \\ \nabla(k(u)\nabla u) + \gamma\sigma(u)|\nabla\phi|^2 = 0 &, \frac{\partial u}{\partial n} + \beta u|_{\partial\Omega} = 0 \end{cases}$$
(2.2)

where ∇ is a gradient operator, Δ is Laplacian operator, $\Omega = [-a, a] \times [-b, b]$, u is a scaled temperature, ϕ is a scaled electrical potential, γ and β are dimensionless





Thermistor With External Circuit

parameters, k(u) the thermal conductivity is assumed to be constant and the electrical conductivity $\sigma(u)$ is a function of u which changes rapidly between u = 1 and u = 2 and is a constant when u < 1 or u > 2. The Figure 2.2.2 is a graph of such a function. Without loss of generality, it can be assumed that b and k(u) are equal to 1. The boundary conditions for the potential ϕ are

$$\phi = \pm v_0 \quad \text{on} \quad y = \pm 1 \tag{2.3}$$

and

$$\frac{\partial \phi}{\partial n} = \frac{\partial \phi}{\partial x} = 0 \quad \text{on} \quad x = \pm a$$
 (2.4)

where v_0 is an unknown constant.

Now consider the case where $\beta = 0$ on x = a (i.e., on side A of Figure 2.2.1). Since $\frac{\partial u}{\partial x} = 0$ on x = a, it is possible for u to be a function of y only. Similarly, $\frac{\partial \phi}{\partial x} = 0$ on x = a and ϕ may also be a function of y only. Thus from the first equation of Eq.(2.1) (or (2.2)),

$$\frac{\partial}{\partial y}(\sigma(u)\frac{\partial \phi}{\partial y}) = 0$$

Since $\sigma(u) \not\equiv 0$

$$rac{\partial \phi}{\partial y} = rac{C}{\sigma(u)}$$

where C is a constant. Since the average current I is defined by

$$I = \frac{1}{a} \int_0^a \sigma(u) \frac{\partial \phi}{\partial y}|_{y=1} dx = \frac{1}{a} \int_0^a C dx = C$$

Thus $\frac{\partial \phi}{\partial y} = \frac{I}{\sigma(u)}$. By symmetry, $\frac{\partial u}{\partial y} = 0$ on y = 0. So Eq.(2.1) and Eq.(2.2) become

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial y^2} + \frac{\gamma I^2}{\sigma(u)} &, 0 < y < 1\\ u_y(0) = 0 &, u_y(1) + \beta u(1) = 0 \end{cases}$$

$$(2.5)$$

and for the steady state

$$\begin{cases} \frac{d^2u}{dy^2} + \frac{\gamma I^2}{\sigma(u)} = 0 &, 0 < y < 1\\ u_y(0) = 0 &, u_y(1) + \beta u(1) = 0 \end{cases}$$
(2.6)

This problem is hereafter labelled NTC.

Now consider the case where $\beta = 0$ on y = 1 (i.e., on the top of Figure 2.2.1). Since $\frac{\partial u}{\partial y} = 0$ on y = 1, the possible solution is that u is only dependent on x, therefore, $\sigma(u)$ is only dependent upon x while ϕ is still a function of y only. From the first equation of Eq.(2.1) (or Eq.(2.2))

$$\frac{\partial}{\partial y}(\sigma(u)\frac{\partial \phi}{\partial y})=0$$

i.e.,

$$\sigma(u)\frac{d^2\phi}{dy^2} = 0$$

Since $\sigma(u) \not\equiv 0$

$$\frac{d\phi}{dy} = C$$

where C is constant. From (2.3), it is easy to see that $\phi = v_0 y$, i.e., $C = v_0$. By symmetry, $\frac{\partial u}{\partial x} = 0$ on x = 0. So Eq.(2.1) and Eq.(2.2) become

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \gamma v_0^2 \sigma(u) &, 0 < x < a \\ u_x(0) = 0 &, u_x(a) + \beta u(a) = 0 \end{cases}$$
(2.7)

 and

$$\begin{cases} \frac{d^2u}{dx^2} + \gamma v_0^2 \sigma(u) = 0 , 0 < x < a \\ u_x(0) = 0 , u_x(a) + \beta u(a) = 0 \end{cases}$$
(2.8)

Having separated the two cases there is no loss in taking a = 1. This case will be labelled the PTC problem.

For Eq.(2.5), Eq.(2.6), Eq.(2.7) and Eq.(2.8), the external circuit is not considered. If the external circuit is connected as in Figure 2.2.3, then

$$V = 2V_0 + IR_0$$

Rescale it as

$$1 = v_0 + I\mu \tag{2.9}$$

where $v_0 = \frac{2V_0}{V}$ (unknown) and $\mu = \frac{R_0}{V}$ (parameter). For Eq.(2.5) and Eq.(2.6), there is

$$v_0 = \phi(1) = \int_0^1 \frac{\partial \phi}{\partial y} dy = \int_0^1 \frac{I}{\sigma(u)} dy = I \int_0^1 \frac{1}{\sigma(u)} dy$$

From (2.9), there is

$$1 = I \int_0^1 \frac{1}{\sigma(u)} dy + I\mu$$

thus

$$I = \frac{1}{\mu + \int_0^1 \frac{1}{\sigma(u)} dy}$$

Therefore, Eq.(2.5) and Eq.(2.6) become

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial y^2} + \frac{\gamma}{(\mu + \int_0^1 \frac{1}{\sigma(u)} dy)^2 \sigma(u)} &, 0 < y < 1\\ u_y(0) = 0 &, u_y(1) + \beta u(1) = 0 \end{cases}$$
(2.10)

 and

$$\frac{d^2 u}{dy^2} + \frac{\gamma}{(\mu + \int_0^1 \frac{1}{\sigma(u)} dy)^2 \sigma(u)} = 0 \quad , 0 < y < 1$$

$$u_y(0) = 0 \qquad , u_y(1) + \beta u(1) = 0 \qquad (2.11)$$

For Eq.(2.7) and Eq.(2.8), there is $\phi = v_0 y$, hence the current is

$$I = \int_0^1 \sigma(u) \frac{\partial \phi}{\partial y}|_{y=1} dx = v_0 \int_0^1 \sigma(u) dx$$

Similarly, from (2.9), there is

$$1 = v_0 + \mu v_0 \int_0^1 \sigma(u) dx$$

hence

$$v_0 = \frac{1}{1 + \mu \int_0^1 \sigma(u) dx}$$

Therefore, Eq.(2.7) and Eq.(2.8) become

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \frac{\gamma \sigma(u)}{(1+\mu \int_0^1 \sigma(u) dx)^2} &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$
(2.12)

 and

$$\begin{cases} \frac{d^2u}{dx^2} + \frac{\gamma\sigma(u)}{(1+\mu\int_0^1\sigma(u)dx)^2} = 0 &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$
(2.13)

In most of this dissertation, $\sigma(u)$ is defined as

$$\sigma(u) = \begin{cases} 1 & , u < 1 \\ e^{-10(u-1)} & , 1 \le u \le 2 \\ e^{-10} & , 2 < u \end{cases}$$
(2.14)

This follows the definition used by Fowler et al [3] and has the appropriate behavior. Thus by the definition of $\sigma(u)$ Eq.(2.5), Eq.(2.6), Eq.(2.10) and Eq.(2.11) have NTC property, Eq.(2.7), Eq.(2.8), Eq.(2.12) and Eq.(2.13) have the PTC property. Hereafter they are referred as PTC or NTC problems respectively. The difference for these two problems is that the nonlinear term in one is the reciprocal of that in the other. Therefore, the nonlinear function is monotonically nonincreasing for the PTC problem and the nonlinear function is monotonically increasing for the NTC problem. This difference leads to quite different properties for the solution of these problems.

2.2 Development and Results

Several papers have been written in this area. Most considered the steady state

equations (in two dimensions) as follows

$$\begin{cases} \nabla(\sigma(u)\nabla\phi) = 0 &, -a < x < a, -b < y < b \\ \nabla(k(u)\nabla u) + \alpha\sigma(u)|\nabla\phi|^2 = 0 &, \frac{\partial u}{\partial n} + \beta u|_{\partial\Omega} = 0, \phi|_{y=\pm b} = \pm\phi_0, \frac{\partial \phi}{\partial n}|_{x=\pm a} = 0 \\ \end{cases}$$
(2.15)

where Δ is Laplacian operator, $\Delta = \nabla^2$, $\Omega = [-a, a] \times [-b, b]$, u is temperature, ϕ is electrical potential, α and β are dimensionless parameters, k(u) is thermal conductivity and electrical conductivity $\sigma(u)$ is defined by

$$\sigma(u) = \begin{cases} 1 & , u \le 1 \\ \delta & , u > 1 \end{cases}$$
(2.16)

where δ is very small real positive number. For time dependent problems,

$$\begin{cases} \nabla(\sigma(u)\nabla\phi) = 0 & , -a < x < a & , -b < y < b \\ \frac{\partial u}{\partial t} = \nabla(k(u)\nabla u) + \alpha\sigma(u)|\nabla\phi|^2 & , \frac{\partial u}{\partial n} + \beta u|_{\partial\Omega} = 0 & , \phi|_{y=\pm b} = \pm\phi_0, \frac{\partial\phi}{\partial n}|_{x=\pm a} = 0 \end{cases}$$

$$(2.17)$$

where t > 0.

For the steady state, X. Chen and A. Friedman [37, 38, 39], G. Cimatti [35, 36], H. Xie and W. Allegretto [41], etc., use the transformation

$$\theta = \frac{1}{2}\phi^2 + \int_{u_0}^u \frac{k(s)}{\sigma(s)} ds$$
 (2.18)

(where ϕ is electrical potential, u is temperature, k and σ are thermal and electrical conductivities, u_0 is a constant) and give existence and uniqueness proofs for the problem.

If we denote

$$F(u) = \int_{u_0}^u \frac{k(s)}{\sigma(s)} ds$$

G. Cimatti, in [35, 36], proved that, if F(u) is bounded as $u \to \infty$, there are two cases, i.e., either there is only one solution for some boundary condition or there is no solution at all for other boundary conditions. Also, in [35], G. Cimatti proved that, if some more restrictions (e.g., bounded, continuous) on k(s) and $\sigma(s)$ are given, the solution for Eq.(2.15) is bounded.

In [37], X. Chen and A. Friedman discussed the case in which

$$\sigma(u) \begin{cases} > 0 , u \le u^* \\ = 0 , u > u^* \end{cases}$$
(2.19)

where u^* is critical temperature and $\sigma(u)$ is continuous at u^* . They construct a continuous and infinitely differentiable function $\sigma_{\epsilon}(u)$ to approximate $\sigma(u)$. Then using a transformation similar to (2.18), they review the existence proof of a solution $(\phi_{\epsilon}, u_{\epsilon})$. After introducing the concept of a weak solution, it is proved that $(\phi_{\epsilon}, u_{\epsilon})$ converges to a weak solution (ϕ, u) of Eq.(2.15). and $u \leq u^*$. Also, in [38], a more specific situation, i.e.,

$$\sigma(u) = \left\{egin{array}{cc} 1 & , u \leq u^* \ 0 & , u > u^* \end{array}
ight.$$

is considered. A transformation similar to (2.18) and a conformal mapping are used to prove the existence and uniqueness.

H. Xie and W. Allegretto [41], instead of discussing the Robin boundary condition in Eq.(2.15), discuss the Dirichlet boundary value problem. Under the hypothesis of that

$$\sigma(u) = C_1 u^{\iota} \exp(-C_2/C_3 u)$$

and

$$k(u) = (C_4 + C_5 u + C_6 u^2)^{-1}$$

where C_1, C_2, C_3, C_4, C_5 and C_6 are physical positive constants and ι is a small positive number or non positive number, they proved that there exists at least one solution to Eq.(2.15).

For the time dependent problem (2.17), there is not as much literature as there is for steady state problem. H. Xie and W. Allegretto [40], F.B. Weissler [33], H. Fujita [34] have done some research in the area of classic nonlinear parabolic equations similar to Eq.(2.17). However, F.B. Weissler and H. Fujita mainly contributed to the blow up problem, though the form of equations is the same, the properties of the nonlinear part are quite different. Here we put the emphasis on the case where the equations describe the heat distribution within thermistors. In [40], H. Xie and W. Allegretto assumed that

$$\sigma(u) = C_1 u^{-\iota} \exp(-C_2/u), \quad 0 < \iota < 1, \quad C_1, C_2 > 0$$

and

$$k(u) = (C_3 + C_4 u + C_5 u^2)^{-1}, \quad C_3, C_4, C_5 \ge 0, \quad C_3 + C_4 + C_5 > 0$$

Additionally, they replaced the first equation of Eq.(2.17) by the simple equation $(\sigma(u)\phi_{x_1})_{x_1} = 0$, i.e., the derivative is taken with respect to only one spatial variable and assumed $\sigma(s)$ and k(s) locally Lipschitz continuous and bounded. Then they integrate the first equation of Eq.(2.17), change Eq.(2.17) into an integro-differential equation. After introducing several functional spaces and a variational form, an operator is determined by the variational form. By proving that there exists at least one fixed point for that operator, then the existence of a solution has been proved. For Eq.(2.17), it seems that it is quite difficult to obtain a general conclusion.

All of the above results are theoretical ones. For numerical methods to get nu-

merical solutions for Eq.(2.15) and Eq.(2.17), not many papers can be found. D.R. Westbrook [1] gave some good ideas for the numerical methods. A new function is constructed and then an iterative method is used. The method overcomes the difficulty to determine the point at which the temperature u = 1 (after scaling) and lots of numerical experiments are done for different parameters α and β . Also, A.C. Fowler, I. Frigaard and S.D. Howison [3] gave a perturbation analysis and a numerical method for the one dimensional time dependent NTC problem. They use equal stepsize for spatial direction and different time stepsize for time direction. From the numerical results, temperature surges appeared at some time point. They also noticed that there exist possible multiple solutions.

In order to obtain numerical solutions for equations similar to Eq.(2.17), the possibility of a moving finite element method [20, 21, 23, 24, 25, 26, 27, 28] is also raised. In this method, the nodes are allowed to change positions with time. The spatial stepsizes are adjusted at every time step. The method is more accurate in some cases.

2.3 Application

There are many applications of thermistors. One of them is as a fuse. In the circuit of Figure 2.2.3, a short circuit is represented by closing the switch, causing a current surge driven by the external voltage V to pass through the circuit resistance R_0 and the thermistor, thereby heating it. The consequent decrease in the electrical conductivity causes the current to fall until equilibrium is reached, with all the heat

generated within the thermistor being lost to its surroundings. In a well-designed thermistor, the final current should be a small fraction of the initial surge.

Chapter 3

Existence and Uniqueness

In this chapter existence and uniqueness are discussed. For convenience the steady state problems are discussed first and then the time dependent problems. In § 3.1, the steady state problem for PTC thermistor has been studied. Before the proof of the existence and uniqueness, the properties of the solutions are thoroughly discussed. An integral operator is introduced so that the original differential equation is transformed into an equivalent operator equation. Since the operator equation has a solution, thus the existence of solution for the differential equation is proved. Using the nonincreasing property of f(u), uniqueness is obtained. In § 3.2, the NTC problem has been discussed. The method is similar to that used in § 3.1. Since in this case, g(u) is not a nonincreasing function, uniqueness is lost. In fact, there are three solutions for some parameters α and β . In § 3.3, a different method is used to prove the existence of solutions. The original boundary problems have been changed into initial value problems. Since the solutions for the initial value problem exist, then the existence of solutions for the boundary value problems is obtained. Here only the monotonic property is used, thus the method is good for both PTC and NTC problems. Though the uniqueness of the original boundary value problem is lost, it can be still proved that if initial value u_0 and β are given, then there is a unique α such that $u(x, u_0, \beta, \alpha)$, a solution for initial value problem, is a unique solution for the original PTC or NTC problem. Since only the monotonic property is used in all proofs in § 3.3, the conclusion can be generalized to all monotonic functions. At the

end of § 3.3, the situation when external circuitry is connected is also discussed. The relations of parameters for which there are one, two or three solutions are given. In § 3.5, a more general monotone method is reviewed. As in § 3.3, this method can be used to prove the existence for both NTC and PTC problems. In § 3.6, the last section of this chapter, existence for time dependent problems has been proved.

3.1 Steady State Problem for PTC

In this section the existence and uniqueness of solutions of the following problem are discussed.

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha f(u) = 0 & , 0 < x < 1\\ u_x(0) = 0 & , u_x(1) + \beta u(1) = 0 \end{cases}$$
(3.1)

and

$$f(u) = \begin{cases} 1 & , u < 1 \\ e^{-10(u-1)} & , 1 \le u \le 2 \\ e^{-10} & , 2 < u \end{cases}$$
(3.2)

where α and β are parameters.

Theorem 3.1.1 For Eq. (3.1) and Eq. (3.2), if $u(x) \le 1.0$ for $0 \le x \le 1$, then

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)$$

is the unique solution. Similarly, if $u(x) \ge 2$ for $0 \le x \le 1$, then

$$u(x) = \frac{\alpha e^{-10}}{\beta} + \frac{\alpha e^{-10}}{2}(1 - x^2)$$

is the unique solution.

Proof: For $u(x) \leq 1$, f(u(x)) = 1, from the first equation of Eq.(3.1) then

$$u(x) = a + bx + cx^2$$

Using the boundary conditions, hence

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)$$

The uniqueness is trivial since when $u(x) \leq 1$, Eq.(3.1) is a linear differential equation.

For $u(x) \ge 2$, the proof is similar.

Property 3.1.1 Suppose the solution of Eq.(3.1) exists and has second derivative, then u(x) is monotonically nonincreasing within the interval [0,1], and also $u(1) \ge 0$.

Proof: Integrate the first equation of Eq. (3.1) from 0 to x, we have

$$u_x(x) + \alpha \int_0^x f(u)dx = 0 \tag{3.3}$$

Since f(u) > 0, the integral must be positive. That means $u_x(x) < 0$, hence u(x) is strictly monotonically nonincreasing.

As $u_x(1) \leq 0$ and $\beta > 0$, u(1) can not be negative because of the boundary condition of Eq.(3.1). #

Corollary 3.1.1 If
$$u(1)$$
 is zero, then $u(x) \equiv 0$. #

This conclusion is trivial. Since $u_{xx}(x)$ is negative, $u_x(x)$ is monotonically nonincreasing. As $u_x(0) = 0$ and $u_x(0) \ge u_x(x) \ge u_x(1) = 0$, hence $u_x(x) \equiv 0$, thus $u(x) \equiv u(1) = 0$ (only possible if $\alpha = 0$).

Theorem 3.1.2 $u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)$ is a solution of Eq.(3.1) iff $\alpha \leq \frac{2\beta}{\beta+2}$ (i.e.,

٢

 $u(1) \leq \frac{2}{\beta+2}$) and $u(x) = \frac{\alpha e^{-10}}{\beta} + \frac{\alpha e^{-10}}{2}(1-x^2)$ is a solution of Eq.(3.1) iff $\alpha \geq 2\beta e^{10}$ (i.e., $u(1) \geq 2$).

Proof: Here the proof is only given for $\alpha \leq \frac{2\beta}{\beta+2}$. If $\alpha \geq 2\beta e^{10}$, the method is the same.

Necessity. If $u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)$ is a solution, then it must satisfy Eq.(3.1), hence

$$-\alpha + \alpha f(\frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)) = 0$$

i.e.,

$$f(\frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)) = 1$$

so by the property of f,

$$\frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2) \le 1$$

Since u(x) is nonincreasing, $u(x) \leq 1$ iff $\frac{\alpha}{\beta} + \frac{\alpha}{2} \leq 1$ for $0 \leq x \leq 1$, therefore $\alpha \leq \frac{2\beta}{\beta+2}$ Sufficiency. If $\alpha \leq \frac{2\beta}{\beta+2}$, construct

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)$$

hence $u(x) \leq \frac{\alpha}{\beta} + \frac{\alpha}{2} \leq 1$, and u(x) satisfy Eq.(3.1).

The more interesting problem is when $1 \le u(x) \le 2$ for some $x \in [0, 1]$. Does the solution exist? Is it unique? To answer those questions, we need to introduce more concepts and results. First, we construct an integral operator from Eq.(3.1),

$$Tu = u_1[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x)dt$$
(3.4)

where u_1 is an arbitrary real number. Consider the initial value problem

$$\begin{array}{rcl}
 u &= Tu \\
 u(1) &= u_1 \\
 u'(1) &= -\beta u_1
\end{array}$$
(3.5)

Actually, if u(x) = Tu(x), then $u(1) = u_1$, and since $u'(x) = -\beta u_1 + \alpha \int_x^1 f(u(t))dt$, hence $u'(1) = -\beta u_1$. Therefore the problem is to find u such that

$$u = Tu$$

Before answering that question, the properties of T are discussed. In the following, when it is said $u(x) \leq v(x)$, it means that in the sense of that for every $x \in [0,1]$, $u(x) \leq v(x)$.

Property 3.1.2 T is a monotonically nondecreasing operator.

Proof: If $u(x) \leq v(x)$, since f is a nonincreasing function, then

 $f(u(t)) \ge f(v(t))$

$$(t-x)f(u(t)) \ge (t-x)f(v(t)) \quad \text{if } t \ge x$$

$$\alpha \int_x^1 (t-x)f(u(t))dt \ge \alpha \int_x^1 (t-x)f(v(t))dt$$

$$-\alpha \int_x^1 (t-x)f(u(t))dt \le -\alpha \int_x^1 (t-x)f(v(t))dt$$

$$Tu = u_1[1+\beta(1-x)] - \alpha \int_x^1 f(u(t))(t-x)dt$$

$$\le u_1[1+\beta(1-x)] - \alpha \int_x^1 f(v(t))(t-x)dt$$

$$= Tv$$

Thus T is a monotonically nondecreasing operator.

Property 3.1.3 If u(x) is the cold (then $u_1 = \frac{\alpha}{\beta}$) or hot (then $u_1 = \frac{\alpha e^{-10}}{\beta}$) solution of Eq.(3.1), then

$$u(x) = Tu(x)$$

Proof: Since u(x) is the cold solution of Eq.(3.1), then $u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)$ and f(u(t)) = 1,

$$Tu = \frac{\alpha}{\beta} [1 + \beta(1 - x)] - \alpha \int_x^1 (t - x) dt$$
$$= \frac{\alpha}{\beta} [1 + \beta(1 - x)] - \alpha \frac{1}{2} (t - x)^2 |_x^1$$
$$= \frac{\alpha}{\beta} + \frac{\alpha}{2} (1 - x^2)$$
$$= u(x)$$

The result follows.

Property 3.1.4 If u(x) is a solution of Eq.(3.1), then u(x) is a solution of Eq.(3.5) where $u_1 = u(1)$.

Proof: Let u(x) be a solution of Eq.(3.1), change the variable in Eq.(3.1) into t, multiply the first equation with t - x and integrate from x to 1, then

$$\int_{x}^{1} u_{tt}(t)(t-x)dt + \alpha \int_{x}^{1} f(u(t))(t-x)dt = 0$$

Use integration by part for the first term of the above equation, thus

$$u_t(t)(t-x)|_x^1 - \int_x^1 u_t(t)dt + \alpha \int_x^1 f(u(t))(t-x)dt = 0$$

so

$$u_t(1)(1-x) - u(t)|_x^1 + \alpha \int_x^1 f(u(t))(t-x)dt = 0$$

by the boundary conditions of Eq.(3.1), then

$$-\beta u(1)(1-x) - u(1) + u(x) + \alpha \int_x^1 f(u(t))(t-x)dt = 0$$

i.e.,

$$u(x) = u(1)[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x)dt$$

that is

$$u(x) = Tu(x)$$

The proof is completed.

Definition 3.1.1 Let (X, ρ) a metric space, A be an operator from X to X, if $\exists \eta, 0 \leq \eta < 1$ such that $\forall x, y \in X$, there is

$$\rho(Ax, Ay) \le \eta \rho(x, y) \tag{3.6}$$

then A is said to be a contraction operator on X.

Lemma 3.1.1 [32] (Banach, fixed point principle) If B is a Banach space, A is a contraction operator on B, there exists one and only one $x^* \in B$ such that

$$x^* = Ax^*$$

#

Lemma 3.1.2 Let B be a Banach space, A be an operator from B to B. If there exists a natural number n such that A^n is a contraction operator on B, then there must exist one and only one fixed point for A.

Obviously, if n = 1, this lemma is just lemma 3.1.1.

Proof: Let $K = A^n$, then K is a contraction operator on B, hence by lemma 3.1.1 there exists a fixed point $x^* \in B$: $x^* = Kx^*$. Now we say that x^* is a fixed
point for A. In fact,

$$AK = AA^n = A^{n+1} = A^n A = KA$$

therefore $K(Ax^*) = A(Kx^*) = Ax^*$, so Ax^* is also a fixed point for K. Since there exists one and only one fixed point for K, thus $Ax^* = x^*$.

If x^1 is any fixed point of A, since $Ax^1 = x^1$, then

$$A^n x^1 = A^{n-1} x^1 = \dots = x^1$$

Thus x^1 is also a fixed point for $K = A^n$. Since there exists only one fixed point x^* for K, therefore $x^1 = x^*$. Then there exists one and only one fixed point for A. #

Now denote C[0,1] as a space of all continuous function defined on the interval [0,1]. Define a norm $||\cdot||$ on C[0,1] as

$$||f|| = \sup_{0 \le x \le 1} |f(x)| = \max_{0 \le x \le 1} |f(x)|$$
(3.7)

Then C[0,1] is a Banach space.

Theorem 3.1.3 Let T be defined on C[0,1] by (3.4), and f be defined by (3.2), then for any α and β , there exists one and only one continuous function $u(x) \in C[0,1]$ such that

$$u = Tu \quad (= u_1[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x)dt)$$
 (3.8)

where u_1 is an arbitrary real number.

Proof: Since Tu is defined as

$$Tu = u_1[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x)dt$$

For any u(x), $v(x) \in C[0,1]$, $x \in [0,1]$, by the mean value theorem and $|f'(*)| \leq 10$,

$$\begin{aligned} |Tu(x) - Tv(x)| &= |\alpha \int_{x}^{1} (f(v(t)) - f(u(t)))(t - x)dt| \\ &= |\alpha| \cdot |\int_{x}^{1} f'(*)(v(t) - u(t))(t - x)dt| \\ &\leq 10|\alpha| \cdot ||u - v|| \int_{x}^{1} (t - x)dt \\ &= \frac{10|\alpha|}{2} ||u - v||(1 - x)^{2} \end{aligned}$$
(3.9)

Now use induction to prove for $x \in [0, 1]$ that

$$|T^{n}u - T^{n}v| \le \frac{|\alpha|^{n} 10^{n}}{(2n)!} ||u - v|| (1 - x)^{2n}$$
(3.10)

For n = 2, use (3.9), by the mean value theorem and $|f'(*)| \le 10$,

$$\begin{aligned} |T^{2}u(x) - T^{2}v(x)| &= |\alpha \int_{x}^{1} (f(Tv(t)) - f(Tu(t)))(t - x)dt| \\ &= |\alpha| \cdot |\int_{x}^{1} f'(*)(Tv(t) - Tu(t))(t - x)dt| \\ &\leq \frac{10^{2}|\alpha|^{2}}{2} ||u - v|| \int_{x}^{1} (1 - t)^{2} (t - x)dt \\ &= \frac{10^{2}|\alpha|^{2}}{1 \cdot 2 \cdot 3 \cdot 4} ||u - v||(1 - x)^{4} \end{aligned}$$

So for n = 2 it is true. Suppose (3.10) is true for n, then

$$\begin{aligned} |T^{n+1}u(x) - T^{n+1}v(x)| &= |\alpha \int_x^1 (f(T^n v(t)) - f(T^n u(t)))(t - x)dt| \\ &= |\alpha| \cdot |\int_x^1 f'(*)(T^n v(t) - T^n u(t))(t - x)dt| \\ &\leq 10|\alpha| \int_x^1 |T^n u - T^n v|(t - x)dt \\ &\leq \frac{10^{n+1}|\alpha|^{n+1}}{(2n)!} ||u - v|| \int_x^1 (1 - t)^{2n} (t - x)dt \\ &= \frac{10^{n+1}|\alpha|^{n+1}}{(2n)!} \frac{||u - v||}{(2n+2)(2n+1)} (1 - x)^{2n+2} \\ &= \frac{10^{n+1}|\alpha|^{n+1}}{(2(n+1))!} ||u - v|| (1 - x)^{2n+2} \end{aligned}$$

Hence (3.10) is true. Now take a natural number n such that

$$\eta = \frac{\alpha^n 10^n}{(2n)!} < 1$$

then

$$||T^{n}u - T^{n}v|| = \sup_{0 \le x \le 1} |T^{n}u - T^{n}v| \le \eta ||u - v||$$

so by lemma 3.1.2, there exists one and only one solution in C[0,1].

#

If $u(x) \in C[0,1]$ is a solution of Eq.(3.8), then u(x) is also a solution of (3.5). In fact, the following is true.

Property 3.1.5 If $u(x) \in C[0,1]$ is a solution of Eq.(3.8), then u(x) is a solution of Eq.(3.5); furthermore it has a second derivative.

Proof: Since u(x) is a solution of Eq.(3.8), that is

$$u(x) = u_1[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x)dt$$

where f is defined by (3.2). Since f is continuous, thus the derivative of right hand side exists. Differentiate the above equation then

$$u'(x) = -\beta u_1 + \alpha \int_x^1 f(u(t))dt$$

Therefore, $u(1) = u_1$ and $u'(1) = -\beta u_1$, which means that u(x) is a solution of Eq.(3.5). Furthermore, from

$$u'(x) = -\beta u_1 + \alpha \int_x^1 f(u(t))dt$$

it is easy to see that the right hand side is still differentiable by the continuity of f. Differentiate both sides of above equation, we have

$$u''(x) = -\alpha f(u(x))$$

Therefore, u(x) has a second derivative.

Before we prove that if α and β are fixed there exists a u(x) for Eq.(3.1), a property of u(x) is discussed first. If u(x) is a solution of Eq.(3.5), it can be denoted as $u(x, u_1, \alpha, \beta)$ since u depends upon x, u_1 , α , β . From property 3.1.5, u(x) is differentiable, so

$$u_x(x, u_1, \alpha, \beta) = -\beta u_1 + \alpha \int_x^1 f(u(t, u_1, \alpha, \beta))dt$$
(3.11)

$$u_{xx}(x, u_1, \alpha, \beta) = -\alpha f(u(x, u_1, \alpha, \beta))$$
(3.12)

Thus from (3.11), if u(x) is a solution of (3.1), then

$$u_x(0, u_1, \alpha, \beta) = -\beta u_1 + \alpha \int_0^1 f(u(t, u_1, \alpha, \beta))dt = 0$$

so

$$u_1 = \frac{\alpha}{\beta} \int_0^1 f(u(t, u_1, \alpha, \beta)) dt$$

From the definition of f by (3.2), $e^{-10} \leq f(u) \leq 1$, thus a necessary condition for $u_x(0, u_1, \alpha, \beta) = 0$ is that

$$rac{lpha e^{-10}}{eta} \leq u_1 \leq rac{lpha}{eta}$$

This condition means for fixed α and β that Eq.(3.1) has a solution only if u_1 is chosen in that region. Now we will prove that if u_1 , β are fixed nonnegative numbers, there is an α such that $u(x, u_1, \alpha, \beta)$ is a solution of Eq.(3.1). That means the following.

Property 3.1.6 For any given positive number u_1 and β , there exists a unique α , such that $u(x, u_1, \alpha, \beta)$ satisfies Eq.(3.1).

Proof: For given positive u_1 and β , for any α , by theorem 3.1.3, there exists $u(x, u_1, \alpha, \beta)$ such that

$$u(x, u_1, \alpha, \beta) = Tu(x, u_1, \alpha, \beta)$$

#

however in general, $u_x(0, u_1, \alpha, \beta) \neq 0$. From (3.11),

$$u_x(0,u_1,0,\beta) = -\beta u_1 < 0$$

and also from (3.11)

$$u_x(0, u_1, \alpha, \beta) = -\beta u_1 + \alpha \int_0^1 f(u(t, u_1, \alpha, \beta))dt > -\beta u_1 + \alpha f(u_1)$$

by the nonincreasing property of f and u. Since $f(u_1) \ge e^{-10} > 0$ hence if α is chosen big enough, $u_x(0, u_1, \alpha, \beta) > 0$. By the continuity of $u_x(0, u_1, \alpha, \beta)$ with respect to α , there must exist an α^* such that $u_x(0, u_1, \alpha^*, \beta) = 0$. In order to prove the uniqueness, the strictly monotonic property of $u_x(0, u_1, \alpha, \beta)$ with respect to α is proved, hence uniqueness is obtained. Suppose there is another α^{**} such that $u_x(0, u_1, \alpha^{**}, \beta) = 0$. Without losing generality, suppose $\alpha^{**} > \alpha^*$. Since both $u(x, u_1, \alpha^*, \beta)$ and $u(x, u_1, \alpha^{**}, \beta)$ are solutions of Eq.(3.1), i.e., solutions of Eq.(3.5), hence

$$u(1, u_1, \alpha^*, \beta) = u(1, u_1, \alpha^{**}, \beta) = u_1$$

 and

$$u_x(1, u_1, \alpha^*, \beta) = u_x(1, u_1, \alpha^{**}, \beta) = -\beta u_1$$

but

$$u_{xx}(1, u_1, \alpha^*, \beta) = -\alpha^* f(u(1, u_1, \alpha^*, \beta)) = -\alpha^* f(u_1) < 0$$

and

$$u_{xx}(1, u_1, \alpha^{**}, \beta) = -\alpha^{**}f(u(1, u_1, \alpha^{**}, \beta)) = -\alpha^{**}f(u_1) < -\alpha^{*}f(u_1) < 0$$

therefore

$$u_{xx}(1, u_1, lpha^{**}, eta) < u_{xx}(1, u_1, lpha^*, eta) < 0$$

which means that in a neighborhood of x = 1 the curves are concave down and

$$u(x, u_1, \alpha^{**}, \beta) < u(x, u_1, \alpha^{*}, \beta)$$
 (3.13)

Now either the inequality (3.13) is true for all $x \in [0, 1]$, or there is a point $x_0 \in [0, 1]$ such that

$$u(x_0, u_1, \alpha^{**}, \beta) = u(x_0, u_1, \alpha^*, \beta)$$

 \mathbf{and}

$$u(x, u_1, lpha^{stst}, eta) < u(x, u_1, lpha^st, eta) \quad ext{for} \quad x_0 < x < 1$$

hence by the nonincreasing property of f

$$f(u(x,u_1,lpha^{**},eta))\geq f(u(x,u_1,lpha^*,eta))$$

and

$$\int_x^1 f(u(t, u_1, \alpha^{**}, \beta)) dt \ge \int_x^1 f(u(t, u_1, \alpha^{*}, \beta)) dt$$

thus

$$u_{x}(x_{0}, u_{1}, \alpha^{**}, \beta) - u_{x}(x_{0}, u_{1}, \alpha^{*}, \beta)$$

$$= \alpha^{**} \int_{x_{0}}^{1} f(u(t, u_{1}, \alpha^{**}, \beta)) dt - \alpha^{*} \int_{x_{0}}^{1} f(u(t, u_{1}, \alpha^{*}, \beta)) dt$$

$$> 0$$
(3.14)

since $\alpha^{**} > \alpha^*$.

From (3.13) for $x_0 < x < 1$

$$u(x, u_1, \alpha^{**}, \beta) - u(x_0, u_1, \alpha^{**}, \beta) < u(x, u_1, \alpha^{*}, \beta) - u(x_0, u_1, \alpha^{*}, \beta)$$

assume x is greater than x_0 and divide the above inequality by $x - x_0$

$$\frac{u(x, u_1, \alpha^{**}, \beta) - u(x_0, u_1, \alpha^{**}, \beta)}{x - x_0} < \frac{u(x, u_1, \alpha^{*}, \beta) - u(x_0, u_1, \alpha^{*}, \beta)}{x - x_0}$$

let $x \longrightarrow x_0$ from right side thus

$$u_x(x_0^+, u_1, \alpha^{**}, \beta) \le u_x(x_0^+, u_1, \alpha^{*}, \beta)$$

which contradicts (3.14). So there is no such x_0 and (3.13) and hence (3.14) is true for all $x \in [0,1]$. In particular $u_x(0, u_1, \alpha^{**}, \beta) > u_x(0, u_1, \alpha^*, \beta)$, which means $u_x(0, u_1, \alpha, \beta)$ is a strictly increasing function of α . Uniqueness then follows. #

In order to prove that $u(x, u_1, \alpha, \beta)$ is continously dependent upon u_1 , Gronwall's inequality is used. Gronwall's inequality is first stated.

Lemma 3.1.3 [30] If u(t) and v(t) are continuous nonnegative functions on interval $0 \le t \le L$ and M is a nonnegative constant, then

$$u(t) \le M + \int_0^t v(s)u(s)ds, \qquad 0 \le t \le L$$

implies

$$u(t) \leq Mexp(\int_0^t v(s)ds), \qquad 0 \leq t \leq L$$

Property 3.1.7 The solution $u(x, u_1, \alpha, \beta)$ of Eq.(3.5) for fixed α and β is continuously dependent upon u_1 . That is $\forall \varepsilon > 0$, $\exists \delta > 0$ such that if

$$|u_1^* - u_1^{**}| < \delta$$

then

$$|u(x,u_1^*,\alpha,\beta)-u(x,u_1^{**},\alpha,\beta)|<\varepsilon$$

Proof: Since u is a solution of Eq.(3.5), thus

$$u(x, u_1^*, \alpha, \beta) = T_* u(x, u_1^*, \alpha, \beta)$$

= $u_1^* [1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t, u_1^*, \alpha, \beta))(t - x) dt$

#

 $\quad \text{and} \quad$

$$u(x, u_1^{**}, \alpha, \beta) = T_{**}u(x, u_1^{**}, \alpha, \beta)$$

= $u_1^{**}[1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t, u_1^{**}, \alpha, \beta))(t - x)dt$

Thus

$$\begin{aligned} &|u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)| \\ &= |(u_1^* - u_1^{**})[1 + \beta(1 - x)] - \alpha \int_x^1 (f(u(t, u_1^*, \alpha, \beta)) - f(u(t, u_1^{**}, \alpha, \beta)))(t - x)dt \\ &\leq |u_1^* - u_1^{**}|(1 + \beta) + 10\alpha \int_x^1 |u(t, u_1^*, \alpha, \beta)) - u(t, u_1^{**}, \alpha, \beta)|(t - x)dt \end{aligned}$$

where the Lipschitz condition $|f'(*)| \leq 10$ for f has been used. Then by lemma 3.1.3

$$|u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)| \le |u_1^* - u_1^{**}|(1+\beta)exp(\int_x^1 10\alpha(t-x)dt)$$

SO

$$|u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)| \le |u_1^* - u_1^{**}|(1+\beta)\exp(\frac{10\alpha}{2}(1-x)^2)$$

hence

$$|u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)| \le \exp(\frac{10\alpha}{2})(1+\beta)|u_1^* - u_1^{**}|$$

Therefore, $\forall \varepsilon > 0$, take $\delta < \frac{1}{e^{5\alpha}(1+\beta)}\varepsilon$, whenever $|u_1^* - u_1^{**}| < \delta$ then

$$|u(x,u_1^*,\alpha,\beta)-u(x,u_1^{**},\alpha,\beta)|<\varepsilon$$

The continuity of $u(x, u_1, \alpha, \beta)$ with respect to u_1 follows.

Corollary 3.1.2 $u_x(x, u_1, \alpha, \beta)$ is also continuously dependent upon u_1 **Proof:** From (3.11),

$$u_x(x, u_1, \alpha, \beta) = -\beta u_1 + \alpha \int_x^1 f(u(t, u_1, \alpha, \beta)) dt$$

#

Then

$$\begin{aligned} &|u_x(x, u_1^*, \alpha, \beta) - u_x(x, u_1^{**}, \alpha, \beta)| \\ &= |\beta(u_1^{**} - u_1^*) + \alpha \int_x^1 (f(u(x, u_1^*, \alpha, \beta)) - f(u(x, u_1^{**}, \alpha, \beta))) dt| \\ &\leq \beta |u_1^* - u_1^{**}| + \alpha 10 \int_x^1 |u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)| dt \\ &\leq \beta |u_1^* - u_1^{**}| + \alpha 10 ||u(x, u_1^*, \alpha, \beta) - u(x, u_1^{**}, \alpha, \beta)|| \end{aligned}$$

where the Lipschitz condition for f is used. By property 3.1.7, the result is true. #

Theorem 3.1.4 For any given positive number α and β , there exists one and only one $u_1 \in \left[\frac{\alpha e^{-10}}{\beta}, \frac{\alpha}{\beta}\right]$ such that $u(x, u_1, \alpha, \beta)$ obtained through Eq.(3.5) is a solution of Eq.(3.1).

Proof: As previously discussed, $u_1 \in \left[\frac{\alpha e^{-10}}{\beta}, \frac{\alpha}{\beta}\right]$ is a necessary condition. From (3.11),

$$u_x(0, u_1, \alpha, \beta) = -\beta u_1 + \alpha \int_0^1 f(u(t, u_1, \alpha, \beta)) dt$$

hence by the property $f(u) \ge e^{-10}$

$$u_x(0,\frac{\alpha e^{-10}}{\beta},\alpha,\beta) = -\alpha e^{-10} + \alpha \int_0^1 f(u(t,\frac{\alpha e^{-10}}{\beta},\alpha,\beta))dt \ge 0$$

and

$$u_x(0,\frac{\alpha}{\beta},\alpha,\beta) = -\alpha + \alpha \int_0^1 f(u(t,\frac{\alpha}{\beta},\alpha,\beta))dt \le -\alpha + \alpha = 0$$

since $f \leq 1$. By corollary 3.1.2, $u_x(x, u_1, \alpha, \beta)$ is a continuous function of u_1 , so there must exist u_1^* in $\left[\frac{\alpha}{\beta}e^{-10}, \frac{\alpha}{\beta}\right]$ such that $u_x(0, u_1^*, \alpha, \beta) = 0$.

As for the uniqueness, this may be demonstrated by a method similar to that used for the proof of property 3.1.6. Suppose there exist two values u_1^* and u_1^{**} which make

$$u_x(0,u_1^*,\alpha,\beta)=u_x(0,u_1^{**},\alpha,\beta)=0$$

Without losing generality, suppose $u_1^* > u_1^{**}$, for any $u(x) \in C[0, 1]$ define sequences

$$u_*^n = u^n(x, u_1^*, \alpha, \beta), \quad u_{**}^n = u^n(x, u_1^{**}, \alpha, \beta)$$

by

$$u_*^{n+1} = T_* u_*^n = u_1^* [1 + \beta(1-x)] - \alpha \int_x^1 f(u_*^n)(t-x) dt$$

and

$$u_{**}^{n+1} = T_{**}u_{**}^n = u_1^{**}[1 + \beta(1-x)] - \alpha \int_x^1 f(u_{**}^n)(t-x)dt$$

where $n = 0, 1, \dots, \text{ and } u_*^0 = u_{**}^0 = u(x)$, then

$$u_*^1 = T_* u_*^0$$

$$= u_1^* [1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x) dt$$

$$> u_1^{**} [1 + \beta(1 - x)] - \alpha \int_x^1 f(u(t))(t - x) dt$$

$$= T_{**} u_{**}^0 = u_{**}^1$$

by property 3.1.2, T is monotone operator, it is easy to prove

$$u^{n}(x, u_{1}^{*}, \alpha, \beta) = T^{n}_{*}u > T^{n}_{**}u = u^{n}(x, u_{1}^{**}, \alpha, \beta)$$
(3.15)

In fact, by induction, suppose

 $u_*^n > u_{**}^n$

hence by the nonincreasing property of f,

$$f(u_*^n) \le f(u_{**}^n)$$

thus

$$u_*^{n+1} = T_* u_*^n$$

.

$$= u_1^*[1+\beta(1-x)] - \alpha \int_x^1 f(u_*^n)(t-x)dt$$

> $u_1^{**}[1+\beta(1-x)] - \alpha \int_x^1 f(u_{**}^n)(t-x)dt$
= $T_{**}u_{**}^n = u_{**}^{n+1}$

Therefore the inequality (3.15) is true. By lemma 3.1.2 and theorem 3.1.3, the two function sequences $\{u^n(x, u_1^*, \alpha, \beta)\}_{n=1}^{\infty}$ and $\{u^n(x, u_1^{**}, \alpha, \beta)\}_{n=1}^{\infty}$ are convergent. For convenience, denote the limits of them as $u(x, u_1^*, \alpha, \beta)$ and $u(x, u_1^{**}, \alpha, \beta)$ respectively. Hence

$$u(x, u_1^*, \alpha, \beta) \ge u(x, u_1^{**}, \alpha, \beta)$$

Equality is only possible if $u_1^* = u_1^{**}$ by Eq.(3.5). Suppose the inequality is true. By the nonincreasing property of f,

$$f(u(x, u_1^*, \alpha, \beta)) \leq f(u(x, u_1^{**}, \alpha, \beta))$$

hence

$$u_{x}(0, u_{1}^{*}, \alpha, \beta) = -\beta u_{1}^{*} + \alpha \int_{0}^{1} f(u(t, u_{1}^{*}, \alpha, \beta)) dt$$

$$< -\beta u_{1}^{**} + \alpha \int_{0}^{1} f(u(t, u_{1}^{*}, \alpha, \beta)) dt$$

$$\leq -\beta u_{1}^{**} + \alpha \int_{0}^{1} f(u(t, u_{1}^{**}, \alpha, \beta)) dt$$

$$= u_{x}(0, u_{1}^{**}, \alpha, \beta)$$

which means u_1^* and u_1^{**} can not make $u_x(0, u_1, \alpha, \beta) = 0$ simultaneously. Thus the conclusion is obtained. #

3.2 Steady State Problem for NTC

In this section, the equation and boundary condition are the same as that in §3.1 except that here f(u) is replaced by $\frac{1}{f(u)}$. For convenience, denote $g(u) = \frac{1}{f(u)}$, hence

$$g(u) = \begin{cases} 1 & , u < 1 \\ e^{10(u-1)} & , 1 \le u \le 2 \\ e^{10} & , 2 < u \end{cases}$$
(3.16)

and Eq.(3.1) can be rewritten as

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha g(u) = 0 , 0 < x < 1\\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(3.17)

Since g(u) is defined as the reciprocal of f(u), the properties of g(u) are changed. One of the important changes is that g(u) is a nondecreasing function. However there are still some similar results. For completeness, the conclusions are given in the following theorems and properties.

Theorem 3.2.1 For Eq.(3.17) and g(u) defined by (3.16), if $u(x) \le 1.0$ for $0 \le x \le 1$, then

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)$$

is the unique solution. Similarly, if $u(x) \ge 2$ for $0 \le x \le 1$, then

$$u(x) = \frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{2}(1-x^2)$$

is the unique solution.

Property 3.2.1 Suppose the solution of Eq.(3.17) exists and has second derivative, then u(x) is monotonically nonincreasing for $x \in [0,1]$, and also $u(1) \ge 0.$ #

#

Corollary 3.2.1 If
$$u(1)$$
 is zero, then $u(x) \equiv 0$. #

Theorem 3.2.2 $u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)$ is a solution of Eq.(3.17) iff $\alpha \leq \frac{2\beta}{\beta+2}$ (i.e., $u(1) \leq \frac{2}{\beta+2}$) and $u(x) = \frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{2}(1-x^2)$ is a solution of Eq.(3.17) iff $\alpha \geq 2\beta e^{-10}$ (i.e., $u(1) \geq 2$). (Note that for $\beta + 2 < e^{10}$ and $2\beta e^{-10} < \alpha < \frac{2\beta}{\beta+2}$ both solutions exist.)

As in § 3.1, an operator Q is defined as

$$Qu = u_1[1 + \beta(1 - x)] - \alpha \int_x^1 g(u(t))(t - x)dt$$
(3.18)

and the corresponding initial value problem is as follows

$$\begin{cases} u = Qu \\ u(1) = u_1 \\ u'(1) = -\beta u_1 \end{cases}$$
(3.19)

Instead of being monotonically nondecreasing, the operator Q is monotonically nonincreasing. Thus

Property 3.2.2 *Q* is a monotonically nonincreasing operator. #

Since g(u) is nondecreasing, the proof is obvious.

Property 3.2.3 If u(x) is the cold (then $u_1 = \frac{\alpha}{\beta}$) or the hot (then $u_1 = \frac{\alpha e^{10}}{\beta}$) solution of Eq. (3.17), then

$$u(x) = Qu(x)$$

#

Property 3.2.4 If u(x) is a solution of Eq.(3.17), then u(x) is a solution of Eq.(3.19) where $u_1 = u(1)$. #

Theorem 3.2.3 Let Q be defined on C[0,1] by (3.18), and g be defined by (3.16), then for any α and β , there exists one and only one continuous function

 $u(x) \in C[0,1]$ such that

$$u = Qu \quad (= u_1[1 + \beta(1 - x)] - \alpha \int_x^1 g(u(t))(t - x)dt)$$
 (3.20)

where u_1 is an arbitrary real number.

The proof is as before except that, now n must be chosen large enough that

$$\eta = \frac{\alpha^n (10e^{10})^n}{(2n)!} < 1$$

Property 3.2.5 If $u(x) \in C[0,1]$ is a solution of Eq.(3.20), then u(x) is a solution of Eq.(3.19); furthermore it has a second derivative. #

Although the following two results are true, it should be noted that we no longer have uniqueness of the solution of the boundary value problem when α is fixed.

Property 3.2.6 For any given positive number u_1 and β , there exists an α , such that $u(x, u_1, \alpha, \beta)$ satisfies Eq. (3.17). #

Theorem 3.2.4 For any given real positive numbers α and β , there exists a $u_1 \in \left[\frac{\alpha}{\beta}, \frac{\alpha e^{10}}{\beta}\right]$ such that $u(x, u_1, \alpha, \beta)$ obtained through Eq.(3.19) is a solution of Eq.(3.17).

In fact, from the numerical solutions obtained in chapter 4 it is easy to see that for given α and β the uniqueness with respect to u_1 is broken.

#

3.3 Existence for Both PTC and NTC

In §3.1 and §3.2, the existence of solutions for Eq.(3.1) and Eq.(3.17) has been proven. For Eq.(3.1) uniqueness is also obtained. Here another way is used to prove the existence. This method applies for both cases. For convenience, Eq.(3.1) and Eq.(3.17) are written in one form as

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha h(u) = 0 , 0 < x < 1 \\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(3.21)

where h(u) can be either f(u) or g(u). In order to solve Eq.(3.21), a variable substitution is made, i.e., let $\xi = \sqrt{\alpha}x$, hence $u_x = \sqrt{\alpha}u_{\xi}$ and $u_{xx} = \alpha u_{\xi\xi}$. So the Eq.(3.21) becomes

$$\begin{cases} \frac{d^2u}{d\xi^2} + h(u) = 0 \quad , 0 < \xi < \sqrt{\alpha} \\ u_{\xi}(0) = 0 \qquad , \sqrt{\alpha}u_{\xi}(\sqrt{\alpha}) + \beta u(\sqrt{\alpha}) = 0 \end{cases}$$
(3.22)

As discussed in §3.1, the solution of Eq.(3.22) is monotonically nonincreasing and u(x) can not be negative. In order to prove the existence of solution for Eq.(3.22), an initial value problem corresponding to Eq.(3.22) is considered first. That is

$$\begin{cases} \frac{d^2u}{d\xi^2} + h(u) = 0 , 0 < \xi \\ u(0) = u_0 , u_{\xi}(0) = 0 \end{cases}$$
(3.23)

where u_0 is an arbitrary positive real number. Formally, the solution of Eq.(3.23) is independent of α .

Lemma 3.3.1 [30] Let a nonnegative and unbounded function $\psi(x)$ be defined on [a,b) and $\psi(x)$ be integrable over interval [a,c] for any c < b. If there is a constant number ν ($0 < \nu < 1$) such that $\lim_{x\to b^-} (b-x)^{\nu}\psi(x)$ exists, then the integral $\int_a^b \psi(x) dx$ for unbounded function $\psi(x)$ converges. If for $\nu \ge 1$, $\lim_{x \to b^-} (b - x)^{\nu} \psi(x) = d > 0$, or $\lim_{x \to b^-} (b - x)^{\nu} \psi(x) = +\infty$, then $\int_a^b \psi(x) dx$ diverges. #

Theorem 3.3.1 There is a solution for Eq. (3.23).

Proof: Multiply first equation of Eq.(3.23) with u_{ξ} and integrate from 0 to ξ , thus

$$\int_0^\xi u_{\xi\xi} u_{\xi} d\xi + \int_0^\xi h(u) u_{\xi} d\xi = 0$$

By property 3.1.1, $u(\xi)$ is strictly monotonically decreasing, i.e., $u(\xi)$ is an invertible function, so denote $u(\xi)$ as u, the second term of the above integral can be written as

$$\int_0^{\xi} h(u) u_{\xi} d\xi = \int_{u_0}^{u} h(u) du = -\int_{u}^{u_0} h(u) du = -H(u)$$

hence

$$\frac{1}{2}u_{\xi}^{2}|_{0}^{\xi} - H(u) = 0$$

Using the boundary condition in Eq.(3.23),

$$u_{\xi}^2 = 2H(u)$$

By property 3.1.1, $u_{\xi} < 0$, hence

$$u_{\xi} = -\sqrt{2H(u)}$$

then

$$-\frac{u_{\xi}}{\sqrt{2H(u)}}=1$$

Since $\lim_{u\to u_0^-} (u_0 - u)^{\frac{1}{2}} / \sqrt{H(u)} = 1 / \sqrt{h(u_0)} > 0$, by the lemma 3.3.1 the integral $\int_u^{u_0} \frac{1}{\sqrt{2H(s)}} ds$ exists. Integrate the above equation from 0 to ξ , thus

$$\xi = \int_{u}^{u_0} \frac{ds}{\sqrt{2H(s)}} = \Phi(u)$$

Since $\xi_u < 0$, ξ is a monotonically decreasing function of u, the inverse function of $\Phi(u)$ exists. Denote the inverse function of $\Phi(u)$ as $u = \phi(\xi)$, $\phi(\xi)$ is a solution of Eq.(3.23).

Theorem 3.3.2 For any given nonnegative real number u_0 , there is a unique α such that $\phi(\xi)$ (obtained in theorem 3.3.1) is a solution of Eq.(3.22).

Proof: The problem is now to find an α that makes $\phi(\xi)$ a solution of Eq.(3.22). This means to find an α such that

$$(u_x + \beta u)|_{x=1} = (\sqrt{\alpha}u_{\xi} + \beta u)|_{\xi=\sqrt{\alpha}} = -\sqrt{2\alpha}H(u(\sqrt{\alpha})) + \beta u(\sqrt{\alpha}) = 0$$

i.e., to find α such that

$$-\sqrt{2\alpha H(\phi(\sqrt{\alpha}))} + \beta \phi(\sqrt{\alpha}) = 0$$

Since H'(u) = -h(u) < 0 and $\phi'(\xi) = -\sqrt{2H(u)} < 0$, H and ϕ are monotonically nonincreasing functions. Hence the composition $(H \circ \phi)$ is monotonically nondecreasing and then $-\sqrt{2\alpha(H \circ \phi)(\sqrt{\alpha})} + \beta\phi(\sqrt{\alpha}) = \Psi(\alpha)$ is monotonically nonincreasing. Obviously, $\Psi(0) = \beta u_0 > 0$. By the definition of the function h(u), we can assume m < h(u) < M for $0 < u < u_0$, thus from Eq.(3.23)

$$-M < u_{\xi\xi} < -m$$

Integrate the above inequality from 0 to ξ and use the boundary condition in Eq.(3.23),

$$-M\xi < u_{\xi} < -m\xi$$

integrate the above inequality from 0 to ξ once more then

$$u_0 - \frac{1}{2}M\xi^2 < u < u_0 - \frac{1}{2}m\xi^2$$

therefore

$$\Psi(\alpha) = (\sqrt{\alpha}u_{\xi} + \beta u)|_{\xi = \sqrt{\alpha}} < -m\alpha + \beta(u_0 - \frac{1}{2}m\alpha)$$

That is $\Psi(\alpha)$ goes to $-\infty$ as $\alpha \longrightarrow \infty$. Thus there is a unique α such that $\phi(\xi)$ is a solution of Eq.(3.22) for given u_0, β . #

From theorem 3.3.2, for any given u_0 there exists a unique α , thus this defines a function $\alpha = \Theta(u_0)$. As $u(1) = u(\xi = \sqrt{\alpha})$ is also uniquely determined for each u_0 , u(1) is also a function of u_0 which can be denoted as $u(1) = U(u_0)$.

3.4 Existence of Solutions for the PTC and NTC Problems with External Circuit

Now consider the solutions of the full problem with the circuit loop. The previous results allow us to draw a curve of "solutions" represented in the α , u(1) plane. This curve may be parameterized by u(0). In the all "cold" portion, by theorem 3.1.2 and theorem 3.2.2, $u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)$, when $u(0) \leq 1$ which means that $\alpha \leq \frac{2\beta}{\beta+2}$, hence $u(1) = \frac{\alpha}{\beta}$. It is preferable to draw the curve in the I, u(1) plane where $\alpha = \gamma I^2$ and $\gamma = 150$ is fixed. Then in the cold portion $u(1) = \frac{\gamma}{\beta}I^2$, which is a parabola.

As for the "hot" portion, the two cases arise. The PTC case is considered first. By theorem 3.1.2, in the hot portion, $u(x) = \frac{\alpha e^{-10}}{\beta} + \frac{\alpha e^{-10}}{2}(1-x^2)$, iff $u(1) \ge 2$ which means $\alpha \ge 2e^{10}\beta$, hence $u(1) = \frac{\alpha e^{-10}}{\beta} = \frac{\gamma e^{-10}}{\beta}I^2$ is a second parabola starting at $I = \sqrt{\frac{2\beta e^{10}}{\gamma}}$. This second parabola may be extended to ∞ in both u(1) and I. These two parabolas are connected by an I, u(1) curve.

The external circuit leads to $I = [1 + \mu \int_0^1 f(u) dx]^{-1}$. From the first equation of

Eq.(3.1), $\alpha \int_0^1 f(u) dx = -u_x |_0^1 = \beta u(1)$, therefore

$$I = [1 + \frac{\mu \beta u(1)}{\gamma I^2}]^{-1}$$

or

$$I + \frac{\mu\beta u(1)}{\gamma I} = 1$$

This gives the third parabola

$$u(1) = \frac{\gamma}{\mu\beta}I(1-I) \tag{3.24}$$

with vertex at $I = \frac{1}{2}$, $u(1) = \frac{\gamma}{4\mu\beta}$ and passing through (0,0) and (1,0). This curve obviously meets the I, u(1) curve of solution of Eq.(3.1) at least once giving one nontrivial solution to the problem with the external circuit, since the solution curve of Eq.(3.1) begins inside the parabola (3.24) and finishes outside of that parabola. We say that the intersections of those two curves are solutions for PTC problem (2.13). That is because the two curves are both parameterized by I and satisfy Eq.(3.1) and Eq.(3.24) respectively. Thus if there are points satisfying both equations, they will satisfy Eq.(2.13) too. The curves may meet twice or three times giving two or three solutions respectively. There will be three intersections, i.e., three solutions, if the point $(I, u(1)) = (\sqrt{\frac{2\beta}{\gamma(\beta+2)}}, \frac{2}{\beta+2})$ is outside the parabola (3.24). This occurs if

$$\frac{2}{\beta+2} > \frac{\gamma}{\mu\beta} \sqrt{\frac{2\beta}{\gamma(\beta+2)}} (1 - \sqrt{\frac{2\beta}{\gamma(\beta+2)}})$$

Simplify the above inequality, then

$$\beta > \frac{2\gamma}{2(1+\mu)^2-\gamma}$$

This gives a condition for the existence of three solutions. From the numerical results and figures in chapter 4, for the exponential functions if the point (I, u(1)) =

 $(\sqrt{\frac{2\beta}{\gamma(\beta+2)}}, \frac{2}{\beta+2})$ is on the parabola (3.24), that means $\beta = \frac{2\gamma}{2(1+\mu)^2-\gamma}$, there are just two solutions (intersections). In chapter 4, the Figure 4.1.1, Figure 4.1.2 and Figure 4.1.3 demonstrate this.

\$ú.

The case corresponding to Eq.(3.17) is similar and is briefly discussed here. In the hot portion, the solution is $u(x) = \frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{\beta}(1-x^2)$, iff $u(1) \ge 2$, i.e., $\alpha \ge 2\beta e^{-10}$, hence $u(1) = \frac{\alpha e^{10}}{\beta} = \frac{\gamma e^{10}}{\beta}I^2$. a parabola starting at $I = \sqrt{2\beta e^{-10}\gamma}$. Obviously, this parabola may be extended to ∞ in both I and u(1). Now the external circuit leads to $I = [\mu + \int_0^1 g(u) dx]^{-1}$. From the equation $\alpha \int_0^1 g(u) dx = -u_x|_0^1 = \beta u(1)$, thus

$$I = [\mu + \frac{\beta u(1)}{\gamma I^2}]^{-1}$$

i.e.,

$$\mu I + \frac{\beta u(1)}{\gamma I} = 1$$

which gives a parabola

$$u(1) = \frac{\gamma}{\beta} I(1 - \mu I) \tag{3.25}$$

with vertex at $I = \frac{1}{2\mu}$, $u(1) = \frac{\gamma}{4\mu\beta}$ and passing through (0,0) and $(\frac{1}{\mu},0)$. Similarly, this curve obviously meets the I, u(1) curve of solution of Eq.(3.17) at least once giving one nontrivial solution, since the curve of solution of Eq.(3.17) begins inside of the parabola (3.25) and finishes outside of that parabola. If they meet twice or three times, there will be two or three solutions respectively. A condition for three possible solutions is that point $(I, u(1)) = (\sqrt{\frac{2\beta}{\gamma(\beta+2)}}, \frac{2}{\beta+2})$ lies outside the parabola (3.25). Hence

$$\frac{2}{\beta+2} > \frac{\gamma}{\beta} \sqrt{\frac{2\beta}{\gamma(\beta+2)}} (1 - \mu \sqrt{\frac{2\beta}{\gamma(\beta+2)}})$$

can be simplified as

$$\beta > \frac{2\gamma}{2(1+\mu)^2 - \gamma}$$

This is a condition that there exist three solutions. Similarly, from the numerical results and figures in chapter 4 for the exponential functions, if $\beta = \frac{2\gamma}{2(1+\mu)^2-\gamma}$, there are two solutions. There are three figures in chapter 4 as Figure 4.3.2, Figure 4.3.3 and Figure 4.3.4 which demonstrate these conclusions.

Therefore, for both Eq.(3.1) and Eq.(3.17), when the circuit loop is considered, there is only one solution when $\beta < \frac{2\gamma}{2(1+\mu)^2-\gamma}$, there are three solutions when $\beta > \frac{2\gamma}{2(1+\mu)^2-\gamma}$ and there are possibly two solutions when $\beta = \frac{2\gamma}{2(1+\mu)^2-\gamma}$.

It is easy to notice that the particular exponential functions chosen for $\sigma(u)$ are not necessary for the proof of all conclusions in §3.1, §3.2 and §3.3. The most important things are the monotonic decreasing or increasing, bounded properties and the Lipschitz condition. Thus we have following conclusions.

Corollary 3.4.1 If f is monotonic nondecreasing or nonincreasing function, which satisfies

$$0 < m \le f(u) \le M < +\infty \qquad u(x) \in C[0,1]$$

and the Lipschitz condition

$$|f(u) - f(v)| \le M_0 |u - v| \qquad \forall u, v \in C[0, 1]$$

where m, M and M_0 are constants, then for any given nonnegative α and β there exists at least one solution for Eq.(3.1). #

In addition, if f is monotonic nonincreasing, a stronger conclusion is achieved, i.e., the uniqueness is obtained. **Corollary 3.4.2** Assume f satisfies all conditions in corollary 3.3.1, furthermore, f is monotonic nonincreasing function, then for any given nonnegative α and β there exists one and only one solution for Eq.(3.1). #

3.5 Monotone Method for the PTC and NTC Problems

In §3.1, §3.2 and §3.3, existence and uniqueness under certain conditions have been proved for steady state PTC and NTC problems, i.e., Eq.(2.8) and Eq.(2.6). Here a monotone method[4] is worth a brief review though uniqueness is not obtained. Consider the general equation with the form

$$\begin{cases} Lu + f(x, u) = 0 , x \in \Omega \\ Bu = s(x) , x \in \partial\Omega \end{cases}$$
(3.26)

where $L = \Delta$, f(x, u) is a nonlinear smooth function of x and u, s(x) is a given function and B is a boundary operator defined as

$$Bu = \frac{\partial u}{\partial n} + \beta(x)u, \qquad x \in \partial \Omega$$

Here $\frac{\partial}{\partial n}$ denotes the outward conormal derivative and $\beta(x)$ is assumed nonnegative everywhere on the boundary $\partial\Omega$, i.e., $\beta(x) \ge 0$ for $x \in \partial\Omega$.

Definition 3.5.1 A smooth function u_0 is said to be an upper solution of Eq. (3.26) if

$$Lu_0 + f(x, u_0) \le 0, \qquad Bu_0 \ge s$$

similarly, v_0 is called a lower solution of Eq. (3.26) if

$$Lv_0 + f(x, v_0) \ge 0, \qquad Bv_0 \le s.$$

The following lemma is then true [4].

Lemma 3.5.1 Let there exist two smooth functions $u_0(x) \ge v_0(x)$ such that

$$Lu_0 + f(x, u_0) \le 0, \qquad Bu_0 \ge s$$

and

$$Lv_0 + f(x, v_0) \ge 0, \qquad Bv_0 \le s.$$

Assume f is smooth function and $\frac{\partial f}{\partial u}$ is bounded on $\min v_0 \leq u \leq \max u_0$. Then there exists a regular solution w of

$$Lw + f(x, w) = 0, \qquad Bw = s$$

such that $v_0 \leq w \leq u_0$.

Therefore we have the following theorems.

Theorem 3.5.1 For Eq. (3.1) and f is defined by (3.2)

$$u_0(x) = K(\frac{1}{\beta} + \frac{1}{2}(1 - x^2))$$
 with $K > \alpha$

and

 $v_0(x)=0$

are upper and lower solutions respectively. Hence there exists a regular solution w of Eq.(3.1) and Eq.(3.2) such that $v_0 \le w \le u_0$.

Proof: Since $e^{-10} \leq f(u) \leq 1$ by definition of f,

$$\frac{d^2u_0}{dx^2} + \alpha f(u_0) = -K + \alpha f(u_0) \le -K + Kf(u_0) = K(f(u_0) - 1) \le 0$$

 and

$$u_{0x}(0) = 0,$$
 $u_{0x}(1) + \beta u(1) = 0,$

#

hence $u_0(x)$ is a upper solution. Similarly, since

$$\frac{d^2 v_0}{dx^2} + \alpha f(v_0) = 0 + \alpha f(v_0) = \alpha f(v_0) \ge 0$$

and

$$v_{0x}(0) = 0,$$
 $v_{0x}(1) + \beta v_0(1) = 0,$

thus $v_0(x)$ is a lower solution. By lemma 3.5.1, there exists a regular w of Eq.(3.1) and Eq.(3.2) such that $v_0 \le w \le u_0$. #

Theorem 3.5.2 For Eq. (3.17) and g is defined by (3.16)

$$u_0(x) = K(\frac{1}{\beta} + \frac{1}{2}(1-x^2))$$
 with $K > \alpha e^{10}$

and

 $v_0(x)=0$

are upper and lower solutions respectively. Hence there exists a regular solution w of Eq.(3.17) and Eq.(3.16) such that $v_0 \le w \le u_0$. #

The proof is same as that for theorem 3.5.1 except that now $1 \le g(u) \le e^{10}$.

3.6 Time Dependent Problem

In this section, existence and uniqueness for time dependent problems are considered. A monotone method the same as that in § 3.5 is used. For convenience, a general form for the time dependent PTC and NTC problems is restated as follows.

$$\begin{cases} \frac{\partial u}{\partial t} = Lu + \alpha h(u) &, 0 < x < 1, 0 < t < t_0 \\ u_x(t,0) = 0, u_x(t,1) + \beta u(t,1) = 0 &, u(0,x) = s(x) \end{cases}$$
(3.27)

where s(x) is in $[0, \frac{\alpha}{\beta} + \frac{\alpha}{2}(1-x^2)]$ when h(x) is defined by (3.2) and s(x) is between 0 and $\frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{2}(1-x^2)$, when h(x) is defined by (3.16). $\Gamma_t = [0,1] \times (0,t_0)$.

In order to use monotone method, we need

Definition 3.6.1 A smooth function u_0 is said to be an upper solution of Eq. (3.27) if

$$Lu_0 + \alpha h(u_0) - \frac{\partial u_0}{\partial t} \le 0, \qquad Bu_0 \ge s$$

similarly, v_0 is called a lower solution of Eq.(3.27) if

$$Lv_0 + \alpha h(v_0) - \frac{\partial v_0}{\partial t} \ge 0, \qquad Bv_0 \le s.$$

where operator B is either a boundary operator or an initial value operator and s is 0 when operator B is a boundary operator. Given upper and lower solutions $u_0(t,x)$ and $v_0(t,x)$, with $v_0 \leq u_0$ on Γ_t , we choose C so large that $\alpha h_u + C > 0$ on the region $(x,t) \in \Gamma_t$, $\min_{\Gamma_t} v_0 \leq u \leq \max_{\Gamma_t} u_0$. Then define u_1 by

$$\begin{cases} Lu_1 - Cu_1 - \frac{\partial u_1}{\partial t} = -[\alpha h(u_0) + Cu_0] &, x \in \Gamma_t \\ u_{1x}(t,0) = 0, u_{1x}(t,1) + \beta u_1(t,1) = 0 &, u_1(0,x) = s(x) \end{cases}$$
(3.28)

By the maximum principle for parabolic equations it is easily seen that $u_1(t,x) \leq u_0(t,x)$ in Γ_t . The mapping $u_0(t,x) \rightarrow u_1(t,x)$ is denoted by $u_1 = Gu_0$. G is a monotone operator[4].

Lemma 3.6.1[4] Let there exist an upper solution $u_0(t, x)$:

$$Lu_0 + \alpha h(u_0) - \frac{\partial u_0}{\partial t} \leq 0,$$
$$Bu_0 \geq s \quad on \quad \partial \Gamma_t,$$

and a lower solution $v_0(t, x)$:

$$Lv_0 + \alpha h(v_0) - \frac{\partial v_0}{\partial t} \ge 0,$$

$$Bv_0 \leq s \quad on \quad \partial \Gamma_t,$$

with $v_0 \leq u_0$. Define sequences u_n and v_n inductively by $u_{n+1} = Gu_n$, $v_{n+1} = Gv_n$. If C is chosen large enough so that

$$lpha rac{\partial h(u)}{\partial u} + C > 0$$
 on $\min_{\Gamma_t} v_0 < u < \max_{\Gamma_t} u_0$,

then the sequences $\{u_n\}$ and $\{v_n\}$ are monotone nonincreasing and nondecreasing respectively. As n tends to infinity they both tend to a unique fixed point u = Gu, which is a strong solution of

$$Lu + lpha h(u) - rac{\partial u}{\partial t} = 0, \quad Bu = s \quad on \quad \partial \Gamma_t.$$

such that $v_0(t,x) \leq u(t,x) \leq u_0(t,x)$

As in §3.5, there are

Theorem 3.6.1 For Eq. (3.27) and h is defined by (3.2)

$$u_0(t,x) = K(rac{1}{eta} + rac{1}{2}(1-x^2)) \qquad with \qquad K > lpha$$

and

$$v_0(t,x)=0$$

are upper and lower solutions respectively. Hence there exists a unique regular solution w of Eq.(3.27) and Eq.(3.2) such that $v_0(t,x) \le w(t,x) \le u_0(t,x)$.

Proof: As h(u) is defined by (3.2), $e^{-10} \le h(u) \le 1$. Also $v_0(0,x) \le s(x) \le u_0(0,x)$ is given condition.

$$\frac{\partial^2 u_0}{\partial x^2} + \alpha h(u_0) - \frac{\partial u_0}{\partial t} = -K + \alpha h(u_0) \le -K + Kh(u_0) = K(h(u_0) - 1) \le 0$$

and

$$u_{0x}(t,0) = 0, \quad u_{0x}(t,1) + \beta u_0(t,1) = 0$$

50

艹
π

hence $u_0(t,x)$ is an upper solution. It is easily seen that

$$\frac{\partial^2 v_0}{\partial x^2} + \alpha h(v_0) - \frac{\partial v_0}{\partial t} = 0 + \alpha h(v_0) \ge 0$$

and

$$v_{0x}(t,0) = 0, \quad v_{0x}(t,1) + \beta v_0(t,1) = 0$$

thus $v_0(t, x)$ is a lower solution. By lemma 3.6.1, there exists a unique regular w of Eq.(3.27) and Eq.(3.2) such that $v_0(t, x) \le w(t, x) \le u_0(t, x)$. #

For the same argument, it is easy to get

Theorem 3.6.2 For Eq. (3.27) and h is defined by (3.16)

$$u_0(t,x) = K(\frac{1}{\beta} + \frac{1}{2}(1-x^2))$$
 with $K > \alpha e^{10}$

and

$$v_0(t,x)=0$$

are upper and lower solutions respectively. Hence there exists a unique regular solution w of Eq.(3.27) and Eq.(3.16) such that $v_0(t,x) \le w(t,x) \le u_0(t,x)$. #

Now consider the existence and uniqueness of solution for Eq.(2.12). For consistency, it is rewritten as

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \frac{\gamma f(u)}{(1+\mu \int_0^1 f(u) dx)^2} & , 0 < x < 1, 0 < t < t_0 \\ u_x(t,0) = 0, u_x(t,1) + \beta u(t,1) = 0 & , u(0,x) = s(x) \end{cases}$$
(3.29)

where f is defined by (3.2).

It is not difficult to see that during the proof [4] of lemma 3.6.1 the important thing is to construct two monotone sequences, one is nondecreasing and the other is nonincreasing. For Eq.(3.29), though there is an integral as part of denominator,

the monotone sequences can be still constructed. Similarly, define an operator G: $u_{n+1} \rightarrow u_n$ as

$$\begin{cases} Lu_{n+1} - Cu_{n+1} - \frac{\partial u_{n+1}}{\partial t} = -\left[\frac{\gamma f(u_n)}{(1+\mu \int_0^1 f(u_n) dx)^2} + Cu_n\right] &, 0 < x < 1, 0 < t < t_0 \\ u_{n+1,x}(t,0) = 0, u_{n+1,x}(t,1) + \beta u_{n+1}(t,1) = 0 &, u_{n+1}(0,x) = s(x) \end{cases}$$

we say G is monotone operator. In fact, suppose $u \ge v$, then

$$(L - C - \frac{\partial}{\partial t})(Gu - Gv) = -\left[\frac{\gamma f(u)}{(1 + \mu \int_0^1 f(u) dx)^2} - \frac{\gamma f(v)}{(1 + \mu \int_0^1 f(v) dx)^2} + C(u - v)\right]$$

Now it is needed to prove that the right hand side is negative for a large enough constant C. In fact, when u > v, $f(u) \le f(v)$. Hence,

$$(1 + \mu e^{-10})^2 \le (1 + \mu \int_0^1 f(u) dx)^2 \le (1 + \mu \int_0^1 f(v) dx)^2 \le (1 + \mu)^2$$

that is

.

$$\frac{1}{(1+\mu e^{-10})^2} \geq \frac{1}{(1+\mu\int_0^1 f(u)dx)^2} \geq \frac{1}{(1+\mu\int_0^1 f(v)dx)^2} \geq \frac{1}{(1+\mu)^2}$$

therefore

$$\begin{aligned} & \frac{f(u)}{(1+\mu\int_0^1 f(u)dx)^2} - \frac{f(v)}{(1+\mu\int_0^1 f(v)dx)^2} \\ & \geq \frac{f(u)}{(1+\mu\int_0^1 f(v)dx)^2} - \frac{f(v)}{(1+\mu\int_0^1 f(v)dx)^2} \\ & \geq \frac{f(u) - f(v)}{(1+\mu\int_0^1 f(v)dx)^2} \\ & \geq \frac{f(u) - f(v)}{(1+\mu e^{-10})^2} = \frac{f'(u^*)(u-v)}{(1+\mu e^{-10})^2} \end{aligned}$$

where the fact of $f(u) - f(v) \le 0$ is used, u^* is a function between u and v. As $|f'(u)| \le 10$, take C large enough such that $C > \frac{10\gamma}{(1+\mu e^{-10})^2}$, then the right hand side

is negative. By the maximum principle for parabolic equations, $Gu \ge Gv$, or G is a monotone operator. Therefore, there is following theorem.

Theorem 3.6.3 For Eq. (3.29) and f is defined by (3.2)

$$u_0(t,x)=K(rac{1}{eta}+rac{1}{2}(1-x^2)) \qquad with \qquad K\geq lpha$$

and

$$v_0(t,x)=0$$

are upper and lower solutions respectively. Hence there exists a unique regular solution w of Eq.(3.29) and Eq.(3.2) such that $v_0(t,x) \le w(t,x) \le u_0(t,x)$. #

Since the nonincreasing property of f(u) is used in the above proof, this method can not be used directly for the NTC problem where the g(u) is nondecreasing. However, it may still be possible to prove the uniqueness if the positive derivative is generalized as a positive operator. As it needs more concepts, such as weak form, weak solution, Banach space, etc., it is not discussed here in details.

Chapter 4

The Numerical Results

In this chapter, the numerical solutions for steady state and time dependent problems are given. There is agreement between the steady state solution calculated in §4.1 and the steady state solution of the time dependent problem as time increases.

4.1 Steady State Problem for PTC

In this part, the steady state problem for PTC is numerically solved for different parameters. The uniqueness and nonuniqueness of the solutions can be numerically obtained and demonstrated by graphs.

4.1.1 Without External Circuit

For convenience, the steady state problem is rewritten as following:

$$\begin{cases} \frac{d^2 u}{dx^2} + \alpha f(u) = 0 , 0 < x < 1 \\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.1)

and

$$f(u) = \begin{cases} 1 & , u < 1 \\ e^{-10(u-1)} & , 1 < u < 2 \\ e^{-10} & , 2 < u \end{cases}$$
(4.2)

where α and β are parameters. Most of the effort is given to the case where $1 \le u \le 2$ for some x since the cases for u < 1 and u > 2 for all x are easy to treat.

Actually, as has been stated in Chapter 3

$$u = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2) \tag{4.3}$$

is a solution whenever $\alpha < \frac{1}{(\frac{1}{\beta} + \frac{1}{2})}$. Similarly

$$u = \frac{\alpha e^{-10}}{\beta} + \frac{\alpha e^{-10}}{2} (1 - x^2) \quad , u > 2$$
(4.4)

is a solution whenever $\frac{\alpha e^{-10}}{\beta} > 2$ i.e. $\alpha > 2e^{10}\beta$. For α between these values u will be within [1,2] for some x and an explicit solution is difficult.

In order to solve equation (4.1) numerically we rewrite it as a system of ordinary differential equations

$$\begin{cases} u_x = v , v(0) = 0 \\ v_x = -\alpha f(u) , v(1) + \beta u(1) = 0 \end{cases}$$
(4.5)

To solve this problem the shooting method[17] is employed. For the details of existence and uniqueness of the method see [17, 18]. Now suppose u(0) = s, then we can get u(x,s), v(x,s) as the solution for

$$\begin{cases} u_x = v , u(0) = s \\ v_x = -\alpha f(u) , v(0) = 0 \end{cases}$$

$$(4.6)$$

If $s = s^*$ is a solution of $v(1, s) + \beta u(1, s) = 0$, then $y(x) = u(x, s^*)$ and $z(x) = v(x, s^*)$ is a solution of (4.5), i.e., a solution of (4.1). In order to obtain a solution s^* of $v(1, s) + \beta u(1, s) = 0$, Newton's iterative method is used. We first make an initial guess s^0 for s, by solving (4.6) we get $u(x, s^n)$ and $v(x, s^n)$. If $v(1, s^n) + \beta u(1, s^n) = 0$,

 s^n is a solution. Otherwise, consider $s^n + \Delta s$ as new tentative solution, and we try to satisfy $v(1, s^n + \Delta s) + \beta u(1, s^n + \Delta s) = 0$. Since v(1, s) and u(1, s) are not linear functions of s, we can't get an exact solution for Δs directly. Using a Taylor expansion, we get $-\frac{v(1,s^n)+\beta u(1,s^n)}{\frac{dv(1,s^n)}{ds}+\beta \frac{du(1,s^n)}{ds}}$ as the approximation of Δs . Now put $s^{n+1} = s^n - \frac{v(1,s^n)+\beta u(1,s^n)}{\frac{dv(1,s^n)}{ds}+\beta \frac{du(1,s^n)}{ds}}$ as the next guess for s. Repeating the same method, we can get s^{n+2} , thus theoretically a sequence $\{s^m\}_{m=0}^{\infty}$. Also, we can get the solution s^* theoretically. In order to get the approximation of Δs , we assume that u(x,s)and v(x,s) are differentiable functions of s and differentiate the system (4.6) with respect to s, we introduce two new unknowns $\overline{u} = \frac{du}{ds}$, $\overline{v} = \frac{dv}{ds}$, and a new system of differential equations can be rewritten as

$$u_{x} = v , u(0) = s^{n}$$

$$v_{x} = -\alpha f(u) , v(0) = 0$$

$$\overline{u}_{x} = \overline{v} , \overline{u}(0) = 1$$

$$\overline{v}_{x} = -\alpha f'(u)\overline{u} , \overline{v}(0) = 0$$

$$(4.7)$$

and $\Delta s = -\frac{v(1,s^n) + \beta u(1,s^n)}{\overline{v}(1,s^n) + \beta \overline{u}(1,s^n)}$, $n = 0, 1, 2, \dots, s^0$ arbitrary.

Now the problem becomes an initial value problem for the system (4.7). On an IBM-RISC 6000 computer running a Unix operating system, we use the IMSL library routine IVPRK (refer to Appendix A) to solve the initial value problem (4.7). During the progress, we met an interesting phenomena. For some parameters α and β if the initial value s^0 is not close to the solution, then an interesting cycle appears. After 2 or 3 steps, a two cycle appears, so that s^{n+2} is the same as s^n and s^{n+1} is the same as s^{n-1} with s^n different from s^{n-1} . If we make an initial guess very close to the solution, it seems to converge very quickly for any parameters α and β . To overcome the problem of cycling a bisection method is introduced. The bisection method guarantees convergence although the rate of convergence is slower than that obtained by Newton iterations. The technique used is that whenever oscillation appears or the increment of Δs is greater than 0.5 then the bisection subroutine is called and the condition for returning to Newton's iterative scheme is that Δs is less than 0.5. Here we choose the condition that Δs is less than 0.5 by numerical experimental experience. Actually, without the bisection method, oscillation will occur, whenever Δs is greater than 1.0.

The bisection routine is used for choosing the initial value for Newton iteration method. The technique is as follows. From any initial value for s and parameters α , β , the scheme is as following:

(1) Using IMSL library routine IVPRK solve initial value problem (4.7) to get values of $u(1, s^n)$, $v(1, s^n)$, $\overline{u}(1, s^n)$ and $\overline{v}(1, s^n)$. Set $\Delta s = -\frac{v(1, s^n) + \beta u(1, s^n)}{\overline{v}(1, s^n) + \beta \overline{u}(1, s^n)}$, and $T_n = v(1, s^n) + \beta u(1, s^n)$. Goto step (2).

(2) Set $s^{n+1} = s^n + \Delta s$ and solve (4.7) to get values of $u(1, s^{n+1})$, $v(1, s^{n+1})$, $\overline{u}(1, s^{n+1})$ and $\overline{v}(1, s^{n+1})$. Calculate $\Delta s = -\frac{v(1, s^{n+1}) + \beta u(1, s^{n+1})}{\overline{v}(1, s^{n+1}) + \beta \overline{u}(1, s^{n+1})}$, and $T_{n+1} = v(1, s^{n+1}) + \beta u(1, s^{n+1})$, goto step (3).

(3) If $\Delta s \leq 0.5$, return to Newton iterations, else goto step (4).

(4) If $T_{n+1} \times T_n \ge 0$, go to step (1), else go to step (5).

(5) Using $s^{n+2} = \frac{s^{n+1}+s^n}{2}$ as initial guess for s to get values of $u(1, s^{n+2}), v(1, s^{n+2}), \overline{u}(1, s^{n+2})$ and $\overline{v}(1, s^{n+2})$, calculating $\Delta s = -\frac{v(1, s^{n+2}) + \beta u(1, s^{n+2})}{\overline{v}(1, s^{n+2}) + \beta \overline{u}(1, s^{n+2})}$, and $T = v(1, s^{n+2}) + \beta u(1, s^{n+2})$. If $\Delta s \leq 0.5$, return to Newton iteration, else goto step (6).

(6) If $T \times T_n \ge 0$, then $T_n = T$, $s^n = s^{n+2}$, go to step (5), else $T_{n+1} = T$, $s^{n+1} = s^{n+2}$, go to step (5).

The main idea for bisection is as follows: Assume s^* to be the root of v(1,s) + v(1,s) +

 $\beta u(1,s) = 0$, first find two points s^0 and s^1 such that $v(1,s) + \beta u(1,s)$ takes different signs at the two points. Then from these two points, we bisect the interval $[s^0, s^1]$ and get a new point s^2 , evaluate the sign of $v(1,s) + \beta u(1,s)$ at point s^2 to choose $[s^0, s^2]$ or $[s^2, s^1]$ as the new interval. Every time there are two points which make $v(1,s) + \beta u(1,s)$ take different signs. Repeating this procedure, a sequence $\{s^k\}$, $k = 1, 2, \ldots$ is obtained. Theoretically, $s^k \to s^*$. As it is known, the convergent rate of this method is linear, so it is used to choose an initial value for Newton's iterative method. In the above bisection scheme, step (1) to step (4) are used to find two points such that $v(1,s) + \beta u(1,s)$ assumes two different signs at these two points; step (5) to step (6) are used to further bisect the interval until the increment $\Delta s \leq 0.5$ and then a return to Newton's iterative method is made.

From the numerical results we can see that for some parameters α and β there are several shifts back and forth between Newton's method and the bisection method, and the method is convergent for any parameters and initial values. During the numerical experimental procedure, an interesting phenomenon is found. If the solutions are between 1 and 2, then for any parameters and initial values, the total number of Newton iterations is almost the same except that the bisection subroutine is used a different number of times. Also, whenever the solutions are bigger than 2 or less than 1, the number of Newton's iterations is always 2 although the number of times the bisection subroutine is used is quite different. Some data are summarized in the Table 4.1.1, where column 6 represents the number of Newton iterations, column 7 represents the number of times that bisection was used, column 8 represents the number of times the bisection routine was called, s represents the initial guess and * means total number of times bisection was used.

Table 4.1.1

Solutions of PTC Problem for Various Parameter $\alpha,\,\beta$

α	β	s	<i>u</i> (0)	u(1)	6	7	8
999.000	0.005	0.5	9.0936	9.0709	2	19	1
999.000	0.010	0.5	4.5581	4.5355	2	18	1
999.000	0.100	0.5	1.8891	1.8106	11	14	1
750.000	0.100	0.5	1.8615	1.7840	10	15	1
500.000	0.100	0.5	1.8226	1.7465	9	13	1
100.000	0.100	0.5	1.6684	1.5981	13	11	1
75.000	0.100	0.5	1.6410	1.5716	12	12	1
50.000	0.100	0.5	1.6023	1.5344	9	10	1
25.000	0.100	0.5	1.5363	1.4709	8	9	1
15.000	0.100	0.5	1.4878	1.4242	8	10	1
14.000	0.100	0.5	1.4812	1.4179	13	12*	3
10.000	0.100	0.5	1.4493	1.3872	7	8	1
7.500	0.100	0.5	1.4221	1.3611	7	9	1
5.000	0.100	0.5	1.3837	1.3242	8	7	1
1.000	0.100	0.5	1.2323	1.1787	7	5	1
0.850	0.100	0.5	1.2171	1.1641	10	5	1
0.500	0.100	0.5	1.1675	1.1165	6	4	1
0.250	0.100	0.5	1.1030	1.0546	9	3	1

α	β	S	u(0)	u(1)	6	7	8
0.100	0.100	0.5	1.0113	0.9642	6	2*	2
0.050	0.100	0.5	0.5250	0.5000	2	1	1
0.005	0.100	0.5	0.0525	0.0500	2	1	1

Table 4.1.1(continued)

Col. 6-No. of Newton iterations, Col. 7-No. of times bisection used, Col. 8-No. of calls to bisection routine.

From Table 4.1.1 it is easy to see that the numerical solutions are exactly the solutions (4.3), (4.4) whenever u < 1 or u > 2. For cases where u is between 1 and 2, it is possible to make some checks on the numerical solutions. Multiply the first equation by u_x in (4.1) and integrate from 0 to 1 with respect to x, then we get

$$\int_0^1 u_{xx} u_x dx + \alpha \int_0^1 f(u) u_x dx = 0$$
(4.8)

Denote u_0 and u_1 as the values of u at 0 and 1 respectively thus

$$\frac{1}{2}u_x^2|_0^1 + \alpha \int_{u_0}^{u_1} f(u)du = 0$$
(4.9)

and also use the boundary conditions in (4.1). Hence we have

$$\frac{1}{2}\beta^2 u^2(1) - \frac{\alpha}{10} \left[e^{-10(u_1 - 1)} - e^{-10(u_0 - 1)} \right] = 0 \tag{4.10}$$

If we substitute α , β , and u_0 , u_1 , obtained by the numerical method, into the left hand side of equation(4.10), it is seen that the error is very small (about 10^{-6}). That suggests that u_0 and u_1 are approximations of the exact solution.

From Table 4.1.1 it is easy to see that for some parameters we can get solution in which critical values of u (i.e., 1 or 2) are obtained, but there do not seem to be
values of parameters make u greater than 2 at some points and less than 1 at some other points within the interval [0,1]. Using the same IMSL library routine we can estimate the locations at which u reaches the critical values 1 or 2. The Table 4.1.2 describes this for some cases, where ξ means the value at which $u(\xi) = 1$ or $u(\xi) = 2$. Here only the estimated intervals are given. These results give some indication of the situation for the time dependent problem.

Table 4.1.2

α	β	u(0)	u(1)	ξ
440.000	0.010	2.0090	1.9990	[0.92,0.96]
0.100	0.100	1.0109	0.9638	[0.48, 0.52]
0.096	0.100	1.0029	0.9554	[0.24, 0.28]

Location of interface ξ for some α , β .

4.1.2 With External Circuit

Here the steady state one dimensional problem with the extra conditon is considered (An external circuit connected to the thermistor). Numerical solutions are discussed.

The equations are now as follows

$$\begin{aligned} \frac{d2u}{dx^2} + \alpha f(u) &= 0 \quad , 0 < x < 1 \\ u_x(0) &= 0 \qquad , u_x(1) + \beta u(1) = 0 \\ 1 &= I + \frac{\mu \beta u(1)}{\gamma I} \quad , \alpha = \gamma I^2 \end{aligned}$$
(4.11)

where γ and β are fixed parameters and f(u) is defined by (4.2). In order to solve Eq.(4.11), it is decomposed into two parts, one as Eq.(4.1) and the other as



$$u(1) = \frac{\gamma I}{\mu \beta} (1 - I) \tag{4.12}$$

Plot of u(1) and I for solution of (4.1), $0 \le I \le 10$.

For convenience, we rewrite the cold and hot solutions of (4.1) here. In fact, as we know, the cold solution ($u \le 1$) of (4.1) can be written as

$$u(x) = \frac{\gamma I^2}{\beta} + \frac{\gamma I^2}{2}(1 - x^2)$$

CHAPTER 4. NUMERICAL RESULTS

and hot solution ($u\geq 2$) of (4.1) can be written as

$$u(x) = \frac{\gamma I^2 e^{-10}}{\beta} + \frac{\gamma I^2 e^{-10}}{2} (1 - x^2)$$



Plot of u(1) and I for solution of (4.1), $0 \le I \le 1$.

By the property discussed in Chapter 3, u(x) is monotonically nonincreasing function of x, so for the cold solution

$$u(1) = \frac{\gamma I^2}{\beta} \tag{4.13}$$

iff $u(0) = \gamma I^2(\frac{1}{\beta} + \frac{1}{2}) \le 1$ and for the hot solution

$$u(1) = \frac{\gamma I^2 e^{-10}}{\beta}$$
(4.14)

iff $u(1) = \frac{\gamma I^2 e^{-10}}{\beta} \ge 2$.



Plot of u(1) and I for solution of (4.1), $1 \le I \le 10$.

In fact, Eq.(4.1) is solved already. Now the problem is how to include (4.12). If γ and β are fixed, the solution of (4.1) is a function of I, so it can be denoted as U(x, I). Let x = 1, if U(1, I) is a solution of (4.11), U(1, I) should coincide with (4.12), i.e., U(1, I) = u(1). Problem (4.11) may be solved by iteration. The method

is as follows. For a given initial I_0 , evaluating $\alpha_0 = \gamma I_0^2$, use the shooting method mentioned in §4.1.1 to solve (4.1) for the fixed α_0 , then get $u(1, I_0)$. Now use (4.12) to determine *I*. Since Eq.(4.12) is quadratic in *I* two *I* are obtained, we choose the



Graphs of u(1) and I for solution of (4.1) and solution of (4.12). Intersection point represents solution of coupled problem.

greater one for $\beta = 0.25$ as I_1 . Next a $u(1, I_1)$ is obtained as a solution of Eq.(4.1) with $\alpha = \gamma I_1^2$. Repeating this procedure, two sequences $\{I_n\}$ and $\{u(1, I_n)\}$ are obtained. If $|I_{n+1} - I_n|$ and $|u(1, I_{n+1}) - u(1, I_n)|$ are less than a given tolerance, stop otherwise continue until the sequences converge. Therefore the numerical solutions of (4.11) are obtained. The sequences converges because as it is seen in Figure 4.1.1, at the intersection (i.e., solution of Eq.(4.11)) of the two curves, the absolute value of the product of the derivatives one with respect to I and one with respect to u(1)of the two curves is less than 1. Hence the combination of the two curves makes a contraction operator about I, therefore the sequence $u(1, I_n)$ is convergent.



Graphs of u(1) and I for solution of (4.1) and solution of (4.12). Intersection points are solutions of coupled problem.

The solution to Eq.(4.11) may not be unique. This can be demonstrated graphically. If the solution of (4.1) at x = 1 is drawn as in Figure 4.1.0 (for fixed $\beta = 0.25$ for consistency, $\alpha = \gamma I^2$, $\gamma = 150$, it may be noted that the figure consists of three parts, the two critical points are $I_1 = 0.038490017$ and $I_2 = 8.568637736$. When $I \leq I_1$ and $I \geq I_2$, u(1) is defined by (4.13) and (4.14) respectively; when $I_1 \leq I \leq I_2$, u(1) is obtained by the method discussed in §4.1.1. For ease of reading the Figure 4.1.0, it is decomposed into two figures as Figure 4.1.0a and Figure 4.1.0b),



Graphs of u(1) and I for solution of (4.1) and solution of (4.12). Intersection points are solutions of coupled problem.

it is easy to see that when condition (4.12) is put into that graph, three cases may occur. One is that (4.12) and (4.1) have one intersection point. The second is that (4.12) and (4.1) have two intersection points and the third is that (4.12) and (4.1) have three intersection points. The three cases can be obtained by adjusting the parameter β when $\gamma = 150$ and $\mu = 20$ are fixed. The method is as follows.

Consider that (4.13) touches with (4.12), hence,

$$\frac{\gamma I}{\beta \mu} (1 - I) = \frac{\gamma I^2}{\beta} \tag{4.15}$$

therefore, $I_0 = \frac{1}{\mu+1}$. That means that (4.13) meets with (4.12) whenever $I = I_0$. However it must be noted that the intersection of (4.12) with (4.13) has meaning only if $u(0) = \gamma I_0^2(\frac{1}{\beta} + \frac{1}{2}) \leq 1$. From $u(0) = \gamma I_0^2(\frac{1}{\beta} + \frac{1}{2}) \leq 1$, there is $\beta \geq \frac{1}{(\mu+1)^2-\frac{1}{2}}$. We say that $\beta_0 = \frac{1}{(\mu+1)^2-\frac{1}{2}} = 0.409836065$ is a critical value. When $\beta = \beta_0$, (4.13) just touches (4.12) which means that there are two solutions; $\beta > \beta_0$, (4.13) has one intersection point with (4.12) and the end of (4.13) goes above the parabola (4.12) which means that there are three solutions; $\beta < \beta_0$, (4.13) can not meet with (4.12) which means that there is only one solution. The Figure 4.1.1 ($\beta = 0.25$), Figure 4.1.2 ($\beta = 0.409836065$) and Figure 4.1.3 ($\beta = 0.60$) demonstrate this. A similar situation can be discussed when (4.14) intersects, does not intersect and just intersects with (4.12) (here just intersects means that the end point of parabola (4.14) is just on the parabola (4.12)).

4.2 Time Dependent Problem for PTC

Here the time dependent one dimensional PTC problem is considered. Numerical solutions are discussed.

The equation is as follows

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \alpha f(u) &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.16)

and

$$f(u) = \begin{cases} 1 & , u \leq 1 \\ e^{-10(u-1)} & , 1 < u < 2 \\ e^{-10} & , 2 \leq u \end{cases}$$
(4.17)

where α and β are parameters.

4.2.1 Moving Meshpoints Method

In order to get the numerical solution of Eq.(4.16), the finite element method is used. At first we expect that the whole solution is within the "cold" region. So the Crank-Nicholson method is used to find the time when the first critical temperature is reached at one point. Whenever the critical temperature is reached, a finite element method is used. Initially, suppose the whole region to be cold, $u_{i,j} \approx u(ih, j\Delta t), i =$ $0, \ldots, N$, and h = 1.0/N, then the Crank-Nicholson difference scheme is as following

$$\frac{u_{i,j+1} - u_{i,j}}{\Delta t} = \frac{1}{2} \left[\frac{u_{i+1,j+1} - 2u_{i,j+1} + u_{i-1,j+1}}{h^2} + \frac{u_{i+1,j} - 2u_{i,j} + u_{i-1,j}}{h^2} \right] + \alpha$$

i.e.,

$$-\frac{r}{2}u_{i+1,j+1} + (1+r)u_{i,j+1} - \frac{r}{2}u_{i-1,j+1}$$
$$= \frac{r}{2}u_{i+1,j} + (1-r)u_{i,j} + \frac{r}{2}u_{i-1,j} + \alpha\Delta t$$
(4.18)

where $r = \frac{\Delta t}{h^2}, i = 0, ..., N$. For the boundary conditions, the central difference is used for good accuracy. In order to use the central difference formula it is necessary

to introduce the "fictitious" temperature $u_{-1,j}$ at the external mesh point $(-h, j\Delta t)$, by imagining the thermistor to be extended a distance h at this end. That is

$$\begin{cases}
\frac{u_{i+1,j}-u_{i-1,j}}{2h} = 0 , & u_{1,j} = u_{-1,j} , & i = 0 \\
\frac{u_{N+1,j}-u_{N-1,j}}{2h} + \beta u_{N,j} = 0 , & u_{N+1,j} = u_{N-1,j} - 2h\beta u_{N,j} , & i = N
\end{cases}$$
(4.19)

The temperature $u_{-1,j}$ is unknown and necessitates another equation. This is obtained by assuming that Eq.(4.16) is satisfied at the end point x = 0. The $u_{-1,j}$ can be eliminated between these equations. A similar method is used for the other end point x = 1.

When the solution at x = 0 reaches 1, the interface point appears. The problem at the next timestep is to choose the new position of the interface point and time stepsize. In order to solve this problem, the following method is employed.

Suppose $x = x_1(t)$ to be the interface at which the temperature is the critical temperature $u_c = 1$ (i.e., $\sigma(u_c) = 1$). Define new variables $\xi = \frac{x}{x_1(t)}, \tau = t$ (ξ is relative coordinate). Then

$$\frac{\partial}{\partial t} = \frac{\partial}{\partial \tau} - \frac{x}{x_1^2} \dot{x}_1 \frac{\partial}{\partial \xi} = \frac{\partial}{\partial \tau} - \xi \frac{\dot{x}_1}{x_1} \frac{\partial}{\partial \xi}$$
$$\frac{\partial}{\partial x} = \frac{1}{x_1} \frac{\partial}{\partial \xi}$$
(4.20)

Let $u = u(\xi, \tau)$, note $\frac{\partial u(1,\tau)}{\partial \tau} = 0$ for all τ because $u(1,\tau) = 1$ for all τ . The Eq.(4.16) becomes

$$\frac{\partial u}{\partial \tau} - \xi \frac{\dot{x}_1}{x_1} \frac{\partial u}{\partial \xi} = \frac{1}{x_1^2} \frac{\partial^2 u}{\partial \xi^2} + \alpha \sigma(u(\xi, \tau))$$
(4.21)

At $\xi = 1$, $0 - \frac{\dot{x}_1}{x_1} \frac{\partial u}{\partial \xi}|_{\xi=1} = \frac{1}{x_1^2} \frac{\partial^2 u}{\partial \xi^2}|_{\xi=1} + \alpha$, since $\frac{\partial u}{\partial \xi}|_{\xi=0} = 0$, $\frac{\partial u}{\partial \xi}|_{\xi=1} = \int_0^1 \frac{\partial^2 u}{\partial \xi^2} d\xi = x_1 \int_0^{x_1} \frac{\partial^2 u}{\partial x^2} dx = x_1^2 \cdot \frac{1}{x_1} \int_0^{x_1} \frac{\partial^2 u}{\partial x^2} dx$ and $-x_1 \dot{x}_1 [\frac{1}{x_1} \int_0^{x_1} \frac{\partial^2 u}{\partial x^2} dx] = \frac{\partial^2 u}{\partial x^2}|_{x=x_1} + \alpha$, it is now assumed that $\frac{\partial^2 u}{\partial x^2}$ changes very little for x_1 small (or let $x_1 \to 0, \dot{x}_1 \to \infty$), then

$$(1 + x_1 \dot{x}_1)(-\frac{\partial^2 u}{\partial x^2}|_{x=0}) = \alpha$$
(4.22)

because $x_1 \dot{x}_1 \left(\frac{\partial^2 u}{\partial x^2}\right)|_{x=0} = -\left(\frac{\partial^2 u}{\partial x^2}\right)|_{x=0} - \alpha$, *i.e.*, $x_1 \dot{x}_1 = -1 - \frac{\alpha}{\left(\frac{\partial^2 u}{\partial x^2}\right)_{x=0}} = \left(\frac{\alpha}{k_0} - 1\right)$. $k_0 = \frac{\partial^2 u}{\partial x^2}|_{x=0}$ is estimated from the previous time step. Thus it is assumed that

$$\frac{d}{dt}(x_1^2) = 2(\frac{\alpha}{k_0} - 1) \tag{4.23}$$

(Note that $x_1 \dot{x}_1 = \frac{1}{2} \frac{d}{dt}(x_1^2)$).

Hence at first step if x_1 is determined by the following method. The time step Δt can be chosen as $x_1^2/2(\frac{\alpha}{k_0}-1)$. In the actual program, x_1 is first assigned a value of h/8 where h is the mesh length and Δt chosen as above. Thereafter the problem is solved in two phases.

Denote the interface point as $x = \xi_2$. The interval [0,1] is divided into two segments, $[0, \xi_2]$ and $[\xi_2, 1]$. Since ξ_2 is very small at the beginning, it is not necessary to put any points within $[0, \xi_2]$. $[\xi_2, 1]$ is divided into nc equal segments. Thus the total number of node points is nt = nc + 2. Denote $hc = (1 - \xi_2)/nc$, node points by x_i . Hence $x_1 = 0$, $x_2 = \xi_2$, $x_i = \frac{nt-i}{nc}\xi_2 + \frac{i-2}{nc}$, $i = 3, \ldots, nt$. Since ξ_2 is not fixed, x_i depends upon time t. Define trial and test function as

$$\phi_i(x,t) = \begin{cases} \frac{x - x_{i-1}}{x_i - x_{i-1}} , [x_{i-1}, x_i] \\ \frac{x_{i+1} - x_i}{x_{i+1} - x_i} , [x_i, x_{i+1}] \end{cases}$$
(4.24)

Multiply Eq.(4.16) by $\phi_j(x,t)$ and integrate from 0 to 1, then

$$\int_0^1 \left(\frac{\partial u}{\partial t} - \frac{\partial^2 u}{\partial x^2} - \alpha \sigma(u)\right) \phi_j(x, t) dx = 0, \qquad (4.25)$$

where j = 1, ..., nt, i.e.,

$$< u_t, \phi_j > + < u_x, \phi_{j,x} > -\beta u(1,t) \delta_{nt,j} - \alpha < \sigma(u), \phi_j > = 0,$$
 (4.26)

where j = 1, ..., nt. Replace u by the approximate form $\sum_{j=1}^{nt} u_j \phi_j$ where u_j are functions of t and ϕ_j are functions of x and t, and integrate by parts, then

$$\sum_{i=1}^{nt} (u_{i,t} < \phi_i, \phi_j > +u_i < \phi_{i,t}, \phi_j > +u_i < \phi_{i,x}, \phi_{j,x} >) - \beta u(1, nt) \delta_{nt,j} - \alpha < \sigma(\sum_{i=1}^{nt} u_i \phi_i), \phi_j >= 0,$$
(4.27)

where j = 1, ..., nt and $\langle \phi_i, \phi_j \rangle = \int_0^1 \phi_i \phi_j dx$. Hence

$$i = 1, (4.28)$$

$$\frac{1}{3}\xi_2 u_{1,t} + \frac{u_1}{\xi_2} - \frac{u_2}{\xi_2} + \frac{1}{6}\dot{\xi_2}(u_1 - u_2) =$$

$$\frac{\alpha\xi_2}{10(u_2 - u_1)} \left[e^{-10(u_1 - 1)} + \frac{e^{-10(u_2 - 1)} - e^{-10(u_1 - 1)}}{10(u_2 - u_1)}\right]$$

$$i = 2, \qquad (4.29)$$

$$\frac{1}{6}\xi_{2}u_{1,t} + \frac{1}{6}(x_{3} - \xi_{2})u_{3,t} + \frac{1}{3}\dot{\xi}_{2}u_{1} + \frac{1}{6}\dot{x}_{3}u_{2} - (\frac{1}{3}\dot{\xi}_{2} + \frac{1}{6}\dot{x}_{3})u_{3}$$

$$-\frac{u_{1}}{\xi_{2}} + u_{2}(\frac{1}{\xi_{2}} + \frac{1}{x_{3} - \xi_{2}}) - \frac{u_{3}}{x_{3} - \xi_{2}} = \frac{\alpha hc}{2}$$

$$-\frac{\alpha\xi_{2}}{10(u_{2} - u_{1})}[e^{-10(u_{2} - 1)} + \frac{e^{-10(u_{2} - 1)} - e^{-10(u_{1} - 1)}}{10(u_{2} - u_{1})}]$$

$$\begin{split} i &= 3, \\ \frac{2}{3}hcu_{3,t} + \frac{1}{6}hcu_{4,t} + \frac{1}{hc}(-u_2 + 2u_3 - u_4) + \frac{3nc - 2}{6nc}u_2\dot{\xi_2} \\ &- \frac{1}{3nc}u_3\dot{\xi_2} - \frac{3nc - 4}{6nc}\dot{\xi_2}u_4 = \alpha hc \end{split}$$

$$i = 4, \dots, nt - 1,$$

$$\frac{hc}{6}(u_{i-1,t} + 4u_{i,t} + u_{i+1,t}) + \dot{\xi}_2(\frac{3nt - 3i + 1}{6nc}u_{i-1} - \frac{2}{6nc}u_i - \frac{3nt - 3i - 1}{6nc}u_{i+1}) + \frac{1}{hc}(-u_{i-1} + 2u_i - u_{i+1}) = \alpha hc$$

$$(4.31)$$

(4.30)

$$i = nt, \qquad (4.32)$$

$$\frac{hc}{6}(u_{nt-1,t} + 2u_{nt,t}) + \frac{1}{6nc}\dot{\xi}_2(u_{nt-1} - u_{nt}) + \frac{1}{hc}(-u_{nt-1} + u_{nt}) + \beta u_{nt} = \frac{\alpha}{2}hc$$

Now denote $\Delta \xi_2$ and Δt as increments of ξ_2 and t respectively. Use $\int_t^{t+\Delta t} f(t)dt \approx \frac{1}{2}(f(t+\Delta t)+f(t))\Delta t$ and $\int_t^{t+\Delta t} f_t g dt \approx (f(t+\Delta t)-f(t))(\frac{g(t+\Delta t)+g(t)}{2})$

Integrate the equations i = 1, ..., nt, using the above approximations. Simplifying the equations we get

$$i = 1, \qquad (4.33)$$

$$\left(\frac{1}{3}\xi_{2} + \frac{1}{4}\Delta\xi_{2} + \frac{\Delta t}{2(\xi_{2} + \Delta\xi_{2})}\right)u_{1}^{n+1} - \left(\frac{1}{6} + \frac{u_{1}^{n}}{12}\right)\Delta\xi_{2}$$

$$-\frac{\Delta t}{2(\xi_{2} + \Delta\xi_{2})} + \left(-\frac{1}{3}\xi_{2} + \frac{\Delta t}{2\xi_{2}}\right)u_{1}^{n} - \frac{\Delta t}{2\xi_{2}}$$

$$-\frac{\alpha}{2}\Delta t\left[\frac{\xi_{2} + \Delta\xi_{2}}{10(1 - u_{1}^{n+1})}\left(e^{-10(u_{1}^{n} - 1)} + \frac{1 - e^{-10(u_{1}^{n+1} - 1)}}{10(1 - u_{1}^{n+1})}\right)\right]$$

$$+\frac{\xi_{2}}{10(1 - u_{1}^{n})}\left(e^{-10(u_{1}^{n} - 1)} + \frac{1 - e^{-10(u_{1}^{n} - 1)}}{10(1 - u_{1}^{n})}\right)\right] = 0$$

$$i = 2, \qquad (4.34)$$

$$\left(\frac{1}{6}\xi_{2} + \frac{1}{4}\Delta\xi_{2} - \frac{\Delta t}{2(\xi_{2} + \Delta\xi_{2})}\right)u_{1}^{n+1} + \left(\frac{u_{1}^{n}}{12} - \frac{3nc - 2}{12nc}u_{3}^{n}\right)$$

$$+ \frac{nc - 1}{6nc} + \frac{\alpha\Delta t}{4nc}\Delta\xi_{2} + \frac{\Delta t}{2}\left(\frac{1}{\xi_{2} + \Delta\xi_{2}} + \frac{1}{hc - \frac{\Delta\xi_{2}}{nc}}\right)$$

$$+ \left(\frac{hc}{6} - \frac{1}{4}\Delta\xi_{2} - \frac{\Delta t}{2(hc - \frac{\Delta\xi_{2}}{nc})}\right)u_{3}^{n+1} - \left(\frac{1}{6}\xi_{2} + \frac{\Delta t}{2\xi_{2}}\right)u_{1}^{n}$$

$$-\left(\frac{hc}{6} + \frac{\Delta t}{2hc}\right)u_3^n + \frac{\Delta t}{2\xi_2} + \frac{\Delta t}{2hc} - \frac{\alpha hc\Delta t}{2}$$
$$-\frac{\alpha\Delta t}{2}\left[\frac{\xi_2 + \Delta\xi_2}{10(1-u_1^{n+1})}\left(1 + \frac{1-e^{-10(u_1^{n+1}-1)}}{10(1-u_1^{n+1})}\right)\right]$$
$$\frac{\Delta\xi_2}{10(1-u_1^n)}\left(1 + \frac{1-e^{-10(u_1^{n-1}-1)}}{10(1-u_1^{n})}\right)\right] = 0$$

i = 3, (4.35)

$$\begin{aligned} (\frac{2}{3} - \frac{\Delta\xi_2}{2(1-\xi_2)} + \frac{\Delta t}{hc(hc - \frac{\Delta\xi_2}{nc})})u_3^{n+1} + (\frac{1}{6} - \frac{nc-1}{4(1-\xi_2)}\Delta\xi_2 \\ - \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})})u_4^{n+1} + \Delta\xi_2(\frac{u_3^n}{6(1-\xi_2)} - \frac{3nc-5}{12(1-\xi_2)}u_4^n + \frac{3nc-2}{6(1-\xi_2)} \\ + \frac{\alpha\Delta t}{2(1-\xi_2)}) - \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})} - (\frac{2}{3} - \frac{\Delta t}{hc^2})u_3^n \\ - (\frac{1}{6} + \frac{\Delta t}{2hc^2})u_4^n - \frac{\Delta t}{2hc^2} - \alpha\Delta t = 0 \end{aligned}$$

$$\begin{aligned} i &= 4, \dots, nt - 1, \qquad (4.36) \\ &\left(\frac{1}{6} + \frac{3nt - 3i}{12(1 - \xi_2)}\Delta\xi_2 - \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})}\right)u_{i-1}^{n+1} + \left(\frac{2}{3} - \frac{\Delta\xi_2}{2(1 - \xi_2)}\right) \\ &+ \frac{\Delta t}{hc(hc - \frac{\Delta\xi_2}{nc})}u_i^{n+1} + \left(\frac{1}{6} - \frac{3nt - 3i}{12(1 - \xi_2)}\Delta\xi_2 - \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})}\right)u_{i+1}^{n+1} \\ &+ \Delta\xi_2\left(\frac{3nt - 3i + 2}{12(1 - \xi_2)}u_{i-1}^n + \frac{u_i^n}{6(1 - \xi_2)} - \frac{3nt - 3i - 2}{12(1 - \xi_2)}u_{i+1}^n + \frac{\alpha\Delta t}{2(1 - \xi_2)}\right) \\ &- \left(\frac{1}{6} + \frac{\Delta t}{2hc^2}\right)u_{i-1}^n - \left(\frac{2}{3} - \frac{\Delta t}{hc^2}\right)u_i^n - \left(\frac{1}{6} + \frac{\Delta t}{2hc^2}\right)u_{i+1}^n - \alpha\Delta t = 0 \end{aligned}$$

$$i = nt, \qquad (4.37)$$

$$\left(\frac{1}{6} - \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})}\right)u_{nt-1}^{n+1} + \left(\frac{1}{3} - \frac{\Delta\xi_2}{4(1 - \xi_2)}\right)$$

$$+ \frac{\Delta t}{2hc(hc - \frac{\Delta\xi_2}{nc})} + \frac{\beta\Delta t}{2hc}\right)u_{nt}^{n+1} + \Delta\xi_2\left(\frac{u_{nt-1}^n}{6(1 - \xi_2)} + \frac{u_{nt}^n}{12(1 - \xi_2)}\right)$$

$$+ \frac{\alpha\Delta t}{4(1 - \xi_2)}\right) - \left(\frac{1}{6} + \frac{\Delta t}{2hc^2}\right)u_{nt-1}^n - \left(\frac{1}{3} - \frac{\Delta t}{2hc^2} - \frac{\beta\Delta t}{2hc}\right)u_{nt}^n = 0$$

In Eq.(4.33), since u_1 is very close to 1 at the beginning, a Taylor series expansion suggest that we use ξ_2 to replace

$$\frac{\xi_2 + \Delta\xi_2}{10(1 - u_1^{n+1})} \left(e^{-10(u_1^n - 1)} + \frac{1 - e^{-10(u_1^{n+1} - 1)}}{10(1 - u_1^{n+1})}\right) \\ + \frac{\xi_2}{10(1 - u_1^n)} \left(e^{-10(u_1^n - 1)} + \frac{1 - e^{-10(u_1^n - 1)}}{10(1 - u_1^n)}\right)$$

A similar method was used in Eq.(4.34). For later times the original equations are used. It is very clear that system is nonlinear in $(u_1, \Delta \xi_2, u_3, \ldots, u_{nt})$. Here it must be noted that the second unknown $\Delta \xi_2$ is the increment of ξ_2 since u_2 is always assumed to be 1. In order to solve equations (4.33-4.37), Newton's method is again applied. Since the variable $\Delta \xi_2$ is different from others, some special techniques are used to solve for it. If the system is denoted as $\mathbf{F}(u_1, \Delta \xi_2, u_3, \ldots, u_{nt}) = 0$ where $u = (u_1, \Delta \xi_2, u_3, \ldots, u_{nt})^T$, the Newtonian iterative scheme can be written as $u^{n+1} = u^n - (\nabla \mathbf{F}(u^n))^{-1}\mathbf{F}(u^n)$. Let $\Delta u = u^{n+1} - u^n$, hence $[\nabla \mathbf{F}(u^n)]\Delta u = -\mathbf{F}(u^n)$ must be solved. First let us examine the form of Jacobian $\nabla \mathbf{F}(u^n)$. Obviously most of the entries are zero, and for convenience a_{ij} means that entry is not zero. The second column corresponds to $\Delta \xi_2$.

$$A = \begin{pmatrix} a_{11} & a_{12} & 0 & 0 & 0 & 0 & \dots & 0 \\ a_{21} & a_{22} & a_{23} & 0 & 0 & 0 & \dots & 0 \\ 0 & a_{32} & a_{33} & a_{34} & 0 & 0 & \dots & 0 \\ 0 & a_{42} & a_{43} & a_{44} & a_{45} & 0 & \dots & 0 \\ 0 & a_{52} & 0 & a_{54} & a_{55} & a_{56} & \dots & 0 \\ \vdots & \vdots & \vdots & & \ddots & \ddots \\ 0 & a_{nt2} & 0 & 0 & 0 & 0 & \dots & a_{ntnt} \end{pmatrix}$$

$$(4.38)$$

The problem then is to solve a system Ax = b where $x = (x_1, x_2, ..., x_{nt})^T$ and $b = (b_1, ..., b_{nt})^T$. Denote $a_2 = (a_{32}, ..., a_{nt2})^T$, $b^1 = (b_3, ..., b_{nt})^T$,

$$C = \begin{pmatrix} a_{33} & a_{34} & 0 & 0 & \dots & 0 \\ a_{43} & a_{44} & a_{45} & 0 & \dots & 0 \\ 0 & a_{54} & a_{55} & a_{56} & \dots & 0 \\ \vdots & & \ddots & \ddots & \\ 0 & 0 & 0 & 0 & \dots & a_{ntnt} \end{pmatrix}$$
(4.39)

 $x^1 = (x_3^1, \ldots, x_{nt}^1)^T$ and $x^2 = (x_3^2, \ldots, x_{nt}^2)^T$. Solve $Cx^1 = b^1$ and $Cx^2 = a_2$ first (where C is a tridiagonal matrix). Then let $x = (x_1, x_2, x_3^1 - x_2 x_3^2, \ldots, x_{nt}^1 - x_2 x_{nt}^2)^T$ and choose x_1 and x_2 so that the remaining equations of the systems Ax = b are satisfied. This method allows the Δu to be found, and hence u^{n+1} . Repeat this until either $\Delta \xi_2 \leq 0$ or $\xi_2 \geq \frac{1}{nt}$. If $\Delta \xi_2 \leq 0$ for the first time step after the introduction of the interface, this means that the interface point ξ_2 is almost equal to 0, i.e., the temperature reaches its critical value (which here is 1) only at the point x = 0. If $\xi_2 > \frac{1}{nt}$, then $[0, \xi_2]$ is partitioned into two pieces. As ξ_2 increases, more node points will be added to the interval $[0, \xi_2]$, i.e., this interval is partitioned into ns equal segments.

In the actual algorithm, when the temperature at the center point becomes "hot", the program uses four subroutines. The first subroutine is devoted to the case in which only one point is a hot point. As time increases, ξ_2 increases. When ξ_2 is greater than h, more nodal points will be added, then the next subroutine is called. The second subroutine deals with the case where there are more than one mesh point on both sides of the interface point. If ξ_2 keeps moving forward, the number of mesh points at the right side of interface point reduce to one. Hence, the third subroutine is engaged. When ξ_2 is very close to the right end point (x = 1), it is considered that the whole region is within the transient region. Therefore, the fourth subroutine is applied. Repeat using the fourth subroutine until the absolute value of biggest difference between newly obtained values and old values at all points satisfies a specified tolerance and then stop.

α	β	<i>u</i> (0)	<i>u</i> (1)	v(0)	v(1)	error
7.50	0.250	1.362675	1.240080	1.362729	1.240063	0.00005
3.50	0.250	1.289988	1.172719	1.290052	1.172717	0.00006
2.50	0.250	1.257984	1.143097	1.258052	1.143100	0.00006
1.00	0.250	1.171109	1.062799	1.171199	1.062828	0.00009
0.90	0.250	1.161152	1.053608	1.161241	1.053635	0.00008
0.80	0.250	1.150019	1.043332	1.150117	1.043369	0.00009
0.70	0.250	1.137422	1.031710	1.137516	1.031741	0.00009
0.50	0.250	1.105845	1.002560	1.106811	1.002506	0.00005
0.40	0.250	1.084141	0.981651	1.084082	0.981590	0.00005
0.35	0.250	1.069982	0.966818	1.069938	0.966847	0.00004
0.30	0.250	1.051984	0.947115	1.051951	0.947085	0.00003

Table 4.	2.	1
----------	----	---

End point values of solutions for various α , β .

u is the solution of the time dependent problem as $t \longrightarrow \infty$.

v is the numerical solution of the steady state problem.

In the first three subroutines, similar methods are used to solve $[\nabla \mathbf{F}(u^n)]\Delta u =$

 $-\mathbf{F}(u^n)$. The criterion for adding and removing nodal points is as follows: when $\xi_2 \leq 0.5$ and $\xi_2 \geq \frac{ns \times h}{2}$, the nodal points between 0 and ξ_2 are doubled. In this way, when ξ_2 is very close to the middle point, the number of nodal points between 0 and ξ_2 is the same as the total number which is taken initially when the whole region was considered as a cold region. When $1 - \xi_2 \leq \frac{nc \times h}{2}$, $\xi_2 \geq 0.75$ and nc > 1, half of nodal points are removed. This means when ξ_2 moves towards 1, less and less nodal points between ξ_2 and 1 remain, the number of nodal points are determined by how close ξ_2 is to 1. When the sum of increment $\xi_2 + \Delta \xi_2$ is greater than 1, the whole region is considered as a transient region. The timestep Δt is also adjusted according to the ratio of previous values of $\Delta \xi_2$ and Δt .

The Table 4.2.1 gives the results obtained using the above mentioned program, where u(0) and u(1) represent numerical solutions of Eq.(4.16) at points 0 and 1 respectively after the steady state conditions have been attained and v(0) and v(1) the corresponding numerical solutions of the steady state problem (4.1) corresponding to Eq.(4.16). "Error" means the maximum error of the two numerical solutions corresponding to Eq.(4.16) and the steady state problem (4.1). From the Table 4.2.1 it is easy to see that biggest error is 0.00009 and smallest 0.00003. Also from Table 4.2.1 it can be seen that when α is between 0.4 and 0.3 inclusively, two phases appear. One phase is for the cold region and another one is for the transient region. This means that for those α the whole region is not in the transient state and therefore the interface point ξ_2 needs to be located.

The Table 4.2.2 gives the results obtained by two methods. One is the program mentioned above and the other is the IMSL shooting method. From the results it can be seen that the outcomes match very well. The value of ξ_2 on the fifth column

is obtained by the above mentioned program and the approximated interval of ξ_2 on the eighth column is obtained by using the IMSL subroutine for the steady state problem.

α	β	u(0)	u(1)	ξ2	v(0)	v(1)	ξ2
0.40	0.25	1.084141	0.981651	0.920	1.084082	0.981590	[0.88,0.92]
0.35	0.25	1.069982	0.966818	0.846	1.069938	0.966847	[0.84,0.88]
0.30	0.25	1.051984	0.947115	0.731	1.051951	0.947085	[0.72,0.76]

Table 4.2.2

Position of the "interface" point of ξ_2 from the time dependent solution as . $t \longrightarrow \infty$ and from the steady state solution.

Also through the program, ξ_2 can be traced during the whole time period before the solution reaches steady state. For some α and β pictures are given for demonstration. From the pictures it is easy to see that for fixed β , when α increases, the time to become hot decreases and the time period for the temperature to reach a pseudo-steady state is also shortened. When the temperature reaches pseudo-steady state, it still needs some Newton iterations for the temperature to reach steady state. The results from the pictures match very well with the theoretical conclusion, i.e., when α is bigger, which means the heat source is stronger, the temperature increases rapidly and reaches the steady state very quickly. Otherwise, the temperature increases very slowly and it takes a longer time to reach steady state.

4.2.2 Fixed Meshpoints Method

A fixed meshpoints method is now used to solve (4.16) where f(u) is defined by (4.2). Here the unconditionally stable Crank-Nicholson difference scheme is used throughout. The x direction stepsize is $\Delta x = \frac{1}{32}$ (we choose this number for comparison with the results of §4.2.1) and time direction stepsize Δt is chosen arbitrarily (of course $r = \frac{\Delta t}{(\Delta x)^2}$ is kept reasonably bounded). The scheme is as follows:

$$\frac{u_{i,j+1} - u_{i,j}}{\Delta t} = \frac{1}{2} \left[\frac{u_{i+1,j+1} - 2u_{i,j+1} + u_{i-1,j+1}}{(\Delta x)^2} + \frac{u_{i+1,j} - 2u_{i,j} + u_{i-1,j}}{(\Delta x)^2} \right] + \alpha f(u_{i,j})$$

i.e.,

$$-\frac{r}{2}u_{i+1,j+1} + (1+r)u_{i,j+1} - \frac{r}{2}u_{i-1,j+1}$$
$$= \frac{r}{2}u_{i+1,j} + (1-r)u_{i,j} + \frac{r}{2}u_{i-1,j} + \alpha\Delta t f(u_{i,j})$$
(4.40)

where $r = \frac{\Delta t}{(\Delta x)^2}$, i = 0, ..., N. For the boundary conditions, a similar central difference at endpoints x = 0 and x = 1 is used as in §4.2.1. That is

$$\begin{cases} \frac{u_{i+1,j}-u_{i-1,j}}{2\Delta x} = 0 , u_{1,j} = u_{-1,j} , i = 0 \\ \frac{u_{N+1,j}-u_{N-1,j}}{2\Delta x} + \beta u_{N,j} = 0 , u_{N+1,j} = u_{N-1,j} - 2\Delta x \beta u_{N,j} , i = N \end{cases}$$

$$(4.41)$$

The advantage of the fixed meshpoints method is that the scheme is easy to form and the numerical solutions are easy to obtain. Actually, the system obtained by discretization is a tridiagonal system. So it is very easy to solve. The disadvantage of this method is that there is no indication where the interface point ξ , where

 $u(t,\xi) = 1$, is located. The numerical results are summarized in Table 4.2.3 where u(0) and u(1) is the numerical solution of (4.16) at endpoints x = 0 and x = 1 respectively and v(0) and v(1) is the numerical solution of (4.1) at endpoints x = 0 and x = 1 and x = 1 respectively.

α	β	u(0)	u(1)	v(0)	v(1)	error
7.50	0.250	1.362750	1.240166	1.362729	1.240063	0.00010
3.50	0.250	1.290071	1.172801	1.290052	1.172717	0.00008
2.50	0.250	1.258070	1.143189	1.258052	1.143100	0.00009
1.00	0.250	1.171216	1.062906	1.171199	1.062828	0.00008
0.90	0.250	1.161258	1.053703	1.161241	1.053635	0.00007
0.80	0.250	1.150134	1.043435	1.150117	1.043369	0.00007
0.70	0.250	1.137532	1.031807	1.137516	1.031741	0.00007
0.50	0.250	1.105827	1.002569	1.106811	1.002506	0.00098
0.40	0.250	1.084067	0.981579	1.084082	0.981590	0.00002
0.35	0.250	1.069945	0.966875	1.069938	0.966847	0.00003
0.30	0.250	1.051941	0.947079	1.051951	0.947085	0.00001

\mathbf{Tabl}	e 4.	2.3
-----------------	------	-----

End point values of solutions.

u is the solution of the time dependent solution as $t \longrightarrow \infty$.

v is the steady state solution.

Comparing with Table 4.2.1, it is seen that the errors with the fixed meshpoint method are almost ten times bigger than the errors with the moving meshpoint method. The biggest error in Table 4.2.3 (for parameters $\alpha = 0.5$ and $\beta = 0.25$) is 0.00098 while the biggest in Table 4.2.1 is 0.00009. That is to say, though the moving meshpoints method is more complicated than the fixed meshpoints method, the moving meshpoints method gives more accurate numerical solutions. Also the moving meshpoints method gives a more accurate location of interface point than the fixed meshpoints method, which only gives a possible interval in which the interface point is located.

The above numerical solution is only for α fixed. Now the situation for α is function of u(x,t) is considered in the following. For convenience, the time dependent equation is written as follows:

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \alpha f(u) &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.42)

where β , μ and γ are parameters, f(u) is defined by (4.2) and α is a function of u, i.e., $\alpha = \frac{\gamma}{(1+\mu\int_0^1 f(u)dx)^2}$. This arises when the external circuit is included. A reasonable value for γ is 150 and for μ a value of 20. Comparing with Eq.(4.16), Eq.(4.42) is more complicated since α is dependent on u(x,t). In order to solve Eq.(4.42), a semi-implict difference scheme is used. Here semi-implicit means that the linear part is implicit but the nonlinear part is explicit. This makes it easy to handle the integral term. That is

$$\frac{u_{i,j+1} - u_{i,j}}{\Delta t} = \frac{u_{i+1,j+1} - 2u_{i,j+1} + u_{i-1,j+1}}{(\Delta x)^2} + \frac{\gamma f(u_{i,j})}{integral}$$

i.e.,

$$-ru_{i+1,j+1} + (1+2r)u_{i,j+1} - ru_{i-1,j+1} = u_{i,j} + \frac{\gamma \Delta t f(u_{i,j})}{integral}$$
(4.43)

where $r = \frac{\Delta t}{(\Delta x)^2}$ and $i = 0, 1, \dots, N$, $integral = [1 + \mu \Delta x_{\frac{1}{2}} \sum_{i=2}^{N} [f(u_{i-1,j}) + f(u_{i,j})]]^2$ i.e., a trapezoidal rule over the values of u at the gridpoints (time step j) is used. For the boundary condition, the central difference (4.41) is employed. Since in Eq.(4.42) α is no longer constant, the numerical experiments are done for various β . As it is mentioned in Chapter 3, as time increases, Eq.(4.42) reaches its steady state, therefore the steady state solutions of Eq.(4.42) should match the solutions of Eq.(4.11). Thus for different β there should exist one, two or three solutions to Eq.(4.42). The numerical results are summarized in following Table 4.2.4, where $I = \sqrt{\frac{\alpha}{\gamma}}$ and $\alpha = \frac{\gamma}{(1+\mu\int_0^1 f(u)dx)^2}$, u(0) and u(1) represent the numerical solutions (at both endpoints x = 0 and x = 1) obtained when the initial values $u_{i,0}$ are all zeros; v(0) and v(1) represent the numerical solutions obtained when the initial values $u_{i,0}$ are greater than 1.0. Theoretically, when $\beta > 0.4098$ there should be three solutions, one is that the whole solution is in the cold region, the second one is a solution spanning both cold and transient regions, the third one is the whole solution in the transient region; when $\beta < 0.4098$ there is only one solution in the transient region. From the Table 4.2.4 it is easy to see that when β < 0.4098 (critical value for β), there is only one numerical solution; when $\beta > 0.4098$, there are two solutions, one is the whole solution in the cold region and another one is the whole solution in the transient region. How about the one which spans both cold and transient region? Why are we unable to get it numerically? The problem is that that steady state solution is unstable. As for the case $\beta = 0.4098$, v(0) reaches 1.0

for the cold branch and it agrees with the numerical results in §4.1.2.

β	I	u(0)	u(1)	I	v(0)	v(1)
.6000	.4762E-1	.7369E+0	.5668E+0	.8771E+0	.1588E+1	.1348E+1
.5000	.4762E-1	.8503E+0	.6802E+0	.8974E+0	.1599E+1	.1381E+1
.4500	.4762E-1	.9258E+0	.7558E+0	.9074E+0	.1605E+1	.1400E+1
.4100	.4762E-1	.9996E+0	.8295E+0	.9154E+0	.1611E+1	.1416E+1
.4099	.4762E-1	.9998E+0	.8297E+0	.9155E+0	.1611E+1	.1416E+1
.4098	.4762E-1	.1000E+1	.8299E+0	.9155E+0	.1611E+1	.1416E+1
.4000	.9174E+0	.1612E+1	.1420E+1	.9174E+0	.1612E+1	.1420E+1
.3500	.9274E+0	.1620E+1	.1443E+1	.9274E+0	.1620E+1	.1443E+1
.3000	.9373E+0	.1629E+1	.1469E+1	.9373E+0	.1629E+1	.1469E+1
.2500	.9473E+0	.1640E+1	.1498E+1	.9473E+0	.1640E+1	.1498E+1
.2000	.9573E+0	.1654E+1	.1533E+1	.9573E+0	.1654E+1	.1533E+1

Table 4.2.4

End point values of solutions of time dependent problem with external circuit as

 $t \longrightarrow \infty$. u and v are solutions obtained from different initial values.

4.3 Numerical Solutions for NTC Problem

In this section, numerical solutions for the NTC problems are given. As in § 4.2, the figures and tables are also included. Numerical results show that the characteristics of solutions of Eq.(3.1) and Eq.(3.17) are mainly determined by the nondecreasing or nonincreasing property of the functions at the right hand side.



Plot of u(1) and I for solution of (4.45), $0 \le u(1) \le 2$.

4.3.1 Without External Circuit

As for NTC problem, the function for right hand side is defined as (reciprocal of f defined by (4.2))

$$g(u) = \begin{cases} 1 & , u \leq 1 \\ e^{10(u-1)} & , 1 < u < 2 \\ e^{10} & , 2 \leq u \end{cases}$$
(4.44)



Plot of u(1) and I for solution of (4.45), $2 \le u(1) \le 20000$ (large range).

and the problem is as follows

$$\begin{cases} \frac{d^2 u}{dx^2} + \alpha g(u) = 0 , 0 < x < 1 \\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.45)



Graphs of u(1) and I for solution of (4.50) and solutions of (4.52). Intersection point represents solution of coupled problem.

Comparing to the original problem discussed in §4.1.1, the only difference is the nonlinear function g(u). However this difference of functions makes the solution u(x) very different. That is, for some fixed α and β , the solution u(x) is not unique. Again

as in §4.1.1,

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2)$$
 (4.46)

iff $u \leq 1$ in the interval [0, 1] and

$$u(x) = \frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{2}(1 - x^2)$$
(4.47)

iff $u \ge 2$ in the interval [0, 1]. For (4.46) if $u(0) \le 1$ then

$$\alpha \le \frac{1}{\frac{1}{\beta} + \frac{1}{2}} = \frac{2\beta}{2+\beta} \tag{4.48}$$

and for (4.47) if $u(1) \ge 2$ then

$$\alpha \ge 2\beta e^{-10} \tag{4.49}$$

Hence for $\alpha \in [2\beta e^{-10}, \frac{2\beta}{2+\beta}]$, (4.46) and (4.47) can both be solutions of (4.45). These are the easily found ones. In fact there exists u(x) between 1 and 2. Thus for certain α and β there exist three solutions. For demonstration, $\beta = 0.25$ is fixed, hence for $\alpha \in [0.000022699, 0.222222222]$, there exist three solutions. For convenience, the graphs demonstrate the relations of u(1) and α (for consistency, here $\alpha = \gamma I^2, \gamma =$ 150). The graphs are called Figure 4.3.0 and Figure 4.3.1.

4.3.2 With External Circuit

Here the steady state one dimensional problem for the NTC with the external circuit is considered. The numerical solution is discussed. The equation is as follows: to find I and u(x), 0 < x < 1 such that

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha g(u) = 0 , 0 < x < 1 \\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.50)

CHAPTER 4. NUMERICAL RESULTS

$$1 = \mu I + \frac{\beta u(1)}{\gamma I}, \qquad \alpha = \gamma I^2 \tag{4.51}$$

where γ and β are fixed parameters and g(u) is defined by (4.44).

From Eq.(4.50) and (4.51) it can be seen that $\alpha = \gamma I^2$ and u(1) are related. Thus it is more complicated to solve Eq.(4.50) and (4.51) than (4.45). In order to get the numerical solution of Eq.(4.50) and (4.51), the IMSL routine is used as in the PTC problem. So the strategy here is to fix *I*, therefore α , use the shooting method (usually with more than one iteration) to obtain u(1), then modify α according to u(1) by using the relation (4.51) and continue iterating, until both *I* and u(1) converge to some values and a numerical solution to Eq.(4.50) and (4.51) is obtained.

From the equation of Eq.(4.51), it is easy to see that

$$u(1) = \frac{\gamma I}{\beta} (1 - \mu I) \tag{4.52}$$

so u(1) is a parabolic function of *I*. On the other hand, from Eq.(4.50) and the definition of g(u) by (4.44), the following are obvious

$$u(x) = \frac{\alpha}{\beta} + \frac{\alpha}{2}(1 - x^2), \text{ provided } u \le 1 \text{ or } \alpha < \frac{2\beta}{\beta + 2}$$

$$(4.53)$$

and

$$u(x) = \frac{\alpha e^{10}}{\beta} + \frac{\alpha e^{10}}{2}(1 - x^2), \text{ provided } 2 \le u \text{ or } \alpha > 2\beta e^{-10}$$
(4.54)

Now we see that if γ and β are fixed, the solution to Eq.(4.50) and (4.51) for any given I is not unique. Actually, the region of I for nonuniqueness can be determined as follows. From (4.3), if $u(0) = \frac{\alpha}{\beta} + \frac{\alpha}{2} \leq 1$, then $u(x) \leq 1$ for $0 \leq x \leq 1$ because u(x) is nonincreasing function of x. Thus, $\alpha = \gamma I^2 \leq \frac{1}{\frac{1}{\beta} + \frac{1}{2}}$, i.e., $I^2 \leq \frac{2\beta}{\gamma(\beta+2)}$ (e.g.,

if $\beta = 0.25, \gamma = 150$, then $I \leq 0.03849$). From (4.54), it is easy to see that, if $u(1) = \frac{\gamma I^2 e^{10}}{\beta} \ge 2$, then $u(x) \ge 2$ for $0 \le x \le 1$. Hence, $\frac{\gamma I^2 e^{10}}{\beta} \ge 2$, i.e., $I^2 \ge \frac{2\beta}{\gamma e^{10}}$ (e.g., if $\beta = 0.25, \gamma = 150$, then $I \ge 0.0003890$). Therefore, for $I \in [\frac{2\beta}{\gamma e^{10}}, \frac{2\beta}{\gamma(\beta+2)}]$, the number of solutions to Eq.(4.50) and (4.51) is at least two. The Figure 4.3.0 and Figure 4.3.1 are demonstration graphs for the nonuniqueness corresponding to I. The graphs consist of three parts, i.e., Figure 4.3.0 for $u(1) \le 1, 1 \le u(1) \le 2$ and Figure 4.3.1 for $2 \le u(1)$ within the same region of *I*. That is to say, the graphs are drawn for the relation of u(1) to I. The numerical results are given in Table 4.3.1. Where $\beta = 0.25$ and $\gamma = 150$ are fixed, I is a parameter changed within the interval mentioned above with stepsize of 20th of the interval length, u(1) and u(0), v(1) and v(0), w(1) and w(0) represent the solutions with respect to $u(x) \ge 2, 2 \ge u(x) \ge 1$ and $1 \ge u(x)$. From Table 4.3.1, it can be seen that at both ends of the interval the two branches of the solution almost joined together. At I = 0.3890E - 3, the difference of u(1) and v(1) is 0.052, at I = 0.3850E - 1, the difference of v(1) and w(1) is 0.0003, much smaller and almost negligible. From Figure 4.3.0, Figure 4.3.2, Figure 4.3.3 and Figure 4.3.4, it is easy to see that the cold and transient solutions joined together at the right end of the interval. Thus the solution u(x) is apparently not differentiable with respect to the parameter α at the joint point. That means there is a wedge at the joint point.

If the values of the solutions are considered at the right end point x = 1 and α is replaced by γI^2 , then

$$u(1) = \frac{\gamma I^2}{\beta}, u \le 1 \tag{4.55}$$

Table 4.3.1

I	<i>u</i> (0)	<i>u</i> (1)	v(0)	v(1)	w(0)	w(1)
.3890E-3	.2250E+1	.2000E+1	.2197E+1	.1948E+1	.9079E-4	.1021E-3
.2295E-2	.7828E+2	.6958E+2	.1680E+1	.1438E+1	.3554E-2	.3159E-2
.4200E-2	.2623E+3	.2331E+3	.1546E+1	.1326E+1	.1191E-1	.1058E-1
.6106E-2	.5543E+3	.4927E+3	.1462E+1	.1257E+1	.2516E-1	.2237E-1
.8011E-2	.9542E+3	.8482E+3	.1401E+1	.1206E+1	.4332E-1	.3851E-1
.9917E-2	.1462E+4	.1300E+4	.1353E+1	.1166E+1	.6638E-1	.5901E-1
.1182E-1	.2078E+4	.1847E+4	.1313E+1	.1133E+1	.9434E-1	.8386E-1
.1373E-1	.2802E+4	.2491E+4	.1279E+1	.1105E+1	.1272E+0	.1131E+0
.1563E-1	.3634E+4	.3230E+4	.1250E+1	.1080E+1	.1650E+0	.1466E+0
.1754E-1	.4574E+4	.4065E+4	.1224E+1	.1058E+1	.2076E+0	.1846E+0
.1944E-1	.5621E+4	.4997E+4	.1200E+1	.1038E+1	.2552E+0	.2269E+0
.2135E-1	.6777E+4	.6024E+4	.1179E+1	.1020E+1	.3077E+0	.2735E+0
.2326E-1	.8041E+4	.7147E+4	.1159E+1	.1004E+1	.3651E+0	.3245E+0
.2516E-1	.9413E+4	.8367E+4	.1141E+1	.9887E+0	.4273E+0	.3799E+0
.2707E-1	.1089E+5	.9682E+4	.1124E+1	.9747E+0	.4945E+0	.4396E+0
.2897E-1	.1248E+5	.1109E+5	.1107E+1	.9616E+0	.5666E+0	.5036E+0
.3088E-1	.1418E+5	.1260E+5	.1090E+1	.9492E+0	.6436E+0	.5721E+0
.3278E-1	.1598E+5	.1420E+5	.1073E+1	.9372E+0	.7255E+0	.6448E+0
.3469E-1	.1789E+5	.1590E+5	.1055E+1	.9253E+0	.8122E+0	.7220E+0
.3659E-1	.1991E+5	.1770E+5	.1035E+1	.9123E+0	.9039E+0	.8035E+0
.3850E-1	.2204E+5	.1959E+5	.1001E+1	.8896E+0	.1001E+1	.8893E+0

End point values for multiple solutions u, v, w of the steady state NTC problem.

.

and

$$u(1) = \frac{\gamma I^2 e^{10}}{\beta}, 2 \le u \tag{4.56}$$



Graphs of u(1) and I for solution of (4.50) and solutions of (4.52). Intersection points are solutions of coupled problem.

Now consider the solution of Eq.(4.50) and (4.51), that is, condition (4.52) is in force for Eq.(4.45). As it is seen, (4.52) is a parabola, which has two intersection points, I = 0 and $I = \frac{1}{\mu}$, when u(1) = 0. The maximum of u(1) is reached at $I = \frac{1}{2\mu}$, and the maximum value of u(1) is $\frac{\gamma}{4\beta\mu}$ which depends on β if γ and μ are fixed. If (4.52) is drawn on Figure 4.3.0, it is obvious that, except for the trivial solution (i.e., $\alpha = 0$ and u = 0), there are three possible cases. Case one, there is only one solution, case two there are two solutions, case three there are three solutions. Since (4.55) is also a parabola about *I*, it is easy to see that, if (4.55) for some β just reaches (4.52) with u(0) < 1 on the $I > \frac{1}{2\mu}$ side, there are exactly two solutions, if (4.55) does not reach (4.52), that means only one solution, if (4.55) intersects with (4.52) and goes outside that parabola, then there are three solutions. In fact, the



Graphs of u(1) and I for solution of (4.50) and solutions of (4.52). Intersection points are solutions of coupled problem.

region of β can be determined for the three cases. Consider that (4.55) just reaches

(4.52), hence $\frac{\gamma I^2}{\beta} = \frac{\gamma I}{\beta}(1-\mu I)$, then $I = \frac{1}{\mu+1}$. From (4.3), $u(0) = \frac{\alpha}{\beta} + \frac{\alpha}{2}$ and u(0) can be at most 1, thus $\frac{\alpha}{\beta} + \frac{\alpha}{2} = 1$, i.e., $\beta = 0.409836065 = \beta_0$. We say that this value is critical value for β . Obviously, (4.3) is true for all $u(0) \leq 1$, especially for u(0) = 1, hence $\alpha = \frac{2\beta}{\beta+2} = 2 - \frac{4}{\beta+2}$, which means α is an nondecreasing function of β . Denote $I_0 = \frac{1}{\mu+1}$ (also $\alpha_0 = \gamma I_0^2$) corresponding to β_0 , then if $\beta > \beta_0$, $I = \sqrt{\frac{\alpha}{\gamma}} > I_0$, which means that right end of (4.55) goes outside of (4.52). So the curves shown in Figure 4.3.0 and Figure 4.3.1 should have three intersection points. The Figure 4.3.2, Figure 4.3.3 and Figure 4.3.4 demonstrate this.

In the above only the intersection of the right end of the "all cold" branch meeting with the parabola (4.52) is discussed. For the all hot branch a similar result can be considered.

4.4 Time Dependent Problem for NTC

For completeness, the numerical solution for the time dependent NTC problem is also done. As it is discussed in § 4.3, the NTC problem has three solutions. For the steady state NTC problem all three solutions can be numerically obtained. However, for the time dependent NTC problem, whenever it has three solutions, one of them is unstable. In fact, starting with initial value u = 0, the numerical solution always converges to all cold solution for $\alpha \in [2\beta e^{-10}, \frac{2\beta}{2+\beta}]$. Starting with initial value u > 2, the numerical solution always converges to the all hot solution for $\alpha \in [2\beta e^{-10}, \frac{2\beta}{2+\beta}]$, in which interval there exist three solutions. Of course, for α is outside that interval there is only one solution either cold or hot. Hence the numerical solution always converges to the corresponding cold or hot solution even if the initial value is between 0 and 2. Thus for $\alpha \in [2\beta e^{-10}, \frac{2\beta}{2+\beta}]$, the third solution is unstable. It can not be numerically obtained. Therefore, the moving meshpoint method is not tried since only the simple all hot or all cold solutions are obtained. For the case with external circuit connected, the results are given in § 4.7. The instability is briefly discussed in Chapter 5.

4.5 Numerical Solutions for Smooth Functions

The given functions for f(u) taken from [3] have discontinuous first derivatives at u = 1 and u = 2. For comparison, numerical results are also obtained for smooth functions, that is, the functions have continuous first derivatives. In § 4.5.1, a smooth function corresponding to the PTC problem is considered. In § 4.5.2, a smooth function related to the NTC problem is studied.

4.5.1 Smooth Function for PTC Problem

For comparison, the function $\sigma(u)$ and hence f(u) for problem (4.1) is modelled by a smooth function. The choice of smooth function is a cubic Hermite interpolation for the function $\sigma = 1$ for $u \leq 1$ and $\sigma = e^{-10}$ for $u \geq 2$. This is given by

$$p(u) = \begin{cases} 1 & , u < 1 \\ (2u-1)(u-2)^2 + e^{-10}(5-2u)(u-1)^2 & , 1 \le u \le 2 \\ e^{-10} & , 2 < u \end{cases}$$
(4.57)

and the derivative of p(u) is as follows

$$p'(u) = \begin{cases} 0 & , u < 1 \\ 6(1 - e^{-10})(u - 2)(u - 1) & , 1 \le u \le 2 \\ 0 & , 2 < u \end{cases}$$
(4.58)



Figure 4.5.1

Graph of nonlinear functions for PTC problems.

The difference of p(u) and $\sigma(u)$ is that $\sigma(u)$ is only continuous but p(u) has continuous first derivative. Now the problem (4.1) can be written as

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha p(u) = 0 , 0 < x < 1 \\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.59)
In order to solve (4.59), the method discussed in §4.1.1 is applied. Similarly, the numerical solutions are summarized in Table 4.5.1. Comparing with Table 4.1.1, it is easy to find that when $1 \le u \le 2$, the numerical solutions of (4.59), with α , β the same, are greater than the solutions of (4.1). In Table 4.5.1 column 6 represents the number of Newton iterations, column 7 represents the number of times that bisection was used, column 8 represents the number of times the bisection routine was called, s represents the initial guess. * means total number of bisection times.

. α	β	s	u(0)	<i>u</i> (1)	6	7	8
999.000	0.005	0.5	9.0936	9.0709	2	40	1
999.000	0.010	0.5	4.5581	4.5355	2	34	1
999.000	0.100	0.5	2.0072	1.9752	1	31	1
750.000	0.100	0.5	2.0040	1.9669	1	31	1
500.000	0.100	0.5	1.9985	1.9679	10	94*	4
100.000	0.100	0.5	1.9884	1.9432	5	59*	2
75.000	0.100	0.5	1.9840	1.9363	5	29	1
50.000	0.100	0.5	1.9802	1.9285	1	29	1
25.000	0.100	0.5	1.9667	1.9127	1	25	1
15.000	0.100	0.5	1.9534	1.8950	1	23	1
14.000	0.100	0.5	1.9509	1.8940	1	22	1
10.000	0.100	0.5	1.9412	1.8721	10	18	1
7.500	0.100	0.5	1.9293	1.8582	12	19	1
5.000	0.100	0.5	1.9089	1.8356	7	18	1

Table 4.5.1

α	β	S	_ u(0)	u(1)	6	7	8
1.000	0.100	0.5	1.7644	1.6875	4	12	1
0.850	0.100	0.5	1.7415	1.6649	3	11	1
0.500	0.100	0.5	1.6504	1.5760	3	15	1
0.250	0.100	0.5	1.4794	1.4110	2	8	1
0.100	0.100	0.5	1.0465	0.9968	3	2	2*
0.050	0.100	0.5	0.5250	0.5000	2	1	1
0.005	0.100	0.5	0.0525	0.0500	2	1	1

Table 4.5.1(continued)

Solution of PTC problem for various α , β . Col. 6–No. of Newton iterations. Col. 7–No. of times bisection used. Col. 8–No. of calls to bisection routine.

4.5.2 Smooth Function for NTC Problem

For the NTC problem (4.45) with function f(u) defined by (4.44), the smooth function for (4.44) is not the directly Hermitian interpolation of (4.44) but the reciprocal of (4.57), that is

$$q(u) = \begin{cases} 1 & , u < 1 \\ \frac{1}{(2u-1)(u-2)^2 + e^{-10}(5-2u)(u-1)^2} & , 1 \le u \le 2 \\ e^{10} & , 2 < u \end{cases}$$
(4.60)

and the derivative of q(u) is as follows

$$q'(u) = \begin{cases} 0 & , u < 1 \\ -\frac{6(1-e^{-10})(u-2)(u-1)}{[(2u-1)(u-2)^2 + e^{-10}(5-2u)(u-1)^2]^2} & , 1 \le u \le 2 \\ 0 & , 2 < u \end{cases}$$
(4.61)



Figure 4.5.2

Graph of nonlinear functions for NTC problems.

Thus the problem (4.45) becomes

$$\begin{cases} \frac{d^2u}{dx^2} + \alpha q(u) = 0 , 0 < x < 1\\ u_x(0) = 0 , u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.62)

7

Similarly, the numerical results are summarized in Table 4.5.2. Comparing with Table 4.3.1, it is easy to see that when $u \leq 1$ and $u \geq 2$, the solutions are the same; but for $1 \leq u \leq 2$, solutions in Table 4.5.2 are greater than those in Table 4.3.1 with the same α and β . However, because of the smoothness of q(u), the properties of the solution have been changed accordingly. From Table 4.3.1, the three branches of the solution can join together at two ends separately. However, from Table 4.5.2,



Plots of u(1) and I for solution of (4.60) and (4.62).

we can see different results. The difference of the values at the right end of the cold and transient solutions is quite different. Also the same method is not convergent 1

for I beyond I = 0.385E - 1. An alternative method is used, that is, fixing u(0) to find α by the shooting method. Now for fixed β , u(0), denote $\overline{u} = \frac{du}{d\alpha}$, $\overline{v} = \frac{dv}{d\alpha}$ as two new unknowns, use the similar method as in §4.1.1. A new system of differential equations can be obtained from (4.62) as follows:

$$u_{x} = v , u(0) = s$$

$$v_{x} = -\alpha^{n} f(u) , v(0) = 0$$

$$\overline{u}_{x} = \overline{v} , \overline{u}(0) = 0$$

$$\overline{v}_{x} = -\alpha^{n} f'(u)\overline{u} - f(u) , \overline{v}(0) = 0$$

$$(4.63)$$



Plots of u(1) and I for solution of the smooth NTC problem using the shooting method (4.63).

where s is fixed, $\alpha^{n+1} = \alpha^n + \Delta \alpha$, $\Delta \alpha = -\frac{v(1,\alpha^n) + \beta u(1,\alpha^n)}{\overline{v}(1,\alpha^n) + \beta \overline{u}(1,\alpha^n)}$, $n = 0, 1, 2, \dots, \alpha^0$ is arbitrary. Here a numerical method similar to that in § 4.1.1 is used to get α . The results are summarized in Table 4.5.3, where u(0) is changed by small stepsize then different values for α are obtained. Using the data obtained in Table 4.5.3, a graph is drawn as Figure 4.5.4 and also Figure 4.5.3 is drawn according to the data in Table 4.5.2 (note, for consistency, here the x coordinate has been changed to I, where $\alpha = \gamma I^2$, $\gamma = 150$). Comparing these two graphs, it is easy to see that two graphs



Figure 4.5.5

Combined plot of u(1) and I for the smooth NTC problem.

can be connected. The graph in Figure 4.5.4 is just that part missed in Figure 4.5.3.

Figure 4.5.5 is the combination of Figure 4.5.3 and Figure 4.5.4. The two graphs joined perfectly. Also from Figure 4.5.4, it seems that the derivatives of $u(1, \alpha)$ with respect to α is infinite at the end of the interval for *I*. So that's why we need to use

Ι	<i>u</i> (0)	<i>u</i> (1)	v(0)	<i>v</i> (1)	w(0)	w(1)
.2295E-2	.7828E+2	.6958E+2	.1998E+1	.1616E+1	.3554E-2	.3159E-2
.4200E-2	.2623E+3	.2331E+3	.1990E+1	.1611E+1	.1191E-1	.1058E-1
.6106E-2	.5543E+3	.4927E+3	.1978E+1	.1619E+1	.2516E-1	.2237E-1
.8011E-2	.9542E+3	.8482E+3	.1958E+1	.1627E+1	.4332E-1	.3851E-1
.9917E-2	.1462E+4	.1300E+4	.1944E+1	.1625E+1	.6638E-1	.5901E-1
.1182E-1	.2078E+4	.1847E+4	.1924E+1	.1622E+1	.9434E-1	.8386E-1
.1373E-1	.2802E+4	.2491E+4	.1897E+1	.1617E+1	.1272E+0	.1131E+0
.1563E-1	.3634E+4	.3230E+4	.1876E+1	.1605E+1	.1650E+0	.1466E+0
.1754E-1	.4574E+4	.4065E+4	.1852E+1	.1592E+1	.2076E+0	.1846E+0
.1944E-1	.5621E+4	.4997E+4	.1825E+1	.1576E+1	.2552E+0	.2269E+0
.2135E-1	.6777E+4	.6024E+4	.1783E+1	.1562E+1	.3077E+0	.2735E+0
.2326E-1	.8041E+4	.7147E+4	.1758E+1	.1538E+1	.3651E+0	.3245E+0
.2516E-1	.9413E+4	.8367E+4	.1730E+1	.1514E+1	.4273E+0	.3799E+0
.2707E-1	.1089E+5	.9682E+4	.1698E+1	.1487E+1	.4945E+0	.4396E+0
.2897E-1	.1248E+5	.1109E+5	.1662E+1	.1458E+1	.5666E+0	.5036E+0
.3088E-1	.1418E+5	.1260E+5	.1623E+1	.1426E+1	.6436E+0	.5721E+0
.3278E-1	.1598E+5	.1420E+5	.1580E+1	.1390E+1	.7255E+0	.6448E+0
.3469E-1	.1789E+5	.1590E+5	.1531E+1	.1349E+1	.8122E+0	.7220E+0

Table 4.5.2

Ι	<i>u</i> (0)	<i>u</i> (1)	v(0)	v(1)	w(0)	w(1)
.3659E-1	.1991E+5	.1770E+5	.1473E+1	.1300E+1	.9039E+0	.8035E+0
.3850E-1	.2204E+5	.1959E+5	.1399E+1	.1237E+1	.1001E+1	.8893E+0

Table 4.5.2 (continued)

End point values of the multiple solution u, v, w of the steady state NTC problem with (4.60).

	·····	· · · · · · · · · · · · · · · · · · ·			
<i>u</i> (0)	<i>u</i> (1)	α	u(0)	u(1)	α
0.99500000	0.88444442	0.22111109	1.17499995	1.04273701	0.24741493
1.00000000	0.88888890	0.22222222	1.17999995	1.04710269	0.24757470
1.00500000	0.89333266	0.22333126	1.18499994	1.05146742	0.24770039
1.009999999	0.89777416	0.22443224	1.19000006	1.05583107	0.24779195
1.01499999	0.90221232	0.22552381	1.19500005	1.06019342	0.24784940
1.01999998	0.90664661	0.22660190	1.20000005	1.06455481	0.24787290
1.02499998	0.91107804	0.22766449	1.20500004	1.06891501	0.24786241
1.02999997	0.91550541	0.22871037	1.20500004	1.06891501	0.24786241
1.03499997	0.91992903	0.22973785	1.21000004	1.07327390	0.24781790
1.03999996	0.92434883	0.23074605	1.21500003	1.07763171	0.24773957
1.044999996	0.92876470	0.23173484	1.22000003	1.08198822	0.24762738
1.04999995	0.93317693	0.23270294	1.22500002	1.08634341	0.24748141

Table 4.5.3

u(0)	u(1)	α	u(0)	u(1)	α
1.05499995	0.93758607	0.23364586	1.23000002	1.09069741	0.24730173
1.1000002	0.97712988	0.24094184	1.23500001	1.09505010	0.24708843
1.14999998	1.02089286	0.24610418	1.24000001	1.09940147	0.24684161
1.15499997	1.02526367	0.24643457	1.24500000	1.10375130	0.24656127
1.15999997	1.02963352	0.24673088	1.25000000	1.10809994	0.24624757
1.164999996	1.03400230	0.24699304	1.29999995	1.15150189	0.24129973
1.169999996	1.03837013	0.24722104	1.35000002	1.19472814	0.23316069

Table 4.5.3(continued)

Values of u(1) and α for given u(0) using shooting method (4.63).

a different method to find the relation of u and α . A similar result can be obtained at the other end of the *I* interval.

4.6 Time Dependent Problems for Smooth Functions

Now the time dependent problems with smooth functions are considered. Numerical results for the PTC problem are considered first followed by those for the NTC problem.

4.6.1 Smooth Function for PTC Problem

There are two cases considered here. One is that the α and β are fixed. The other is that α depends upon u. Here the smooth function is defined by (4.57). For convenience, the time dependent problem with smooth function is written as

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \alpha p(u) &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$
(4.64)

Use the Crank-Nicholson difference scheme defined by (4.40) and (4.41). The numerical results are summarized in Table 4.6.1 where u(0) and u(1) is the steady state of the numerical solution of (4.64) at endpoints x = 0 and x = 1 respectively and v(0)and v(1) is numerical solution of (4.59) at endpoints x = 0 and x = 1 respectively. The biggest error in Table 4.6.1 is 0.00025 and the smallest error is 0.000004. Comparing with Table 4.2.3, it is easy to see that for the same parameters the values of u(0) and u(1) in Table 4.6.1 are greater than that in Table 4.2.3.

For the case when α depends on u(x,t), the equation is the same as (4.64) except that $\alpha = \frac{\gamma}{(1+\mu\int_0^1 p(u)dx)^2}$. The scheme (4.43) and boundary central difference scheme (4.41) are used. The numerical results are summarized in Table 4.6.2. Similarly as in § 4.2.2, $I = \sqrt{\frac{\alpha}{\gamma}}$ and $\alpha = \frac{\gamma}{(1+\mu \int_0^1 p(u)dx)^2}$, u(0) and u(1) represent the numerical solutions (at both endpoints x = 0 and x = 1) obtained with initial values $u_{i,0}$ all zeros; v(0) and v(1) represent the numerical solutions obtained with initial values $u_{i,0}$ greater than 1.0.

α	β	u(0)	u(1)	v(0)	v(1)	error
3.50	0.250	1.836574	1.674007	1.836824	1.674157	0.000250
2.50	0.250	1.797332	1.631792	1.797487	1.631837	0.000155
1.00	0.250	1.641450	1.476472	1.641456	1.476449	0.000023
0.90	0.250	1.617426	1.453506	1.617425	1.453485	0.000021
0.80	0.250	1.588639	1.426191	.1.588637	1.426177	0.000014
0.70	0.250	1.553291	1.392912	1.553296	1.392908	0.000004
0.50	0.250	1.448911	1.295864	1.448927	1.295889	0.000025
0.40	0.250	1.363989	1.217883	1.364019	1.217913	0.000030
0.35	0.250	1.304856	1.163926	1.304904	1.163972	0.000048
0.30	0.250	1.225620	1.091959	1.225690	1.092025	0.000070
0.25	0.250	1.106199	.984013	1.106335	0.984140	0.000136

Table 4.6.1

End point values of the time dependent solution as $t \longrightarrow \infty$ and of the steady state solution.

From Table 4.6.2, it is easy to see that when $\beta < 0.4098$, only one numerical solution is obtained; when $\beta \ge 0.4098$, there are two numerical solutions obtained. Actually, there are three steady state solutions when $\beta > 0.4098$. However, from

Table 4.6.2, only two solutions are obtained. The reason is the same as in § 4.2.2, i.e., the third steady state solution is unstable so it is not obtained numerically. Also, comparing Table 4.6.2 with Table 4.2.4, it is seen that, when $\beta < 0.4098$, the cold solutions are the same in Table 4.2.4 and Table 4.6.2. They should of course be the same since when $\beta < 0.4098$, u(0) and u(1) are values for the all cold solution. When $\beta < 0.4098$, the one solution in Table 4.6.2 is greater than that in Table 4.2.4.

β	I	u(0)	u(1)	I	v(0)	v(1)
.6000	.4762E-1	.7369E+0	.5668E+0	.8237E+0	.1984E+1	.1815E+1
.5000	.4762E-1	.8503E+0	.6802E+0	.8568E+0	.1985E+1	.1840E+1
.4500	.4762E-1	.9258E+0	.7558E+0	.8726E+0	.1986E+1	.1852E+1
.4100	.4762E-1	.9996E+0	.8295E+0	.8850E+0	.1986E+1	.1862E+1
.4099	.4762E-1	.9998E+0	.8297E+0	.8850E+0	.1986E+1	.1862E+1
.4098	.4762E-1	.1000E+1	.8299E+0	.8850E+0	.1986E+1	.1862E+1
.4000	.8706E+0	.1987E+1	.1865E+1	.8880E+0	.1987E+1	.1865E+1
.3000	.9176E+0	.1988E+1	.1890E+1	.9176E+0	.1988E+1	.1890E+1
.2500	.9319E+0	.1989E+1	.1904E+1	.9319E+0	.1989E+1	.1904E+1
.2000	.9459E+0	.1990E+1	.1917E+1	.9459E+0	.1990E+1	.1918E+1

Table 4.6.2

End point values of time dependent solutions as $t \longrightarrow \infty$ for different initial values.

4.6.2 Smooth Function for NTC Problem

As in § 4.6.1, two cases are considered. One case is that α is fixed and the other is that α is a functional of u(x,t). For convenience, the time dependent problem is written as

$$\frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \alpha q(u) \quad , 0 < x < 1$$

$$u_x(0) = 0 \qquad , u_x(1) + \beta u(1) = 0$$
(4.65)

where α and β are fixed constant parameters, q(u) is defined by (4.60). The numerical results are summarized in Table 4.6.3. Similarly as in § 4.3, there should be three solutions for $I \in [0.3890 \times 10^{-3}, 0.3850 \times 10^{-1}]$ (where $I = \sqrt{\frac{\alpha}{150}}$). However, only two solutions are obtained. The solution which is in transient region is unstable, so it is impossible to get it numerically. From Table 4.6.3, it is easy to see that the two solutions are either less than 1 or greater than 2, actually for these two cases the exact solutions are given. Thus it is practical to compare the numerical results and exact solutions. They match very well.

As for α depends upon u(x,t), the problem is same as Eq.(4.65) and except that $\alpha = \frac{\gamma}{(\mu + \int_0^1 q(u)dx)^2}$ which is different from PTC problem. The numerical results are listed in Table 4.6.4, where u(0), u(1) and v(0), v(1) are numerical solutions at the end points x = 0 and x = 1 respectively, and also the numerical solutions of u and vare obtained by the same program except that the initial values $u(j), j = 1, \dots, 51$ for u are all zero and initial values $v(j), j = 1, \dots, 51$ for v are greater than 1. From Table 4.6.4, it is easy to see that, when β is close to or greater than 0.4098, there are two numerical solutions (actually, there should be three solutions, one of them

Table 4.6.3

Ι	<i>u</i> (0)	u(1)	w(0)	w(1)
.3890E-03	.225076E+01	.200426E+01	.102417E-03	.907926E-04
.2295E-02	.783094E+02	.696083E+02	.355357E-02	.315874E-02
.4200E-02	.262269E+03	.233128E+03	.119053E-01	.105825E-01
.6106E-02	.554322E+03	.492731E+03	.251645E-01	.223685E-01
.8011E-02	.954162E+03	.848144E+03	.433173E-01	.385043E-01
.9917E-02	.146221E+04	.129974E+04	.663825E-01	.590067E-01
.1182E-01	.207723E+04	.184642E+04	.943042E-01	.838260E-01
.1373E-01	.280278E+04	.249136E+04	.127245E+00	.113106E+00
.1563E-01	.363217E+04	.322860E+04	.164899E+00	.146577E+00
.1754E—01	.457412E+04	.406589E+04	.207663E+00	.184590E+00
.1944E—01	.561877E+04	.499446E+04	.255090E+00	.226747E+00
.2135E-01	.677711E+04	.602410E+04	.307679E+00	.273492E+00
.2326E-01	.804392E+04	.715016E+04	.365192E+00	.324615E+00
.2516E-01	.941174E+04	.836599E+04	.427291E+00	.379814E+00
.2707E-01	.108949E + 05	.968440E+04	.494629E+00	.439670E+00
.2897E-01	.124780E+05	.110916E+05	.566500E+00	.503555E+00
.3088E-01	.141776E+05	.126023E+05	.643662E+00	.572144E+00
.3278E-01	.159759E+05	.142008E+05	.725306E+00	.644716E+00
.3469E-01	.178919E+05	.159039E+05	.812291E+00	.722037E+00
.3659E-01	.199055E+05	.176938E+05	.903708E+00	.803296E+00
.3850E-01	.220379E+05	.195892E+05	.100052E+01	.889349E+00

End point values of solutions as $t \longrightarrow \infty$ of (4.65).

is unstable, so it is difficult to get it numerically). For $\beta < 0.4098$, there is only one solution. It is noticed that both Table 4.6.2 and Table 4.6.4 have same phenomena, that is, the temperatures at center point are almost the same while the temperatures at right end point increase as β decreases.

β	Ι	u(0)	u(1)	I	v(0)	v(1)
.6000	.4762E-1	.7369E+0	.5668E+0	.5502E-2	.1996E+1	.1268E+1
.5000	.4762E-1	.8503E+0	.6802E+0	.4794E-2	.1996E+1	.1352E+1
.4500	.4762E-1	.9258E+0	.7558E+0	.4414E-2	.1996E+1	.1399E+1
.4100	.4762E-1	.9996E+0	.8295E+0	.4096E-2	.1996E+1	.1438E+1
.4099	.4762E-1	.9998E+0	.8297E+0	.4095E-2	.1996E+1	.1438E+1
.4098	.4762E-1	.1000E+1	.8299E+0	.4095E-2	.1996E+1	.1438E+1
.4000	.4762E-1	.1020E+1	.8496E+0	.4015E-2	.1996E+1	.1449E+1
.3000	.4015E-2	.1996E+1	.1560E+1	.4015E-2	.1996E+1	.1560E+1
.2500	.2681E-2	.1996E+1	.1622E+1	.2681E-2	.1996E+1	.1622E+1
.2000	.2183E-2	.1996E+1	.1689E+1	.2183E-2	.1996E+1	.1689E+1

Table 4.6.4

End point values for solutions of coupled NTC problem

as $t \longrightarrow \infty$ for different initial values.

4.7 Conclusion for the Numerical Results

In the above several sections, different numerical methods are used. The numeri-

cal experiments for steady state PTC and NTC problems, time dependent PTC and NTC problems, PTC and NTC problems with smooth functions are



Lower curve is u(1) and upper curve is u(0) as functions of t

performed. From the numerical results, the properties of PTC and NTC problems are quite different. One of the important different properties is that if the external circuit is not connected for certain fixed α and β PTC problems have only one solution but NTC problems have three solutions. However, if the external circuit is connected, the situations are the same and they all have one, two or three solutions with respect to different parameters α and β . Another common property is that all time dependent solutions with the external circuit have surges. For Figure 4.7.1, Figure 4.7.2, Figure 4.7.3 and Figure 4.7.4, $\beta = 0.25$ is fixed. The value $\beta = 0.25$ is chosen because for this β there is only one solution which is greater than 1, thus the surges will appear.



Lower curve is u(1) and upper curve is u(0) as functions of t

In order to get data for the four figures in this section, the Crank-Nicholson difference scheme is used. The interval [0, 1] is divided into 50 equidistant pieces so that there are 51 unknowns. In the figures, u(0) and u(1) are the numerically obtained values at end points x = 0 and x = 1 respectively. t coordinate is for time.

For Figure 4.7.1 and Figure 4.7.2, the equations are as follows:

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \frac{\gamma}{(1+\mu \int_0^1 f(u)(-\sigma r - p(u))dx)^2} f(u)(-\sigma r - p(u)) &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$

where f is defined by (4.2) and p(u) is defined by (4.57), β is a fixed parameter. As



Lower curve is u(1) and upper curve is u(0) as functions of t

for Figure 4.7.3 and Figure 4.7.4, the equations are as follows:

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \frac{\gamma}{(\mu + \int_0^1 g(u)(-or - q(u))dx)^2} g(u)(-or - q(u)) &, 0 < x < 1\\ u_x(0) = 0 &, u_x(1) + \beta u(1) = 0 \end{cases}$$

where g(u) is defined by (4.44) and q(u) is defined by (4.60). The difference of the two equations above is in the denominator of the second term on the right hand side. This difference makes the figures look a little bit different. For Figure 4.7.1 and Figure 4.7.2, the graphs change smoothly when surges appear. For Figure 4.7.3 and Figure 4.7.4, the graphs change rapidly and steeply when surges appear. Also the difference between u(0) and u(1) for PTC problems are smaller than that for NTC problems.



Lower curve is u(1) and upper curve is u(0) as functions of t

From the four figures, it is easy to see that for smooth functions more time

is required to reach their steady state. For the smooth functions, whenever the solutions reach their steady states, they will satisfy the stopping criterion (the absolute maximum of difference between two time steps at each nodes is less than a fixed small number) very quickly. Another common property for the problems with smooth functions is that the numerical values are greater than that obtained from the corresponding original ones.

As it is known, when $\beta < 0.4098$, there is only one solution which is greater than 1 and the surges also appear. Thus the numerical experiments are also done for $\beta =$ 0.40, 0.35, 0.30, 0.20, 0.15, 0.10 and 0.05, the results and figures look similar. They all are between 1 and 2. The difference between u(0) and u(1) for NTC problems is greater than that for PTC problems. The graphs for NTC problems change more rapidly and steeply than for PTC problems.

Chapter 5

Convergence and Stability Analysis

Since the steady state solutions for time dependent problems are not unique for some parameters, is it possible to get all solutions numerically? In the actual numerical experiments, it seems difficult to obtain some solutions. As mentioned in $\S4.2.2$ for Eq.(4.42), it is difficult to get the numerical solution, corresponding to the one spanning cold and transient regions, which theoretically exists. Now through discussion of linear stability, though a direct proof is not obtained, the numerical results would give some indication for that situation.

5.1 Linear Stability Analysis

Here a general idea of linear stability analysis [13, 14, 29, 30, 31, 49] is discussed. Generally for

$$\frac{du}{dt} = F(x, u), \qquad t > 0 \quad \& \quad u(0, x) = u_0(x) \tag{5.1}$$

where F(x, u) is a continuously differentiable function of x and u, x is a space coordinate in the interval [0, l] and l is a constant, $u_0(x)$ is a function of x and independent on t. Let $u_1(x)$ satisfy

$$F(x,u) = 0 \tag{5.2}$$

Then $u_1(x)$ is said a steady state solution of (5.1) with initial value $u_1(x)$.

Definition 5.1 Suppose the solution of Eq.(5.1) exists and is denoted as u(t, x). A steady state solution $u_1(x)$ is said to be stable if for any ε there exists a δ_{ε} such that

$$|u(t,x)-u_1(x)| \le \varepsilon, \quad t>0$$

whenever $|u(0,x) - u_1(x)| < \delta_{\varepsilon}$. If a steady state solution of (5.1) is not stable, it is said to be unstable.

Obviously, if a steady state solution of (5.1) is unstable, it is hard to get that solution by a numerical method because any error for initial values will grow. However, is it possible for us to know for what kind of function F the corresponding solution is stable?

Let $u_1(x)$ be a solution of (5.2) and u(t,x) be a solution of (5.1) corresponding to the initial value $u_0(x)$ which is close to $u_1(x)$. Let δ be an arbitrary small number and v(t,x) an arbitrary function such that

$$u(t,x) = u_1(x) + \delta v(t,x)$$

Substitute u(t, x) into (5.1), thus

$$\delta \frac{dv}{dt} = F(x, u_1(x) + \delta v(t, x)) - F(x, u_1(x))$$

$$= F_u(x, u_1(x))\delta v(t, x) + R(x, u_1(x), v(t, x), \delta)$$
(5.3)

If δ is very small and R is of higher order in δ , $\lim_{\delta \to 0} \frac{|R|}{\delta} = 0$, it would appear that the behavior of the solution of (5.3) is determined by the linearized equation

$$\frac{dv}{dt} = F_u(x, u_1(x))v(t, x) \tag{5.4}$$

A basic result about stability may be stated as follows. For convenience, denote

$$\kappa \equiv F_u(x, u_1(x)) \tag{5.5}$$

Theorem 5.1 [13, 14, 29, 31, 49] Let $u_1(x)$ be a steady state solution of (5.1) and assume that

$$F(x, u_1(x) + \delta v(t, x)) = F_u(x, u_1(x))\delta v(t, x) + R(x, u_1(x), v(t, x), \delta)$$

where the remainder term R is $O(\delta^2)$ for δ sufficiently small. Then $u_1(x)$ is stable if $\kappa < 0$ and unstable if $\kappa > 0$, where κ is given by (5.5). #

The theorem 5.1 is for the case where F_u is a scalar function of x and u, etc.. If it is an operator, the result is true except that instead of using the sign of F_u the sign of the eigenvalues is used. The more general result may be stated as follows:

Theorem 5.2 [14, 29, 31, 49] Let $u_1(x)$ be a steady state solution of (5.1) and assume that

$$F(x, u_1(x) + \delta v(t, x)) = F_u(x, u_1(x))\delta v(t, x) + R(x, u_1(x), v(t, x), \delta)$$

where the remainder term R is $O(\delta^2)$ for δ sufficiently small and F_u is a self-adjoint operator. Then $u_1(x)$ is stable if all eigenvalues of F_u are less than zero and unstable if at least one of the eigenvalues of F_u is greater than zero. #

To find an eigenvalue for our operator is not easy. However, for a self-adjoint operator, even if it is not easy to find its eigenvalues, it is still possible to tell whether the operator has a negative eigenvalue or not. This result is based on the Rayleigh quotient. If $A: X \to X$ is a self-adjoint operator, where X is a Hilbert space,

$$R(x) = \frac{(Ax, x)}{(x, x)}$$

is said to be the Rayleigh quotient [14, 31]. If ||x|| = 1, then R(x) = (Ax, x). It is easy to see that if x is an eigenvector and λ is an eigenvalue, then $R(x) = \lambda$. Define

$$L_A = \inf_{x \in X, x \neq 0} R(x) = \inf_{x \in X, ||x|| = 1} (Ax, x)$$
(5.6)

and

$$U_A = \sup_{x \in X, x \neq 0} R(x) = \sup_{x \in X, ||x|| = 1} (Ax, x)$$
(5.7)

Hence, if A is bounded below, L_A is finite; if A is bounded above, U_A is finite.

Theorem 5.3 [14, 31] Let A be symmetric and bounded below. If there is an element $x \in X$ for which the infimum in (5.6) is attained, (L_A, x) is an eigenpair and L_A is the lowest eigenvalue of A. Similarly, if there is an element $y \in X$ for which the supremum in (5.7) is attained, (U_A, y) is an eigenpair and U_A is the largest eigenvalue of A. #

From theorem 5.3, if A is a self-adjoint operator and there is a function which makes the Rayleigh quotient negative, then there is at least one negative eigenvalue for A. Similarly, if there is a function which makes the Rayleigh quotient positive, then A has at least one positive eigenvalue.

5.2 The Numerical Results for Stability and Stability Analysis

Based on the general ideas in §5.1, the NTC problem met in §4.4 and §4.6.2 will be discussed now. Suppose $u_0(x)$, $\alpha_0 = \gamma I_0^2$ is a steady state solution of (2.10). Thus

$$\begin{cases} \frac{d^2 u_0(x)}{dx^2} + \alpha_0 g(u_0(x)) = 0 &, 0 < x < 1\\ u_{0,x}(t,0) = 0 &, u_{0,x}(t,1) + \beta u_0(t,1) = 0 \end{cases}$$
(5.8)

where g is defined by (3.16) (or $\frac{1}{\sigma(u)}$), $\alpha_0 = \gamma I_0^2$, $I_0 = \frac{1}{\mu + \int_0^1 g(u_0(x)) dx}$.

Using the linearized stability analysis method, set

$$u(t,x) = u_0(x) + \delta v(t,x) \tag{5.9}$$

with δ small. Substituting in (2.10), then

$$\begin{cases} \frac{\partial u}{\partial t} = \frac{\partial^2 u}{\partial x^2} + \gamma I^2 g(u) &, 0 < x < 1\\ u_x(t,0) = 0 &, u_x(t,1) + \beta u(t,1) = 0 \end{cases}$$
where $I = \frac{1}{\mu + \int_0^1 g(u) dx}$. Since $\frac{\partial u_0(x)}{\partial t} = 0$ and $u_0(x)$ satisfies (5.8), therefore
$$\begin{cases} \delta \frac{\partial v}{\partial t} = \frac{\partial^2 u_0}{\partial x^2} + \delta \frac{\partial^2 v}{\partial x^2} + \gamma I^2 g(u_0 + \delta v) &, 0 < x < 1\\ v_x(t,0) = 0 &, v_x(t,1) + \beta v(t,1) = 0 \end{cases}$$
(5.10)

So if we can solve Eq.(5.10) and get v(t, x), then according to the decaying or growing of v(t, x) we can say that the steady state solution of (2.10) is stable or unstable. However it is not easy to get the solution to Eq.(5.10). A linearized method is used to obtain an approximate solution to Eq.(5.10). Using a Taylor expansion, thus

$$g(u_0 + \delta v) = g(u_0) + g'(u_0)\delta v + R_1$$

where R_1 is of higher order in δ , and

$$I = \frac{1}{\mu + \int_0^1 g(u)dx} = \frac{1}{\mu + \int_0^1 (g(u_0) + g'(u_0)\delta v + R_1)dx}$$

= $\frac{1}{\mu + \int_0^1 g(u_0)dx + \delta \int_0^1 g'(u_0)vdx + \cdots} = \frac{1}{\frac{1}{I_0} + \delta \int_0^1 g'(u_0)vdx + \cdots}$
= $\frac{I_0}{1 + \delta I_0 \int_0^1 g'(u_0)vdx + \cdots} = I_0(1 - \delta I_0 \int_0^1 g'(u_0)vdx + \cdots)$

where \cdots means the omitted part contains terms with higher order (at least 2) of δ . Similarly, using the above results

$$\gamma I^2 = \gamma I_0^2 (1 - 2\delta I_0 \int_0^1 g'(u_0) v dx + \cdots)$$

and

$$\begin{split} \gamma I^2 g(u) &= \gamma I_0^2 (1 - 2\delta I_0 \int_0^1 g'(u_0) v dx + \cdots) (g(u_0) + g'(u_0) \delta v + \cdots) \\ &= \gamma I_0^2 g(u_0) + \delta (\gamma I_0^2 g'(u_0) v - 2\gamma I_0^3 g(u_0) \int_0^1 g'(u_0) v dx) + \cdots \end{split}$$

If we omit all terms which contain higher orders of δ and divide by δ on both sides, then Eq.(5.10) becomes

$$\begin{cases} \frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} + \gamma I_0^2 g'(u_0) v - 2\gamma I_0^3 g(u_0) \int_0^1 g'(u_0) v dx &, 0 < x < 1\\ v_x(t,0) = 0 &, v_x(t,1) + \beta v(t,1) = 0 \end{cases}$$
(5.11)

As it is known, the operator on the right hand side of Eq.(5.11) is not a selfadjoint operator. Even if there is a function which makes the right hand side to be negative, it could not be said that there is a negative eigenvalue for that operator. However, in the NTC problem since $g'(u_0) > 0$ the term $-2\gamma I^3 g'(u_0) \int_0^1 g(u_0) v dx$ is a stablizing term (by experience), adding this term to the right hand side makes the right hand side operator self-adjoint. If the modified operator is unstable, the original operator should be more unstable. Thus, instead of considering Eq.(5.11), the following equation is considered.

$$\begin{cases} \frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} + \gamma I_0^2 g'(u_0) v - 2\gamma I_0^3 g(u_0) \int_0^1 g'(u_0) v dx \\ -2\gamma I_0^3 g'(u_0) \int_0^1 g(u_0) v dx &, 0 < x < 1 \\ v_x(t,0) = 0 &, v_x(t,1) + \beta v(t,1) = 0 \end{cases}$$
(5.12)

Therefore, if the operator on the right hand side of Eq.(5.12) has positive eigenvalue, then the steady state solution $u_0(x)$ is unstable. The following method is just one to demonstrate that the right hand side operator of Eq.(5.12) has positive eigenvalue. To form the Rayleigh quotient $(Ax, x)/||x||^2$, now multiply the right hand

side of the first equation of (5.12) by v and integrate from 0 to 1 with respect to x, thus

$$R(v)\int_0^1 v^2 dx = \int_0^1 v_{xx}v dx + \gamma I_0^2 \int_0^1 g'(u_0)v^2 dx - 4\gamma I_0^3 \int_0^1 g(u_0)v dx \int_0^1 g'(u_0)v dx$$

Simplifying

$$R(v)\int_0^1 v^2 dx = vv_x|_0^1 - \int_0^1 v_x^2 dx + \gamma I_0^2 \int_0^1 g'(u_0)v^2 dx - 4\gamma I_0^3 \int_0^1 g(u_0)v dx \int_0^1 g'(u_0)v dx$$

Using the boundary conditions

$$R(v) \int_{0}^{1} v^{2} dx = -\beta v^{2}(1) - \int_{0}^{1} v_{x}^{2} dx + \gamma I_{0}^{2} \int_{0}^{1} g'(u_{0}) v^{2} dx \qquad (5.13)$$
$$-4\gamma I_{0}^{3} \int_{0}^{1} g(u_{0}) v dx \int_{0}^{1} g'(u_{0}) v dx$$

To examine if there is a function which makes the right hand side of Eq.(5.13) to be positive or not choose $v(x) = \cos(\lambda x)$ then

$$R(v) \int_0^1 \cos^2(\lambda x) dx = -\beta \cos^2(\lambda) - \int_0^1 \lambda^2 \sin^2(\lambda x) dx$$
$$+\gamma I_0^2 \int_0^1 g'(u_0) \cos^2(\lambda x) dx$$
$$-4\gamma I_0^3 \int_0^1 g(u_0) \cos(\lambda x) dx \int_0^1 g'(u_0) \cos(\lambda x) dx$$

hence

$$\frac{1}{2}R(v)(1+\frac{1}{2\lambda}sin(2\lambda)) = -\beta cos^2(\lambda) - \frac{\lambda^2}{2}(1-\frac{sin(2\lambda)}{2\lambda}) +\gamma I_0^2 \int_0^1 g'(u_0)cos^2(\lambda x)dx -4\gamma I_0^3 \int_0^1 g(u_0)cos(\lambda x)dx \int_0^1 g'(u_0)cos(\lambda x)dx$$
(5.14)

For convenience, denote

$$R(v) = A(u_0, \alpha, \mu, \beta, \lambda) = \frac{1}{\frac{1}{2}(1 + \frac{1}{2\lambda}sin(2\lambda))} [-\beta cos^2(\lambda)$$
(5.15)

$$-\frac{\lambda^2}{2}(1-\frac{\sin(2\lambda)}{2\lambda})+\gamma I_0^2\int_0^1 g'(u_0)\cos^2(\lambda x)dx$$
$$-4\gamma I_0^3\int_0^1 g(u_0)\cos(\lambda x)dx\int_0^1 g'(u_0)\cos(\lambda x)dx]$$

Thus (5.15) is the Rayleigh quotient. That's why in the following the sign of $A(u_0, \alpha, \mu, \beta, \lambda)$ is discussed. Considering the boundary conditions, choose λ so that

$$v_x(0) = 0,$$
 $v_x(1) + \beta v(1) = 0$

Therefore

$$-\lambda sin(\lambda) + \beta cos(\lambda) = 0$$

i.e.,

$$\lambda tan(\lambda) = \beta \tag{5.16}$$

As it is known, there are infinitely many solutions to Eq.(5.16). However, to prove that the operator on the right hand side of Eq.(5.12) has a positive eigenvalue, if we can find one λ for which $\cos(\lambda x)$ makes $A(u_0, \alpha, \mu, \beta, \lambda)$ positive, then based on the general discussion in § 5.1, the operator on the right hand side of Eq.(5.12) has positive eigenvalue, so the smallest positive solutions of (5.16) can be taken. Results obtained giving λ for various β are summarized in Table 5.1.1.

Table 5.1.1

β	λ	β	λ	β	λ	β	λ
.100	.31105286	.250	.48009443	.400	.59324193	.550	.68005705
.150	.37787765	.300	.52179116	.450	.62444466	.600	.70506549
.200	.43284073	.350	.55922329	.500	.65327120	.650	.72850811

Smallest solution λ of (5.16) for given β .

Now we have the values of β and λ . The three solutions of I_0 (i.e., α_0) and u_0 for the steady state situation can be obtained numerically. Thus, when $\beta > 0.4098$, there are three solutions, that means there exist three different values of α for a fixed β .

β	λ	α	β	λ	α	
0.450		0.34013605e+00			0.34013605e+00	
	0.62444466	0.24069230e-02	0.500	0.65327120	0.28149190e-02	
		0.33633682e+00			0.33233449e+00	
	0.68005705	0.34013605e+00		0.70506549	0.34013605e+00	
0.550		0.32305950e-02	0.600		0.36516950e-02	
		0.32855827e+00			0.32496199e+00	
		0.34013605e+00		0.75055808	0.34013605e+00	
0.650	0.72850811	0.40758720e-02	0.700		0.45013170e-02	
		0.32153054e+00			0.31825082e+00	

Table 5.1.2

Values of α and λ for given β .

As discussed in $\S4.3.2$, there always exists a cold solution with

$$\frac{\gamma I^2}{\beta} = \frac{\gamma I}{\beta} (1 - \mu I)$$

thus $I_0 = \frac{1}{1+\mu}$. Therefore for $\beta > 0.4098$, there always exists a cold solution with $I_0 = \frac{1}{1+\mu}$, i.e., α_0 is fixed though β is different. It is easy to see from Figure 4.3.4,

one other α (corresponding to the solution spanning cold and transient regions) should be close to α_0 and the third one corresponding to the whole solution in the

β	λ	α	u(0)	u(1)	$A(u_0, lpha, \mu, eta, \lambda)$
0.45	0.62444466	0.34013605E+0	0.92592603	0.75585800	-0.77986223
		0.33633682E+0	1.03792620	0.83581495	2.38273615
		0.24069231E-2	1.64676213	1.22786081	-33.17415905
0.50	0.65327120	0.34013605E+0	0.85034019	0.68027216	-0.85352647
		0.33233449E+0	1.06160045	0.82755232	3.27942348
		0.28149190E-2	1.64484048	1.18660295	-35.73973656
0.55	0.68005705	0.34013605E+0	0.78849727	0.61842924	-0.92495513
		0.32855827E+0	1.08037066	0.81649649	3.87784278
		0.32305950E-2	1.64317393	1.14788091	-38.15972233
0.60	0.70506549	0.34013605E+0	0.73696148	0.56689346	-0.99423480
		0.32496199E+0	1.09650326	0.80412000	4.42084587
		0.36516951E-2	1.64167249	1.11149311	-40.44283008
0.65	0.72850811	0.34013605E+0	0.69335431	0.52328628	-1.06144810
		0.32153055E+0	1.11053240	0.79098970	4.90334769
		0.40758718E-2	1.64029598	1.07726705	-42.60123539
0.70	0.75055808	0.34013605E+0	0.65597671	0.48590869	-1.12667477
		0.31825081E+0	1.12317467	0.77747375	5.25375581
		0.45013172E-2	1.63901687	1.04503202	-44.64574623

Table 5.1.3

Table to demonstrate the stability of solution of the NTC problem.

transient region is far away from the other two. The different values of α corresponding to the β and λ are listed in Table 5.1.2. The values of α and β are obtained by the methods discussed in § 4.1.2 and §4.2.2.

Now the value of $A(u_0, \alpha, \mu, \beta, \lambda)$ can be numerically evaluated. Using the data given in Table 5.1.2 and the corresponding $u_0(x)$ (actually the values are numerically obtained at meshpoints), the numerical values of $A(u_0, \alpha, \mu, \beta, \lambda)$ and the values of $u_0(0)$ and $u_0(1)$ are listed in Table 5.1.3. Here the integrals in (5.15) are evaluated by a trapezoidal rule over the values of u at the meshpoints. From Table 5.1.3, it is found that the values of $A(u_0, \alpha, \mu, \beta, \lambda)$ assume positive numbers for the solution spanning cold and transient regions, that means that the operator on the right hand side of Eq.(5.12) has positive eigenvalue, thus the corresponding solutions are unstable. So this numerical result gives some indication why we can not get that solution for the time-dependent problem even if we take the initial values very close to the solution. As for the other two solutions, it seems that they are stable (the values of $A(u_0, \alpha, \mu, \beta, \lambda)$ assume negative numbers at those solutions) and the numerical results for steady state problems agree very well with the numerical results obtained for time-dependent problems.

Chapter 6

Conclusion

In this dissertation, attention has been focused on a system of nonlinear parabolic equations which model the thermistor. First, a partial differential equation which describes the heat and current distribution within a thermistor is given, then under some specified boundary conditions two types of nonlinear one (space) dimensional parabolic equations are derived. One type has the PTC property, and the other has NTC property.

Theoretically, existence and uniqueness (whenever applicable) are studied both for steady state and time dependent problems. The properties of the solutions are thoroughly discussed. Three methods are employed to prove the existence and uniqueness for steady state problems. One is to transform the original differential equations into integral equations. By proving the existence and uniqueness of the solutions for integral equations, existence and uniqueness for original problems are obtained.

The other method is to change the variables. Instead of the original boundary value problems being considered, corresponding initial value problems are studied. From the relationship of the original boundary value problems and the initial value problems, the existence and uniqueness are obtained again.

The third method is a monotone method. By using concept of upper and lower solutions, existence and uniqueness for steady state problems are obtained. As for the time dependent problems, if the external circuit is not connected, existence and uniqueness are obtained too. If the external circuit is connected, existence and uniqueness are obtained only for the PTC type problem.

Many numerical experiments are done both for steady state problems and time dependent problems. There are three methods used to obtain numerical solutions. The first one is a shooting method which is used for steady state problems. The second one is moving meshpoint method in which the stepsize, interval length and the number of meshpoints are changing according to some interface conditions. The third one is fixed meshpoints in which the stepsize, interval length and the number of meshpoints are fixed. Obviously, the moving meshpoint method is more complicated than the fixed meshpoint method. Numerical results showed that accuracy of the numerical solutions obtained by the moving meshpoint method is better than that obtained by the fixed meshpoint method. Since the algorithm for the moving meshpoint method is much more complicated than that for the fixed meshpoint method, the advantage of the moving meshpoint method is jeopardized though the precision is improved by using moving meshpoint method. The fixed meshpoint method gives steady state solutions which agree sufficiently with those obtained by other methods. Hence the moving meshpoint method is only used once for PTC problem. For most problems considered the conductivity used is continuous but not differentiable everywhere, for comparison, the same problems with smooth conductivity are numerically solved. Many numerical results are obtained. Numerical results for typical parameters are listed in several tables and some of them are demonstrated in figures. The solutions for the problems with smooth conductivity change more smoothly at interface points than the solutions for original problems with the non-differentiable conductivity.

As one of the three (if there are three) solutions can not be found numerically, the concept of stability and instability is reviewed. Thus the stability and instability of solutions are briefly discussed.

In this dissertation, not only the NTC problem is studied as in most other literature [3, 4, 35, 36, 37, 38, 40, 41], the PTC problem is also studied in detail. The different properties of PTC and NTC problems are also reflected in their solutions. Those properties are studied thoroughly. For PTC problem, if the external circuit is not connected, for the given parameters α and β , there always exists one solution. But for NTC problem, if the external circuit is not connected, for some given parameters α and β , there exist three solutions, for some other given parameters α and β , there exist only one solution. All solutions of the steady state PTC and NTC problems can be numerically obtained. As for time dependent problems without external circuit, the solution of the PTC problem can be numerically obtained since there is only one solution and the numerical solution agrees sufficiently well with the numerical solution of the corresponding steady state PTC problem. The NTC problem has three solutions, but only two of them can be numerically obtained. One of them seems quite unstable and for any initial value the solution converges to one or another of the stable solutions, the third solution can not be obtained numerically even if it exists theoretically. If the external circuit is connected, both for PTC and NTC problems there exist three situations. That is, there exist one, two or three solutions for different parameters α and β . If there exist three solutions for the time dependent external circuit connected problems, the common property of the solutions for PTC and NTC problems is that one of the three solutions is numerically unstable. If there exist one or two solutions, all of them can be numerically obtained.

Though many theoretical and numerical results have been obtained, there are still some uncertainties, such as uniqueness (at most locally) for NTC problems and a direct proof of the instability, etc. Currently, there seems to be little literature contributed to two dimensional problems.

Bibliography

- D.R. Westbrook, The Thermistor: A Problem in Heat and Current Flow. Numer. Methods for PDE. 5, pp 259-273 (1989).
- [2] S. Nelson, M. Neuman, Generalization of Projection Method with Applications to SOR Theory for Hermitian Positive Semidefinite Linear Systems. Numer. Math. 51, pp 123-141 (1987).
- [3] A.C. Fowler, I. Frigaard, S.D. Howison, Temperature Surges in Current-Limiting Circuit Devices. SIAM J.Appl. Math. Vol 52, N0.4, pp.998-1011, 1992.
- [4] D.H. Sattinger, Monotone Methods in Nonlinear Elliptic and Parabolic Boundary Value Problems. Indiana Univ. Math. J. Vol. 21, No. 11 (1972). pp 979–1000.
- [5] D.M. Young, Iterative Solution of Large Linear Systems. Academic Press. 1971.
- [6] L.A. Hageman, D.M. Young, Applied Iterative Method. Academic Press. 1981.
- [7] G.W. Stewart, Introductions to Matrix Computations. Academic Press. 1973.
- [8] J. Crank, Free and Moving Boundary Problems. Clarendon Press. 1988.
- [9] G.D. Smith, Numerical Solution of Partial Differential Equation. Clarendon Press. 1985.
- [10] R.S. Varga, Matrix Iterative Analysis. Prentice-Hall, Inc. 1962.
- [11] Y.S. Li, Y.Q. Huang, Numerical Approximation. People's Education Publishing House. 1978.
- [12] O.C. Zienkiewicz, The Finite Element Method in Engineering Science. McGraw-Hill Publishing Company Ltd. 1971.
- [13] I. Stakgold, Green's Functions And Boundary Value Problems. A Wiley-Interscience Publication. 1979.
- [14] I. Stakgold, Boundary Value Problems of Mathematical Physics. The Macmillan Company, New York. v2, 1967.
- [15] E. Zeidler, Nonlinear Functional Analysis and Its Application. Springer-Verlag. v1, 1986.
- [16] M.H. Protter, H.F. Weinberger, Maximum Principles in Differential Equations. Springer-Verlag. 1984.
- [17] H.B. Keller, Numerical Solution of Two Point Boundary Value Problems. SIAM(24), 1976.
- [18] H.B. Keller, On the solution of singular and semidefinite linear system by iteration. J. SIAM Numer Anal. Ser. B, Vol. 2, 1965.
- [19] M.A. Hastaoğlu, Numerical Solution of Three dimensional Moving Boundary Problem: Melting and Solidification with Blanketing of a Third Layer. Chemical Engineering Science, Vol. 42, No. 10, pp2417-2423, 1987.
- [20] B.M. Herbst, S.W. Schoombie, A Moving Petrov-Galerkin Method for Transport Equations. Int. J. for Numer. Meth. in Eng., Vol. 18, 1321-1336, 1982.

- [21] B.M. Herbst, S.W. Schoombie, A.R. Mitchell, Equidistributing Principle in Moving Finite Element Methods. J. of Comput. and Applied Math. 9, pp 377-389, 1983.
- [22] I. Babuška, W.C. Rheinboldt, Reliable Error Estimation and Mesh Adaptation for the Finite Element Method, in: J.T. Oden, Ed. Computational Methods in Nonlinear Mechanics, North-Holland Publishing Company, pp 67-108, 1980.
- [23] D.R. Lynch, W.G. Gray, Finite Element Simulation of Flow Deforming Regions.J. of Comput. Physics 36, pp135-153, 1980.
- [24] D.R. Lynch, K. O'Neill, Continuously Deforming Finite Elements for the Solution of Parabolic Problems, with and without Phase Change. Int. J. for Numer. Meth. in Eng., Vol. 17,pp81-96, 1981.
- [25] R.J. Gelinas, S.K. Doss, K. Miller, The Moving Finite Element Methods: Application to General Partial Differential Equations with Multiple Large Gradients.
 J. of Comput. Physics 40, pp 202-249, 1981.
- [26] R. Thrasher, K. Sepehrnoori, On Equidistributing Principles in Moving Finite Element Methods. J. of Comput. and Applied Math. 16, pp 309-318, 1986.
- [27] M.J. Baines, Recent Developments in Moving Finite Element Methods, in: J.R. Whiteman, Ed. The Mathematics of Finite Elements and Applications VI. Academic Press, 1988.
- [28] M.J. Baines, Moving Element Methods for Time Dependent Problems. in: Numerical Methods for Fluid Dynamics, IV, Oxford, 1988.

- [29] C. Rogers, W.F. Ames, Nonlinear Boundary Value Problems in Science and Engineering. Academic Press, Inc. 1989.
- [30] J.D. Logan, Applied Mathematics, a contemporary approach. John Wiley & Sons Inc., 1987
- [31] D.X. Xia, Z.R Wu, S.C. Yan, W.C. Shu, Theory of Real Function and Functional Analysis. People's Education Publishing House. Beijing, 1979.
- [32] Department of Mathematics, Wuhan University, Mathematical Analysis. People's Education Publishing House. 1978
- [33] F.B. Weissler, Single Point Blow-up for a Semilinear Initial Value Problem. J. Differential Equ. 55, pp204-224 (1984).
- [34] H. Fujita, On the Nonlinear Equations $\Delta u + e^u = 0$ and $\frac{\partial v}{\partial t} = \Delta v + e^v$. Bull. Amer. Math. Soc. 75, pp132-135, 1969.
- [35] G. Cimatti, A bound for the Temperature in the Thermistor Problem. IMA J. Appl. Math. 40, pp 15-22 (1988).
- [36] G. Cimatti, Remark on Existence and Uniqueness for the Thermistor Problem Under Mixed Boundary Conditions. Quarterly Appl. Math. Vol. XLVII, No. 1. pp 117-121. 1989.
- [37] X. Chen, A. Friedman, The Thermistor Problem for Conductivity which Vanishes at Large Temperature. IMA Preprint Series #792. May, 1991.
- [38] X. Chen, A. Friedman, The Thermistor Problem with One-zero Conductivity. IMA Preprint Series #793. May, 1991.

- [39] X. Chen, Existence and Regularity of Solutions of A Nonlinear Degenerate Elliptic System Arising from A Thermistor Problem. Preprint.
- [40] W. Allegretto, H. Xie, Existence of Solutions for the Time-Dependent Thermistor Equations. IMA J. Appl. Math. 48, pp271-281, 1992.
- [41] H. Xie, W. Allegretto, C^α(Ω) Solutions of a Class of Nonlinear Degenerate Elliptic Systems Arising in the Thermistor Problem. SIAM J. Math Anal. 22.(1991).
 No.6. PP1491-1499.
- [42] K. Miller, Recent Results on Finite Element Methods with Moving Nodes, in I. Babuška, O.C. Zienkiwicz, J. Gago, E.R. de A. Oliveira, Ed. Accuracy Estimates and Adaptive Refinements in Finite Element Computations. Jhon Wiley & Sons Ltd. 1986.
- [43] D.R. Wang, Numerical and Optimal Methods for Solving Nonlinear Equations. People's Education Publishing House. 1979.
- [44] J.L. Lions, E.M. Magenes, Non-Homogeneous Boundary Value Problems and Applications. Vol. I, Springer-Verlag, 1972.
- [45] J.L. Lions, E.M. Magenes, Non-Homogeneous Boundary Value Problems and Applications. Vol. II, Springer-Verlag, 1972.
- [46] A. Friedman, Partial Differential Equations of Parabolic Type. Prentice-Hall, INC. 1964.
- [47] A. Friedman, Partial Differential Equations Holt, Rinehart and Winston, INC. 1969.

- [48] F. Treves, Basic Linear Partial Differential Equations. Academic Press. 1975.
- [49] J. Smoller, Shock Waves and Reaction-Diffusion Equations. Springer-Verlag. 1983.
- [50] IMSL Math/Library. Fortran Subroutines for Mathematical Application. 1987.
- [51] O.A. Ladyzhenskaya, V.A. Solonnikov and N.N. Ural'cheva, Linear and Quasilinear Equations of Parabolic Type. AMS Translations of Mathematical Monographs, Vol. 23, 1968.

Appendix A

IMSL Subroutine IVPRK[50]

There are lots of IMSI subroutines installed in AIX machines. Here only the related subroutine is included.

Name: IVPRK (Single Precision)

Purpose: Solve an iniatial-value problem for ordinary differential equations using the Runge-Kutta-Verner fifth-order and sixth-order method.

Usage: CALL IVPRK(IDO,N,FCN,T,TEND,TOL,PARAM,Y)